UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, DC 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of The Securities Exchange Act of 1934

Date of report (Date of earliest event reported): June 19, 2010

CORN PRODUCTS INTERNATIONAL, INC.

(Exact Name of Registrant as Specified in Its Charter)

Delaware (State or Other Jurisdiction of Incorporation) **1-13397** (Commission File Number)

22-3514823 (IRS Employer Identification No.)

5 Westbrook Corporate Center, Westchester, Illinois

(Address of Principal Executive Offices)

60154-5749 (Zip Code)

(708) 551-2600

(Registrant's Telephone Number, Including Area Code)

Not Applicable

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Item 1.01. Entry Into Material Definitive Agreement.

On June 19, 2010, Corn Products International, Inc. ("Corn Products" or the "Company") entered into an International Share and Business Sale Agreement (the "Sale Agreement") with Akzo Nobel N.V. (the "Seller"), pursuant to which Corn Products has agreed to purchase certain business entities and assets comprising the Seller's specialty starches business, commonly known as "National Starch" ("National Starch"), for a purchase price of \$1.3 billion in cash, subject to certain post-closing adjustments (the "Acquisition"). The Acquisition has been approved by the respective boards of directors of Corn Products and the Seller, does not require approval by the stockholders of either Corn Products or the Seller and is expected to close during the third quarter of 2010.

The Sale Agreement contains customary representations, warranties and covenants by the Seller and Corn Products, and is subject to customary closing conditions, including the receipt of certain regulatory approvals, such as the expiration or earlier termination of the waiting period under the Hart-Scott-Rodino Act. Corn Products' obligation to close the Acquisition is not conditioned upon its ability to secure financing for the Acquisition. The parties are obligated, subject to certain limitations, to indemnify each other under the Sale Agreement for breaches of warranties, nonfulfillment or breaches of covenants and agreements, and for certain third-party claims. The Seller and Corn Products have agreed to enter into related transaction agreements at the closing, including a transition services agreement and an intellectual property license.

On June 19, 2010, in connection with the execution of the Sale Agreement, Corn Products has signed a commitment letter (the "Commitment Letter") with JPMorgan Chase Bank, N.A. ("JPMCB") and J.P. Morgan Securities Inc. (together with JPMCB, "J.P. Morgan"). Under the Commitment Letter, JPMCB is authorized to act as administrative agent (the "Agent"), J.P. Morgan Securities Inc. is authorized to act as sole lead arranger and sole book runner. The Commitment Letter provides that JPMCB will commit to provide an aggregate principal amount of \$1.35 billion under a 364-day term loan credit facility (the "Bridge Facility"). If the Bridge Facility is drawn, Corn Products would use the proceeds to finance the Acquisition and related acquisition expenses. The commitment to provide the Bridge Facility is subject to certain conditions, including the absence of a Group Material Adverse Effect (as defined in the Sale Agreement), the negotiation of definitive documentation, pro forma compliance with the financial covenants contained in the definitive documentation and other customary closing conditions more fully set forth in the Commitment Letter.

Corn Products will pay certain customary fees and expenses in connection with obtaining the Bridge Facility, including an underwriting fee, which was paid by the Company following execution of the Commitment Letter.

J.P. Morgan is providing certain advisory services to Corn Products in connection with the Acquisition.

The foregoing summaries of the Sale Agreement and the Commitment Letter are qualified in their entirety by the terms and conditions of the Sale Agreement and the Commitment Letter, which are filed as Exhibits 2.1 and 10.1 to this Current Report on Form 8-K, respectively, and are incorporated by reference herein.

The Sale Agreement has been included to provide investors and security holders with information regarding its terms. The warranties and covenants contained in the Purchase Agreement were made only for purposes of that agreement and as of specific dates; were solely for the benefit of the parties to the Sale Agreement; may be subject to limitations agreed upon by the parties, including being qualified by confidential disclosures made for the purposes of allocating contractual risk between the parties to the Sale Agreement instead of establishing these matters as facts; and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors. Investors should not rely on the warranties and covenants or any description thereof as characterizations of the actual state of facts or condition of National Starch, the Seller or Corn Products or any of their respective subsidiaries or affiliates. Moreover, information concerning the subject matter of the warranties and covenants may change after the date of the Sale Agreement, which subsequent information may or may not be fully reflected in public disclosures by Corn Products.

Item 7.01. Regulation FD Disclosure

On June 21, 2010, Corn Products issued a press release announcing the execution of the Sale Agreement. A copy of the press release is attached hereto as Exhibit 99.1 and is incorporated by reference herein.

2

The information contained in, or incorporated into, this Item 7.01, including Exhibit 99.1 attached hereto, is being furnished and shall not be deemed "filed" for the purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference into any registration statement or other filing under the Securities Act of 1933, as amended, or the Exchange Act, except as shall be expressly set forth by specific reference to such filing.

Forward-Looking Statements

This filing contains or may contain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. The Company intends these forward-looking statements to be covered by the safe harbor provisions for such statements. These statements include, among other things, any predictions regarding the Company's prospects or future financial condition, earnings, revenues, expenses or other financial items, any statements concerning the Company's prospects or future operations, including management's plans or strategies and objectives therefor, expectations regarding the proposed acquisition, including synergies, time of closing, accretion and credit ratings, and any assumptions, expectations or beliefs underlying the foregoing. These statements can sometimes be identified by the use of forward looking words such as "may," "should," "will," "anticipate," "believe," "plan," "project," "estimate," "expect," "intend," "continue," "pro forma," "forecast" or other similar expressions or the negative thereof. All statements other than statements of historical facts in this news release or referred to in this news release are "forward-looking statements." These statements are based on current expectations, but are subject to certain inherent risks and uncertainties, many of which are difficult to predict and are beyond the control of the Company. Although the Company believes the expectations reflected in these forward-looking statements are based on reasonable assumptions, securityholders are cautioned that no assurance can be given that these expectations will prove correct. Actual results and developments may differ materially from the expectations expressed in or implied by these statements, based on various factors, including the effects of the global economic recession and its impact on sales volumes and pricing of our products, the Company's ability to collect its receivables from customers and ability to raise funds at reasonable rates; fluctuations in worldwide markets for corn and other commodities, and the associated risks of hedging against such fluctuations; fluctuations in the markets and prices for co-products, particularly corn oil; fluctuations in aggregate industry supply and market demand; the behavior of financial markets, including foreign currency fluctuations and fluctuations in interest and exchange rates; continued volatility and turmoil in the capital markets; the commercial and consumer credit environment; general political, economic, business, market and weather conditions in the various geographic regions and countries in which the Company or National Starch manufactures and/or sells products; future financial performance of major industries served by the Company or National Starch, including, without limitation, the food and beverage, pharmaceuticals, paper, corrugated, textile and brewing industries; energy costs and availability, freight and shipping costs, changes in regulatory controls regarding quotas, tariffs, duties, taxes and income tax rates; operating difficulties; boiler reliability; the Company's ability to effectively integrate acquired businesses; labor disputes; genetic and biotechnology issues; changing consumption preferences and trends; increased competitive and/or customer pressure in the cornrefining industry; and the outbreak or continuation of serious communicable disease or hostilities including acts of terrorism. Factors relating to the proposed acquisition that could cause actual results and developments to differ from expectations include: required regulatory approvals may not be obtained in a timely manner, if at all; the proposed acquisition may not be consummated in a timely manner or at all; the anticipated benefits of the proposed acquisition, including synergies, may not be realized; the integration of National Starch's operations with those of Corn Products may be materially delayed or may be more costly or difficult than expected, and we may be unable to maintain our current credit ratings. Forward-looking statements speak only as of the date on which they are made and the Company does not undertake any obligation to update any forward-looking statement to reflect events or circumstances after the date of the statement as a result of new information or future events or developments. If the Company does update or correct one or more of these statements, investors and others should not conclude that it will make additional updates or corrections. For a further description of these and other risks, see "Risk Factors" included in our Annual Report on Form 10-K for the year ended December 31, 2009 and subsequent reports on Forms 10-Q or 8-K.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

3

Exhibit

Exhibit

SIGNATURES				
	nt to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the reunto duly authorized.			
	CORN PRODUCTS INTERNATIONAL, INC.			
Date: June 21, 2	By: /s/ Cheryl K. Beebe Cheryl K. Beebe Vice President and Chief Financial Officer			
EXHIBIT INDEX Exhibit No. 2.1 Exhibit International Share and Business Sale Agreement, dated as of June 19, 2010, between Akzo Nobel N.V. and Corn Products				
10.1	International, Inc. Commitment Letter from JPMorgan Chase Bank, N.A. and J.P. Morgan Securities, dated June 19, 2010			

6

* Certain schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K. The registrant agrees to furnish

4

 $Commitment\ Letter\ from\ JPMorgan\ Chase\ Bank,\ N.A.\ and\ J.P.\ Morgan\ Securities,\ dated\ June\ 19,\ 2010$

supplementally a copy of any omitted schedule to the Securities and Exchange Commission upon request.

10.1

99.1

99.1

Press Release dated June 21, 2010

Press Release dated June 21, 2010

Confidential

Execution version

International Share and Business Sale Agreement

relating to

National Starch ("Specialty Starches")

between

Akzo Nobel N.V.

Corn Products International, Inc.

dated 19 June 2010

Confidential

Execution version

Contents

Clause

	Clause	Page
1	INTERPRETATION	6
1.1	Definitions	6
1.2	References to persons and companies	7
1.3	Headings and references to Clauses, Schedules, Parts and Paragraphs	7
1.4	Seller's Group, Share / Business Sellers, Purchaser's Group, Share / Business Purchasers	7
1.5	Other references	7
1.6	Information	7
1.7	Legal terms	7
2	SALE AND PURCHASE	8
2.1	Shares; US Interests; Businesses	8
2.2	Employees and Retirement Benefit arrangements	8
2.3	Nomination of other purchasers	9
2.4	Singular liability and right to claim	9
2.5	Local Transfer Documents	9
3	CONSIDERATION	10
3.1	Purchase Price	10
3.2	Payments at Closing	11
3.3	Allocation	11
3.4	VAT	11
3.5	Adjustment to Purchase Price	11
4	CONDITIONS PRECEDENT	11
4.1	Conditions	11
4.2	Responsibility for satisfaction	13
4.3	(Non-)Satisfaction/Waiver	14
5	PRE-CLOSING COVENANTS	15
5.1	Conduct of business	15
5.2	Access to Operations	16
5.3	Intra-group agreements	16
5.4	Estimated Statements	17
5.6	Cooperation in respect of Financing	18
6	CLOSING	19
6.1	Date and place	19
6.2	Payment of Estimated Purchase Price and Closing events	20
6.3	Repayment of Estimated Intra-Group Indebtedness	20

Confidential

Execution version

6.4	Breach of Closing obligations	21
6.5	Deferred Closing	21
7	POST-CLOSING ADJUSTMENTS	23
7.1	Working Capital Statements and Net Debt Statements	23
7.2	Adjustments to Purchase Price and repayment of Intra-Group Indebtedness	25
8	POST-CLOSING OBLIGATIONS	27
8.1	Indemnity against Group Liabilities	27
8.3	Release of Guarantees	27
8.4	Trade receivables/payables; Post-Closing receipts	28
8.5	Use of names	28
8.6	Retention of records	28
8.7	Insurance	29
8.8	Tax Indemnity	30
8.9	Intellectual Property Matters	30
8.10	Hive-out Matters	30
8.11	Payment of Employee bonuses related to the Transaction	31
9	WARRANTIES	31
9.1	Seller's Warranties	31
9.2	Disclosure	32
9.3	Liability for breach	33
9.4	Purchaser's Warranties	33
10	LIMITATION OF LIABILITY	34
10.1	Time limitation	34
10.2	Minimum claims	34
10.3	Aggregate minimum claims	34
10.4	Maximum liability	34
10.5	Provisions	35
10.6	Matters arising after Signing / Effective Time	35
10.7	Insurance	35
10.8	Net financial benefit	36
10.9	Mitigation of Losses	36
10.10	Right to recover	36
10.11	Tax indemnity and Tax warranty claims	37
11	CLAIMS	37
11.1	Notification of potential claims	37
11.2	Notification of claims	38
11.3	Commencement of proceedings	38
	3	

Confidential

Execution version

11.4	Investigation	38
11.5	Procedure for third party claims	38
11.6	Tax indemnity and Tax warranty claims	40
12	RESTRICTIONS	40
12.1	Restrictions on Seller	40
12.2	Restriction on Purchaser	41
12.3	Specific performance	41
13	CONFIDENTIALITY	41
13.1	Announcements	41
13.2	Confidentiality undertaking	41
14	MISCELLANEOUS	43
14.1	Further assurances	43
14.2	Whole agreement	43
14.3	No assignment	43
14.4	Waiver	43
14.5	Variation	43
14.6	Third party rights	43
14.7	Rescission	44
14.8	Method of payment	44
14.9	Costs and transfer Taxes	44
14.10	Interest	44
14.11	Notices	45
14.12	Invalidity	45
14.13	Counterparts	46

14.14 Notary 46 14.15 Dispute resolution 46 Governing law 14.16 46

Schedules

Schedule 1 **Definitions**

Schedule 2 Group Companies, Businesses

Schedule 3 **Business Properties**

Schedule 4 Guarantees

Schedule 5 **Intra-Group Retained Agreements**

Schedule 6 **Employees**

Schedule 7 Benefit Arrangements

4

Confidential

Execution version

Schedule 8 Closing

Schedule 9 **Local Transfer Documents**

Schedule 10 IP Licence

Allocation and Base Working Capital Schedule 11

VAT Schedule 12

Schedule 13 Closing statements and Reporting Accountants

Schedule 14 Tax and Intellectual Property Schedule 15 Warranties and Disclosure

Schedule 16 Accounts

Schedule 17 Transitional Services Agreement

Conduct of Business Schedule 18 Schedule 19 Commitment Letter Schedule 20 Senior Employees

5

Confidential

Execution version

International Share and Business Sale Agreement

THIS AGREEMENT IS MADE BETWEEN:

Akzo Nobel N.V., a public limited liability company incorporated under the laws of the Netherlands, with corporate seat in Amsterdam, and whose address is at Strawinskylaan 2555, Amsterdam, the Netherlands ("Seller"),

and

Corn Products International, Inc., a corporation organized and existing under the law of Delaware, with its registered office and principal place of business at 5 Westbrook Corporate Center, Westchester, Illinois 60154, United States of America ("Purchaser").

WHEREAS:

- Seller wishes to sell and procure the transfer of the Group as a going concern to Purchaser, who wishes to acquire the same, on and subject to the terms and conditions set out in this Agreement.
- The Group conducts the business of the research and development into and production, marketing, sale and distribution of specialty starches on a worldwide basis, principally focused on the food industry and also serving papermaking markets and the supply of industrial starch applications.
- Seller and Purchaser entered into the Confidentiality Agreement, pursuant to which certain confidential information relating to the Group was made available to Purchaser and its Representatives.
- Seller's board of management and supervisory board, and Purchaser's board of directors have given their approval with respect to the entering into of this Agreement and the transactions contemplated hereby.

IT IS AGREED AS FOLLOWS:

INTERPRETATION

In this Agreement, unless the context otherwise requires, the provisions in this Clause 1 apply throughout:

1.1 Definitions

Capitalised words, including those used in the preamble of this Agreement, shall have the meanings set out in Schedule 1.

6

Confidential

Execution version

1.2 References to persons and companies

References to:

- 1.2.1 a person include any individual, company or unincorporated association (whether or not having separate legal personality); and
- 1.2.2 a company include any company, corporation, limited liability company, partnership or any legal entity, wherever incorporated or organised.

1.3 Headings and references to Clauses, Schedules, Parts and Paragraphs

- 1.3.1 Headings have been inserted for convenience of reference only and do not affect the interpretation of any of the provisions of this Agreement.
- 1.3.2 A reference in this Agreement to a Clause or Schedule is to the relevant Clause of or Schedule to this Agreement; to a Part is to the relevant Part of the relevant Schedule; and to a Paragraph is to the relevant Paragraph of (the relevant Part of) the relevant Schedule.

1.4 Seller's Group, Share / Business Sellers, Purchaser's Group, Share / Business Purchasers

- 1.4.1 Any reference in this Agreement to the term "Seller's Group shall", shall be interpreted as meaning that Seller shall and/or shall procure that the relevant other members of Seller's Group shall perform the relevant obligation.
- 1.4.2 Any reference in this Agreement to a liability or obligation of a Share Seller, the US Seller or Business Seller shall be deemed to include an obligation on the part of Seller to procure that the relevant liability is discharged or obligation is performed, on and subject to the terms and conditions set out in this Agreement.
- 1.4.3 Any reference in this Agreement to the term "**Purchaser's Group shall**", shall be interpreted as meaning that Purchaser shall and/or shall procure that the relevant other members of Purchaser's Group shall perform the relevant obligation.
- 1.4.4 Any reference in this Agreement to a liability or obligation of a Share Purchaser or Business Purchaser shall be deemed to include an obligation on the part of Purchaser to procure that the relevant liability is discharged or obligation is performed, on and subject to the terms and conditions set out in this Agreement.

1.5 Other references

- 1.5.1 Whenever used in this Agreement, the words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation".
- 1.5.2 Any reference in this Agreement to any gender shall include all genders, and words importing the singular shall include the plural and vice versa.

1.6 Information

References to books, records or other information include books, records or other information in any form including paper, electronically stored data, magnetic media, film and microfilm.

1.7 Legal terms

In respect of any jurisdiction other than the Netherlands, a reference to any Netherlands legal

7

Confidential

Execution version

term shall be construed as a reference to the term or concept which most nearly corresponds to it in that jurisdiction.

2 SALE AND PURCHASE

2.1 Shares; US Interests; Businesses

- 2.1.1 On and subject to the terms and conditions of this Agreement, Seller hereby, itself and on behalf of the Share Seller, the US Seller and the Business Sellers, agrees to sell the Group to Purchaser, which hereby, itself and on behalf of the Share Purchasers and the Business Purchasers, agrees to purchase the Group, which sale comprises all of the right, title and interest as at Closing of Seller, the Share Seller, the US Seller and the Business Sellers in and to:
 - (a) the Shares;
 - *(b)* the US Interests;
 - (c) the Moveable Assets;
 - (*d*) the Business Inventory;
 - (e) the Business Receivables;
 - (f) the Business Properties (on the terms set out in Schedule 3 (Part 3);
 - *(g)* the Business Contracts (on the terms set out in Schedule 8 (Part 2);
 - (h) the Business Permits (to the extent that same can lawfully be assigned);
 - (i) the Business Intellectual Property;
 - (*j*) the Claims (to the extent that same can lawfully be assigned);
 - (k) the Business Books and Records; and
 - the Goodwill;

which, in the case of Clauses 2.1.1(a) and 2.1.1(b), shall be free and clear from all Encumbrances and, in the case of Clauses 2.1.1(c) through 2.1.1(l), shall be free and clear from all Encumbrances other than Permitted Encumbrances.

- 2.1.2 Any rights and assets of any member of Seller's Group (excluding the Group Companies) which are not expressly included in Clause 2.1.1 are excluded from the sale and transfer of the Group pursuant to this Agreement.
- 2.1.3 Purchaser shall procure the acceptance, as of Closing, by the relevant Business Purchasers of the transfer and the assumption, due and punctual payment, satisfaction, discharge, performance or fulfilment by the relevant Business Purchasers of all Business Liabilities.

2.2 Employees and Retirement Benefit arrangements

2.2.1 Save as provided in Schedule 7, the provisions of Schedule 6 shall apply in respect of the Employees.

8

Confidential

Execution version

2.2.2 The provisions of Schedule 7 shall apply in respect of the Retirement Benefit arrangements relating to Relevant Participants.

2.3 Nomination of other purchasers

- 2.3.1 Purchaser has nominated certain wholly owned members of Purchaser's Group to purchase the Shares, US Interests and Businesses, as set out in Schedule 2 (Part 1) and Schedule 2 (Part 2). Subject to Clause 2.3.2, Purchaser shall be entitled to nominate, at its own cost, by notice to Seller delivered no later than 5 (five) Business Days prior to the Closing Date, one or more other purchasers to purchase the Shares and US Interests, or any part thereof, and/or one or more of the Businesses (and to assume the corresponding Business Liabilities). Schedule 2 (Part 1) and Schedule 2 (Part 2) shall be deemed to be amended accordingly so as to give effect to any such nomination.
- 2.3.2 Purchaser may only nominate a wholly owned member of Purchaser's Group as a purchaser under this Agreement.

2.4 Singular liability and right to claim

- 2.4.1 None of the Share Seller, US Seller or Business Sellers, with the exception of Seller, shall have any liability under this Agreement, it being agreed that Seller shall be fully liable under this Agreement for any breach thereof by, and any obligation of, any of the Share Seller, US Seller or Business Sellers. Only Purchaser may seek recourse against Seller for any such breach or failure to perform any such obligation by Seller, the Share Seller, US Seller or a Business Seller under this Agreement.
- 2.4.2 None of the Share Purchasers or Business Purchasers, with the exception of Purchaser, shall have any liability under this Agreement, it being agreed that Purchaser shall be fully liable under this Agreement for any breach thereof by, and any obligation of, any of the Share Purchasers or Business Purchasers. Only Seller may seek recourse against Purchaser for any such breach or failure to perform any such obligation by Purchaser, a Share Purchaser or a Business Purchaser under this Agreement.

2.4.3 Notwithstanding the provisions of Clauses 2.4.1 and 2.4.2, a third party stipulation (*'derdenbeding'*) expressly identified as such in this Agreement, shall be for the benefit of and enforceable by the relevant third parties, provided that the Parties exclude the applicability of the articles 6:254, 6:255 and 6:256 of the Netherlands Civil Code.

2.5 Local Transfer Documents

2.5.1 At Closing and at any applicable Deferred Closing, Seller and Purchaser shall execute or, as the case may be, shall procure the execution by the relevant members of Seller's Group and the relevant members of Purchaser's Group of such agreements, deeds, transfers, conveyances and other documents (in accordance with the applicable local Law and otherwise as may be agreed between Seller and Purchaser) to implement the transfer of the Shares, the US Interests, the Businesses and the Business Assets at such Closing or Deferred Closing, said documents to be substantially in the form of the documents forming part of Schedule 9 or in such other form as the Parties may mutually agree prior to such Closing or Deferred Closing (collectively, the "Local Transfer Documents" and

9

Confidential

Execution version

individually, a "Local Transfer Document").

- 2.5.2 To the extent that the provisions of a Local Transfer Document are inconsistent with other provisions of this Agreement (which term, for purposes of this Clause 2.5.2 shall exclude all Local Transfer Documents):
 - (a) the said other provisions of this Agreement shall prevail; and
 - (b) Seller and Purchaser shall procure that, so far as permissible under the Laws of the relevant jurisdiction, the provisions of the relevant Local Transfer Document are adjusted to the extent necessary to give effect to the provisions of this Agreement (or, to the extent this is not permissible, Seller shall indemnify, defend and hold harmless Purchaser against all Losses suffered by any member of Purchaser's Group or, as the case may be, Purchaser shall indemnify, defend and hold harmless Seller against all Losses suffered by any member of Seller's Group, in either case through or arising from the inconsistency between the Local Transfer Document and the other provisions of this Agreement).
- 2.5.3 To the extent that the provisions of a Local Transfer Document are inconsistent with the provisions of any other Local Transfer Document in the same jurisdiction, the provisions of the Local Transfer Document last executed shall prevail, provided that if the relevant Local Transfer Documents are executed on the same day, the provisions of the Local Transfer Document with respect to such jurisdiction appearing first on Schedule 9 shall prevail.
- 2.5.4 If there is an adjustment to the payment made on account of the Purchase Price under Clause 7 which relates to a part of the Group which is the subject of a Local Transfer Document, then, if required to implement the adjustment and so far as permissible under the Laws of the relevant jurisdiction, Seller's Group shall and Purchaser's Group shall enter into a supplemental agreement reflecting such adjustment and the allocation of such adjustment.
- 2.5.5 The Parties shall not, and the Parties shall cause their respective Affiliates not to, bring any claim against the other Party or any of its respective Affiliates, as the case may be, in respect of or based upon any Local Transfer Document. The Local Transfer Documents shall not limit any rights or remedies or modify any limitations of liability of the Parties contained in this Agreement.

3 CONSIDERATION

3.1 Purchase Price

The purchase price for the Group (the "Purchase Price") shall be an amount equal to the aggregate of the following amounts:

- 3.1.1 the Offer Value; plus
- 3.1.2 Working Capital Adjustment for the Group; less
- 3.1.3 Net Debt for the Group; less

10

Confidential

Execution version

3.1.4 if applicable, the Final DBO Adjustment Amount;

subject to adjustment as set out in Clause 3.5.

3.2 Payments at Closing

At Closing, an estimate of the Purchase Price (the "Estimated Purchase Price") equal to the aggregate of the following amounts shall be paid in terms of Clause 6.2.1:

- 3.2.1 the Offer Value; plus
- 3.2.2 Estimated Working Capital Adjustment for the Group; less
- 3.2.3 Estimated Net Debt for the Group.

3.3 Allocation

The Estimated Purchase Price and the Purchase Price shall be allocated in accordance with Schedule 11 (Part 1) and Seller's Group shall and Purchaser's Group shall file all Tax Returns and make all Tax payments in a manner consistent with such allocation.

3.4 VAT

The consideration given under this Agreement in respect of the sale of the Businesses, the US Interests and the Shares is exclusive of any VAT properly chargeable thereon in respect of which the provisions of Schedule 12 shall apply.

3.5 Adjustment to Purchase Price

If any payment is made by Seller to Purchaser or by Purchaser to Seller in respect of any claim (i) for any breach of this Agreement (including, for the avoidance of doubt, a breach of a Seller's Warranty), (ii) pursuant to an indemnity under this Agreement or (iii) pursuant to Clause 7, to the extent permitted by applicable Law the payment shall be made by way of adjustment of that portion of the Purchase Price allocated in Schedule 11 to the particular category of Business Asset or the Shares to which the payment and/or claim relates under this Agreement and the Purchase Price shall be deemed to be adjusted by the amount of such payment, provided that to the extent that such an adjustment (i) is less than USD 500,000 (five hundred thousand US dollars) or (ii) would result in the portion of the Purchase Price allocated in Schedule 11 to the particular category of Business Asset being less than USD 1.00 (one US dollar), then, in each case, the amount of such adjustment shall instead be applied as an adjustment to the portion of the Purchase Price allocated in Schedule 11 to the Shares.

4 CONDITIONS PRECEDENT

4.1 Conditions

The respective obligations of the Parties to perform the Closing shall be conditional upon satisfaction or waiver of the following Conditions Precedent:

- *4.1.1* Either
 - (a) the European Commission:

11

Confidential

Execution version

- (i) decides that the Transaction does not give rise to a concentration falling within the scope of the European Council Regulation (EEC) 139/2004 concerning the control of concentrations between undertakings (as amended) (the "EU Regulation");
- (ii) whether or not conditional, declares the Transaction compatible with the common market under Article 6(1)(b) or, if the European Commission has initiated proceedings pursuant to Article 6(1)(c), under Article 8(1) or 8(2) of the EU Regulation, or is deemed to have done so under Article 10(6) of the EU Regulation; or
- (iii) refers the Transaction or any part thereof to the authorities of one or more Member States under Article 9(3) of the EU Regulation or is deemed to have done so under Article 9(5) of the EU Regulation:
 - (A) and each such authority grants a clearance, whether or not conditional, in respect of all those parts of the Transaction which are referred to it, or is deemed to have granted such a clearance; and
 - (*B*) in respect of all those parts not referred to any authority of any Member State, the requirements of Clause 4.1.1(a)(ii) are satisfied;

or

- (b) in the event that the Transaction does not constitute a concentration as referred to in Clause 4.1.1(a)(i), and only to the extent that such clearance is required, under mandatory Law, for Closing to take place, the relevant authorities of each of the relevant Member States grant clearance, whether or not conditional, under the Competition Laws for such part of the Transaction which falls under their jurisdiction.
- 4.1.2 In the event that under mandatory Law, in order for Closing to take place, the Transaction requires authorisation, approval or clearance from any Governmental Authority of any Non-EU Country under the Competition Laws of such Non-EU Country, including under the HSR Act, either

- (a) such Governmental Authority grants the required authorisation, approval or clearance, whether or not conditional, or (b) the required authorisation, approval or clearance, is obtained by the expiry or termination of the relevant mandatory waiting period.
- 4.1.3 No Governmental Authority that has jurisdiction enacts, issues, promulgates, enforces or enters any Law, judgment, decree or injunction (whether temporary, preliminary or permanent) that is in effect immediately prior to Closing and has the effect of enjoining, restraining, preventing, prohibiting or making illegal this Agreement or the completion of the Transaction.
- 4.1.4 Seller is not in breach of its pre-Closing covenants set out in Clauses 5.1 through 5.4, except where such breach does not have a Group Material Adverse Effect.
- 4.1.5 No Group Material Adverse Effect occurs between the Signing Date and the first Closing.

12

Confidential

Execution version

4.2 Responsibility for satisfaction

- 4.2.1 Save as provided in Clause 4.2.2, each of the Parties shall use its reasonable best efforts to ensure satisfaction of and compliance with all of the Conditions Precedent.
- 4.2.2 Notwithstanding Clause 4.2.1, Purchaser shall:
 - (a) as soon as practicable, and in any event no later than twenty (20) Business Days after the Signing Date, prepare and file with the Governmental Authorities the notices and applications necessary to (i) satisfy the Conditions Precedent set out in Clauses 4.1.1 and 4.1.2 and (ii) obtain all other Anti-trust Approvals, provided that Seller has the right to approve the same (such approval not to be unreasonably withheld, conditioned or delayed), and provided further that Seller and Purchaser together file, with the United States Federal Trade Commission and United States Department of Justice the notification and report form, if any, required for the Transaction under the HSR Act;
 - (b) supply as promptly as practicable any additional information and documentary material that may be requested by any Governmental Authority in connection with (i) any of the Conditions Precedent set out in Clauses 4.1.1 and 4.1.2 or (ii) any other Anti-trust Approvals, provided that, with respect to any written submission, information or documentary materials, Seller has the right to approve the same (such approval not to be unreasonably withheld, conditioned or delayed); and
 - take promptly any and all steps necessary to avoid or eliminate each and every impediment to, and procure as soon as practicable, the fulfilment of the Conditions Precedent set out in Clauses 4.1.1 and 4.1.2 and the obtaining of all other Anti-trust Approvals by no later than 31 December 2010 (the "Outside Date") or, with respect to any Non-Material Business or Non-Material Group Company, 30 June 2011, including by (i) taking or giving to the competent Governmental Authorities a binding undertaking to take any action that may be necessary or appropriate in order to obtain clearance of the Transaction (including by agreeing to sell, lease, license or otherwise dispose of, or to hold separate pending such disposition, and promptly to effect the sale, lease, license, disposal and holding separate of, assets, rights, product lines, licenses, categories of any assets or businesses or other operations, or interests therein, of Purchaser or its subsidiaries, including the shares, properties and all other assets to be acquired (directly or indirectly) by Purchaser hereunder, and the entry into agreements with, and submission to orders of, the relevant Governmental Authority giving effect thereto, that may be required by any relevant Governmental Authority) or (ii) duly and promptly complying with any condition that any relevant Governmental Authority may impose or require to approve the consummation of the Transaction and that cannot be resisted in accordance with the terms of this Agreement. For the avoidance of doubt, Purchaser shall take any and all actions necessary in order to ensure that (x) no requirement for a waiver, consent or approval of any Governmental Authority with respect to any Competition Laws, (y) no decree, decision, judgment, injunction, temporary restraining order or any other order

13

Confidential

Execution version

in any suit or proceeding, and (z) no other matter relating to any antitrust or competition Law, would preclude consummation of the Transaction by the Outside Date.

- 4.2.3 Purchaser shall bear all filing fees and other costs (excluding Seller's attorneys' fees) incurred in relation to any Anti-trust Approval. Purchaser shall also bear all costs, penalties and fines resulting from not seeking to obtain Anti-trust Approval in any jurisdiction where it is determined that filing should have taken place, provided that Seller shall be responsible for any such costs, penalties and fines to the extent that information provided by it in connection with the Parties' analysis of filing obligations in any such jurisdiction was inaccurate and resulted in such costs, penalties and fines.
- 4.2.4 Without prejudice to Clause 4.2.2, Seller and Purchaser shall:

- (a) promptly cooperate with and provide all necessary information and assistance required by any Governmental Authority in connection with the Conditions Precedent set out in Clauses 4.1.1 and 4.1.2 and in connection with any other Anti-trust Approvals, upon being requested to do so by the other Party; and
- (*b*) promptly inform the other Party of any communication received from, or given by it to, any Governmental Authority with respect to any of the Conditions Precedent set out in Clauses 4.1.1 and 4.1.2 and in connection with any other Anti-trust Approvals.
- 4.2.5 Notwithstanding the foregoing provisions of this Clause 4.2, except in accordance with Clause 6.5, neither Seller nor any other member of Seller's Group shall have any obligation to hold separate or divest any assets, rights, product lines, licenses, categories of assets or businesses or other operation, or interests therein, of Seller or any other member of Seller's Group.
- 4.2.6 Without prejudice to Clauses 4.2.2(c) and 4.2.4, in the event that any administrative or judicial action or proceeding is instituted (or threatened to be instituted) by a Governmental Authority or any other person challenging the Transaction (or any part thereof), each Party shall co-operate in all respects with the other Party and use its reasonable best efforts to defend, contest and resist any such action or proceeding and to have vacated, lifted, reversed or overturned any order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts the consummation of the Transaction (or any part thereof).

4.3 (Non-)Satisfaction/Waiver

- *4.3.1* As soon as reasonably practicable, but in any event within two (2) Business Days of becoming aware of the same:
 - (a) each Party shall give notice to the other Party of the satisfaction of any of the Conditions Precedent set out in Clause 4.1.1 or 4.1.2; and
 - (b) each Party shall give notice to the other Party of any development that would be reasonably likely to result in the non-satisfaction of any of the Conditions Precedent.

14

Confidential

Execution version

- 4.3.2 The Conditions Precedent set out in Clauses 4.1.1, 4.1.2 and 4.1.3 may only be waived if waived by both Parties, and the Conditions Precedent set out in Clauses 4.1.4 and 4.1.5 may only be enforced or waived by Purchaser, provided that the occurrence of a Closing shall be deemed to be a waiver of each Condition Precedent other than any Deferred Condition.
- 4.3.3 Subject to Clause 6.5, if:
 - (a) the Conditions Precedent set out in Clauses 4.1.1 and 4.1.2 are not satisfied or waived with respect to the USA, Germany, Canada, Thailand or the UK on or before the Outside Date, then Seller may in its sole discretion,
 - (b) the Condition Precedent set out in Clause 4.1.3 is not satisfied or waived on or before the Outside Date, then each Party may in its sole discretion,
 - (c) the Condition Precedent set out in Clause 4.1.4 is not satisfied or waived on or before the Outside Date, then Purchaser may in its sole discretion,
 - (*d*) the Condition Precedent set out in Clause 4.1.5 is not satisfied or waived at any time before the Outside Date, then Purchaser may in its sole discretion, or
 - (e) any Governmental Authority of competent jurisdiction enacts, issues, promulgates, enforces or enters any non-appealable judgement, decree or injunction permanently enjoining, restraining, preventing or otherwise prohibiting or making illegal this Agreement or the completion of the Transaction (or any part thereof), then each Party may in its sole discretion,

in addition to and without prejudice to all other rights or remedies available, terminate this Agreement by notice to the other Party, provided that the right to terminate this Agreement pursuant to Clause 4.3.3(a), (b) or (e) shall not be available to any Party whose failure to fulfil any of its obligations under this Agreement has been the cause of the event or condition giving rise to such termination right. In addition, the Parties may terminate this Agreement by mutual written consent. Clauses 1, 13 and 14.2 through 14.16 shall survive any termination of this Agreement.

5 PRE-CLOSING COVENANTS

5.1 Conduct of business

Seller undertakes to procure that between Signing and Closing the relevant members of the Seller's Group, and between the Closing Date and any Deferred Closing, the applicable Business Seller (with respect to the relevant Non-Material Business) and/or each relevant Non-Material Group Company, shall carry on the Operations as a going concern in the ordinary course, consistent with past practice, and, in relation thereto, shall, to the extent permitted by Law (including Competition Laws), comply with the specific limitations and prohibitions set out in Schedule 18 (Part 1). Notwithstanding these specific limitations and prohibitions, the relevant members of Seller's Group shall have the right to take the actions described in Schedule 18 (Part 2).

Confidential

Execution version

5.2 Access to Operations

- 5.2.1 Subject to Clause 5.2.2, Seller shall procure that, until the Closing Date or, if applicable and earlier, termination of this Agreement, the Business Sellers and the Group Companies give Purchaser and its Representatives, upon reasonable notice to Seller, reasonable access during normal working hours to the premises, Employees, accountants, Properties and books and records of or relating to the Operations to the extent necessary for Purchaser to ensure the transfer of the Group in an efficient and timely manner at Closing and to begin transition planning without unreasonable interference or interruption, and to the extent permitted by competition Laws, Seller shall provide to Purchaser and its Representatives copies of any A.T. Kearney work product with respect to the Operations.
- 5.2.2 Purchaser shall not have any right to any such access in any manner or to any extent as:
 - (a) would be contrary to any applicable Law (including Competition Laws); or
 - (b) would reasonably be likely to cause undue disruption to the relevant Operations or their management.

5.3 Intra-group agreements

- 5.3.1 Seller covenants that, except as expressly set out in this Agreement or otherwise agreed by the Parties, all existing agreements and arrangements (except for the Intra-Group Retained Agreements) between one or more members of Seller's Group (excluding the Group Companies) on the one hand and one or more Group Companies on the other hand shall be, prior to or at Closing, terminated or amended to remove any such Group Company as a party thereto, in each case without any direct or contingent liability or obligation to such Group Company, except to the extent reflected in the calculation of Net Debt and as contemplated by Clause 7.2.3.
- 5.3.2 Except as otherwise provided in this Agreement, subject to Closing, neither Seller nor Purchaser shall knowingly take or omit to take any action, which would reasonably be likely to result in any member of Seller's Group or Purchaser's Group, as applicable, being relieved of or unable to perform its obligations under any Intra-Group Retained Agreement.
- 5.3.3 Prior to the Closing Date, Seller and Purchaser shall negotiate in good faith to review the Surface Chemistry Agreement and seek to agree a mutually acceptable amendment to it. If such an amendment to the Surface Chemistry Agreement is not agreed by the Closing Date, then Seller and Purchaser shall cause their relevant Affiliates to execute an amendment agreement to the Surface Chemistry Agreement to the effect that:
 - (a) neither Seller nor any Affiliate of Seller shall have any right under the Surface Chemistry Agreement to restrict in any way any member of the Purchaser's Group (including the US Company) from directly or indirectly (i) developing, manufacturing, selling, distributing or otherwise supplying Specialty Starches for use for or to any person outside the Seller's Group for use in any Customer Field or (ii) conducting any research with any person outside the Seller's Group that produces products in any Customer Field;
 - (b) neither Purchaser nor any Affiliate of Purchaser shall have any right under the

16

Confidential

Execution version

Surface Chemistry Agreement to directly or indirectly (i) require in any way any member of the Seller's Group (including Akzo Nobel Surface Chemistry LLC) to purchase Specialty Starch from any member of the Purchaser's Group (including the US Company) or (ii) restrict any member of the Seller's Group (including Akzo Nobel Surface Chemistry LLC) from conducting any research in the Customer Field with any person that produces Specialty Starches; and

(c) the obligation of the relevant member of the Purchaser's Group (including the US Company) to supply the relevant products under the Surface Chemistry Agreement in any one consecutive period of 12 (twelve) months shall not exceed 110% (one hundred and ten percent) of the volume of products delivered in the 12 (twelve) months immediately preceding the Closing Date.

5.4 Estimated Statements

- 5.4.1 Subject to Clause 6.5, no later than 3 (three) Business Days prior to the date set pursuant to Clause 6.1.1 for Closing, Seller shall deliver to Purchaser the Estimated Net Debt Statements and the Estimated Working Capital Statements for each of the Company (including all other Group Companies other than the US Company and its Subsidiary) and the US Company (including its Subsidiary) and each relevant Business.
- 5.4.2 Each Estimated Working Capital Statement shall be drawn up in the form set out in Schedule 13 (Part 1) and each Estimated Net Debt Statement shall be drawn up in the form set out in Schedule 13 (Part 2).
- 5.4.3 Any amounts included in the determination of any Estimated Net Debt Statement or any Estimated Working Capital Statement in currencies other than USD shall be translated into USD at the exchange rates as reasonably determined by Seller at the date of drawing up the Estimated Net Debt Statements.

5.4.4 The Estimated Net Debt Statements and the Estimated Working Capital Statements shall for purposes of the calculation and allocation of the Estimated Purchase Price not be subject to review by Purchaser and shall be final and binding on the Parties.

5.5 Financial Statements

5.5.1 In the period prior to the Closing Date, subject to Clause 5.5.2, (a) Seller shall deliver to Purchaser, in respect of each fiscal quarter that falls between April 1, 2010 and the Closing Date, within 9 weeks after the end of such fiscal quarter, an unaudited statement of acquired assets and liabilities of the Group as of the end of such fiscal quarter, and related unaudited statements of operating activities and identifiable cash flows of the Group for such quarter and for the period from 31 December 2009 through the end of such quarter, prepared in accordance with IFRS and consistent with the preparation of the Accounts (the "Financial Statements"), which Financial Statements, for the avoidance of doubt, shall be limited to the aforesaid three primary statements, reviewed but not audited, and shall not be required to contain any other information normally contained in a report prepared under IAS 34 other than customary footnotes necessary for interim financial statements; and (b) at Purchaser's cost and expense, Seller shall reasonably cooperate and shall procure that

17

Confidential

Execution version

the Group's management reasonably cooperates with Purchaser in a timely manner as reasonably requested by Purchaser in connection with Purchaser's preparation of pro forma financial information in respect of the Group as required by Regulation S-X under the Securities Act and the filing of the necessary financial statements and pro forma financial information with the SEC, including using reasonable efforts to obtain from Seller's accountants relevant consent letters necessary in connection with the foregoing.

5.5.2 Except as expressly set out in the foregoing provisions of this Clause 5.5: (a) filing financial statements of the Group with the SEC pursuant to the requirements of Regulations S-K and S-X or other requirements is the sole responsibility of Purchaser; (b) in no event shall Seller or any Group Company or any of their Affiliates or Representatives be liable to Purchaser for any failure of the Financial Statements to comply with the requirements of Regulations S-K and S-X or to be prepared prior to expiry of a period of 71 (seventy-one) days after the date on which the Purchaser's Form 8-K reporting the Closing must be filed with the SEC, except to the extent that any such failure directly results from a breach by Seller of its obligations under Clause 5.5.1, in which event, as Purchaser's sole and exclusive remedy, Purchaser shall have the right to repayment by Seller of all amounts paid by Purchaser to Seller under Clause 5.5.1; (c) Purchaser shall reimburse Seller or any Group Company, promptly upon request, for any out-of-pocket expenses incurred by Seller or any Group Company or any of their Affiliates or Representatives relating to their actions taken pursuant to and in accordance with this Clause 5.5, including out-of-pocket expenses incurred in connection with the preparation of the Financial Statements; and (d) except as set out in Clause 6.1.1, Purchaser shall not be entitled to, and shall not directly or indirectly attempt to, delay Closing pending the provision of any of the financial statements contemplated by any of the provisions of this Clause 5.5.

5.6 Cooperation in respect of Financing

- 5.6.1 Subject to Clause 5.6.2, at the request, cost and expense of Purchaser, prior to the Closing Seller's Group shall cooperate reasonably, and shall request its Representatives to reasonably cooperate, in connection with the financing to be obtained by Purchaser under the Commitment Letter and/or any equity, debt or other securities issuance to refinance or replace all or any part of such financing (together, the "Financing"), in each case for the purchase of the Group pursuant to this Agreement (provided that such requested cooperation does not unreasonably interfere with the ongoing management or operations of the Group, or require travel or demands on the time of management or employees that in the view of Seller is unreasonably disruptive to the management or operations of the Group), including (a) furnishing pertinent information regarding the Group as may be reasonably requested by Purchaser, (b) participating in a reasonable number of meetings, presentations and sessions with rating agencies and (c) using reasonable efforts to obtain accountants' comfort letters and accountants' consents as reasonably requested by Purchaser.
- 5.6.2 The obligations of Seller under this Clause 5.6 shall be limited as follows: (i) arrangement of the Financing is the sole responsibility of Purchaser, (ii) Seller and its Representatives

18

Confidential

Execution version

shall not be required to enter into any agreements or incur any Liabilities in respect of the Financing, none of the Group Companies or their respective Affiliates and Representatives shall be required to incur any Liabilities in respect of the Financing prior to Closing, and Purchaser shall indemnify and hold harmless each of Seller, the Group Companies and their respective Representatives for and against any Losses arising from or relating to their involvement or cooperation in connection with the Financing and any information utilized in connection therewith, including under any securities Law, provided that the foregoing provisions of this Clause 5.6.2(ii) shall be without prejudice to any claim by Purchaser for breach of the Seller's Warranties or Seller's obligations under this Agreement, (iii) Purchaser shall reimburse Seller or any Group Company, promptly upon request, for any out-of-pocket expenses incurred by Seller or any Group Company or any of their Affiliates or Representatives relating to their cooperation in connection with the Financing; (iv) none of Seller, any Group Company or any Representative of them shall have any obligation to meet or participate in any meetings with any potential lender, investor or member of a banking syndicate, in a manner that would unreasonably interfere with the operation of Seller's or any Group Company's business, (v) nothing in this Clause 5.6 shall be deemed to permit or require the disclosure of information that would otherwise be prevented under Clause 13 (Confidentiality); (vi) obtaining the Financing, and cooperation by Seller, the Group Companies and their Representatives in obtaining the Financing, is not a condition to the

obligations of Purchaser hereunder and (vii) except as set out in Clause 6.1.1, Purchaser shall not be entitled to, and shall not directly or indirectly attempt to, delay Closing in order to draw amounts under any debt facility or to arrange any equity, debt or other securities issuance to refinance or replace all or any part of the bridge facility referred to in the Commitment Letter.

6 CLOSING

6.1 Date and place

- 6.1.1 Subject to Clause 6.5 and to the satisfaction (or waiver under Clause 4.3.2) of each of the Conditions Precedent, Closing shall take place at 11.00 CET on the date that is the later of (a) the fifth Business Day after satisfaction or waiver of each Condition Precedent (other than those Conditions Precedent that by their terms are to be satisfied at Closing, but subject to the satisfaction or waiver of those Conditions Precedent) and (b) the date that is 4 (four) weeks after Seller delivers the Financial Statements last due pursuant to Clause 5.5 (or such earlier date as may be specified in writing by Purchaser at least 2 (two) Business Days prior thereto), unless another date and time is agreed to by the Parties in writing.
- 6.1.2 Notwithstanding the period of 4 (four) weeks after the delivery by Seller of the Financial Statements pursuant to Clause 5.5 not having lapsed, at the election of Seller, the date for Closing shall be determined under Clause 6.1.1(a) (without regard to Clause 6.1.1(b)), provided that in such event, in the event that Purchaser, for purposes of paying to Seller the Closing Payment Amount, makes use of its bridge financing facility under the

19

Confidential

Execution version

Commitment Letter, Seller shall pay Purchaser at Closing an amount in cash equal to USD 14,800,000 (fourteen million eight hundred thousand US dollars). For the avoidance of doubt, amounts paid by Seller pursuant to this Clause 6.1.2 shall not be treated as adjustments to the Purchase Price pursuant to Clause 3.5, to the extent permitted by Law.

6.1.3 Closing shall take place in Amsterdam, the Netherlands, at the offices of Seller's Lawyers or at such other location as the Parties may agree to in writing.

6.2 Payment of Closing Payment Amount and Closing events

- 6.2.1 Subject to Clause 6.2.2, Purchaser shall pay the aggregate of the Estimated Purchase Price and the Estimated Intra-Group Indebtedness (the "Closing Payment Amount") to bank account number 69.32.13.876 (a designated account, 'kwaliteitsrekening') in the name of the offices of the Notary, on the Business Day before and with value per the Closing Date, which amount shall be held for Purchaser until execution of the notarial deed of transfer of the Shares under Paragraph 2 of Schedule 8 (Part 1), after which the Notary shall hold the Estimated Purchase Price on behalf of Seller and pay said amount in accordance with Seller's instructions.
- 6.2.2 In the event that Purchaser uses its Committed Financing Facility for the payment of the Estimated Purchase Price, then Purchaser shall procure that the Closing Payment Amount is paid to bank account number 69.32.13.876 (a designated account, 'kwaliteitsrekening'), in the name of the offices of the Notary before 11:00 CET on the Closing Date and with value per the Closing Date, which amount shall be held for Purchaser until execution of the notarial deed of transfer of the Shares under Paragraph 2 of Schedule 8 (Part 1), after which the Notary shall hold the Estimated Purchase Price on behalf of Seller and pay said amount in accordance with Seller's instructions.
- 6.2.3 At Closing, the Parties shall procure that those of the obligations set out in Schedule 8 (Part 1) for which they are respectively responsible, are performed for the transfer of the Shares, the US Interests, the Business Assets and the Business Liabilities, provided that if Seller does not deliver one of the certificates as set out in Paragraphs 1(a)(ii) and 1(a)(iii) of Schedule 8 (Part 1) and, if applicable, the undertaking described in Paragraph 1(a)(iii) of Schedule 8 (Part 1), Purchaser shall be relieved of its obligations to perform the Closing and shall be entitled, as its sole and exclusive remedy, by notice served on or after the date set for Closing in Clause 6.1, to terminate this Agreement (save that Clauses 1, 13 and 14.2 through 14.16 shall survive termination), in which event the Parties shall forthwith take all such action as is necessary to reverse any action already taken under Clause 6.2 or 6.3 or Schedule 8 (Part 1).

6.3 Repayment of Estimated Intra-Group Indebtedness

6.3.1 The payment of the Estimated Intra-Group Indebtedness as part of the Closing Payment Amount shall be deemed to be payment (a) by Purchaser (made on behalf of the relevant Group Companies) to Seller (received on behalf of the relevant members of Seller's Group (excluding the Group Companies)) of the Estimated Intra-Group Payables, and (b) by Seller (made on behalf of the relevant members of Seller's Group (excluding the Group

20

Confidential

Execution version

6.3.2 Notwithstanding anything to the contrary contained in this Agreement, to the extent that the sum of the Estimated Cash Balances and the Estimated Intra-Group Receivables (net of Estimated Intra-Group Payables) exceeds USD 50,000,000 (fifty million US dollars) in aggregate (such excess, the "Excess Cash Amount"), the payment obligation under this Agreement in respect of the Excess Cash Amount shall become due, without deduction or set-off (subject to the proviso set out below) 90 (ninety) days after the Closing Date (or such other date as may be agreed to in writing by the Parties), provided that the Excess Cash Amount shall be reduced by any out-of-pocket expenses reasonably incurred by the relevant members of Purchaser's Group and any Taxes incurred by the relevant members of Purchaser's Group in respect of the extraction of any portion of the Excess Cash Amount (it being understood, for the avoidance of doubt, that such Taxes shall include those Taxes incurred by the relevant members of Purchaser's Group in respect of the distribution of any portion of the Excess Cash Amount (i) from the relevant Group Company to the Company, (ii) from the Company to the relevant Share Purchaser and (iii) from the relevant Share Purchaser to Purchaser). Purchaser shall take reasonable steps to eliminate or reduce the Taxes described in the proviso to the preceding sentence and shall take into consideration any suggestions made by Seller in relation thereto

6.4 Breach of Closing obligations

Subject to Clause 6.5, if any Party breaches any material obligation in Clauses 6.2 or 6.3 or Schedule 8 (Part 1) in relation to Closing, Purchaser, in the case of breach by Seller, or Seller, in the case of breach by Purchaser, shall be entitled (in addition to and without prejudice to all other rights or remedies available, including the right to claim damages) by notice served on or after the Closing Date:

- 6.4.1 to terminate this Agreement (save that Clauses 1, 13 and 14.2 through 14.16, and this Clause 6.4.1 shall survive termination), in which event the Parties shall forthwith take all such action as is necessary to reverse any action already taken under Clause 6.2 or 6.3 or Schedule 8 (Part 1);
- 6.4.2 to effect Closing so far as practicable having regard to a default or defaults which have occurred; or
- 6.4.3 to set a new date for Closing (not being more than 20 (twenty) Business Days after the previously set date for Closing) in which case the provisions of Clauses 6.2, 6.3 and 6.4 and Schedule 8 (Part 1) shall apply to Closing as so deferred.

6.5 Deferred Closing

In the event that the Conditions Precedent have been satisfied or waived (other than those Conditions Precedent the satisfaction of which is to be determined at Closing), save for either of the Conditions Precedent set out in Clauses 4.1.1 and 4.1.2 (each a "**Deferred Condition**"), then, if such Deferred Condition relates to a jurisdiction other than the USA, Germany, Canada, Thailand or the UK (any such other jurisdiction, a "**Deferred Jurisdiction**"), then:

21

Confidential

Execution version

- 6.5.1 subject to Clause 6.5.2, Closing shall proceed in accordance with the terms of Clauses 6.1, 6.2 and 6.3 in respect of the Shares, US Interests and all other Businesses (other than those relating to the Deferred Jurisdictions) as if the Deferred Conditions were not Conditions Precedent to said Closing;
- 6.5.2 if a Deferred Jurisdiction relates to any Group Company outside the USA, Germany, Canada, Thailand or the UK (each, a "Non-Material Group Company"), Seller shall procure that, to the extent reasonably practicable, all of the shares of each such Non-Material Group Company are transferred directly or indirectly from the Company to another member of Seller's Group (excluding the Group Companies) (the "Interim Holdco") as soon as reasonably practicable and, at 11.00 CET on the fifth Business Day after completion of such transfer(s), Closing in respect of the Shares, US Interests and all other Businesses (other than those relating to such Deferred Jurisdiction) shall proceed in accordance with the terms of Clause 6.5.1;
- 6.5.3 notwithstanding there being one or more Deferred Jurisdictions, the full amount of the Estimated Purchase Price shall be paid at the first Closing;
- 6.5.4 Closing in respect of each Deferred Jurisdiction (each, a "**Deferred Closing**") shall take place at 11.00 CET on the first Business Day of the month following the month in which notification of the satisfaction or waiver of the Deferred Conditions relating to the Deferred Jurisdiction occurs and shall take place in Amsterdam, the Netherlands, at the offices of Seller's Lawyers, or at such other location, time or date as may be agreed between Seller and Purchaser and the Parties shall take the relevant actions described in Clauses 2.5.1 and 6.2.3;
- as soon as reasonably practicable after each Deferred Closing and subject to Paragraph 2.1(vii) of the Tax Indemnity, the Parties shall account to each other in respect of the Business outside of the USA, Germany, Canada, Thailand or the UK (each, a "Non-Material Business") and/or Non-Material Group Company to which the Deferred Closing relates in order to place each Party in the financial position (including in respect of Taxes, but excluding in respect of the time value of money) in which it would have been in respect thereof had the Deferred Closing occurred on the Closing Date, it being agreed that, subject to the occurrence of the Deferred Closing, the risk and benefit in respect of such Non-Material Business and/or Non-Material Group Company shall be deemed to have passed to Purchaser at the Effective Time;
- 6.5.6 without detracting from the Parties' obligations under Clause 4.2, if any of the Deferred Conditions has not been satisfied or waived on or before the date that is 6 (six) months after the Outside Date, Seller and Purchaser shall consult in good faith and appoint an independent trustee who shall be instructed to procure the sale of any affected assets (the "Non-Cleared Assets") to a third party. In procuring such sale, Purchaser and Seller shall act in good faith and in accordance with their respective reasonable requirements. Purchaser and Seller agree that such sale shall be conducted by the independent trustee, who shall be instructed to sell within a period of no longer than six (6) months, the Non-Cleared Assets, to a third party on the best price and other terms reasonably available.

Confidential

Execution version

Seller shall have no liability to Purchaser in relation to the price obtained for such sale (other than set out below) or in relation to any other term or condition of such sale, and shall not in any circumstances act as a trustee or fiduciary for Purchaser. Purchaser shall be entitled to receive within 3 (three) Business Days after the completion of such sale the proceeds of the sale of any Non-Cleared Assets (net of any Taxes, fees and expenses including the fees and expenses of the independent trustee, arising in connection with such sale);

- 6.5.7 the Parties shall co-operate to ensure that, until the Deferred Closing occurs in respect of, or Clause 6.5.6 becomes applicable to, a Deferred Jurisdiction, services and commercial relations between, on the one hand, the Group Companies and Businesses in respect of which Closing does take place, and, on the other hand, the Non-Material Group Companies and Non-Material Businesses, are maintained and rendered in the ordinary course of business;
- 6.5.8 nothing in this Agreement shall be interpreted or implemented in any manner that would require the Parties to violate any Competition Laws, and the operation of any other provision of this Agreement shall not permit the transfer to Purchaser or any Affiliate thereof (including any Group Company directly or indirectly owned by Purchaser following the Closing) of a Non-Cleared Asset or any beneficial interest therein unless and until the Conditions Precedent set out in Clauses 4.1.1 and 4.1.2 have been fulfilled insofar as is required to allow such asset to be lawfully and validly transferred to Purchaser or such Affiliate;
- 6.5.9 Clauses 6.2.3 and 6.4 shall apply *mutatis mutandis* to this Clause 6.5, provided that any termination of this Agreement shall only apply to the extent relating to those Deferred Jurisdictions in respect of which no Deferred Closing has occurred by date of such termination; and
- 6.5.10 all Tax costs incurred by Seller's Group in connection with any restructuring actions taken pursuant to Clause 6.5.2 to procure the effecting of any Deferred Closing shall be shared equally between Purchaser and Seller, and before effecting any such restructuring actions, Seller shall consult with Purchaser as to the estimated Tax costs which Seller expects to be incurred by Seller's Group in connection with such restructuring actions. Seller shall take reasonable steps to eliminate or reduce such Tax costs and shall take into consideration any suggestions made by Purchaser in relation thereto.

7 POST-CLOSING ADJUSTMENTS

Working Capital Statements and Net Debt Statements

- 7.1.1 Within 40 (forty) Business Days after the Effective Time, Seller shall prepare and deliver to Purchaser:
 - (a) a draft of the Working Capital Statement for each of the Company (including all other Group Companies other than the US Company and its Subsidiary) and the US Company (including its Subsidiary) and each relevant Business in the form set out in

23

Confidential

Execution version

Schedule 13 (Part 1); and

- (b) a draft of the Net Debt Statement for each of the Company (including all other Group Companies other than the US Company and its Subsidiary) and the US Company (including its Subsidiary) in the form set out in Schedule 13 (Part 2).
- 7.1.2 Each Working Capital Statement and Net Debt Statement shall be prepared in accordance with Schedule 13 (Part 3).
- 7.1.3 The draft Working Capital Statements and draft Net Debt Statements shall be delivered to Purchaser at the same time, it being further agreed that:
 - (a) in the event that Purchaser disagrees with any draft Working Capital Statement or draft Net Debt Statement, Purchaser shall within 30 (thirty) Business Days after receipt thereof, deliver notice of such disagreement to Seller, such notice (the "Notice of Disagreement") to specify (i) each line item in the relevant draft statement(s) with which Purchaser disagrees, including the amount of the adjustment required by Purchaser to be made in respect of such line item, and (ii) in reasonable detail, the reason for Purchaser's disagreement in respect of each such line item, provided, however, that Purchaser shall not deliver a Notice of Disagreement unless the adjustments identified therein would result in a change to the aggregate Working Capital and Net Debt of the Group in an amount no less than USD 2,000,000 (two million US dollars);
 - (*b*) if Purchaser does not deliver a Notice of Disagreement in terms of Clause 7.1.3(a), the draft Working Capital Statements and draft Net Debt Statements shall be final and binding on Seller (and each relevant other member of Seller's Group) and Purchaser (and each relevant other member of Purchaser's Group) for all purposes;
 - (c) if Purchaser delivers a Notice of Disagreement in terms of Clause 7.1.3(a), then:
 - (i) at the election of Seller each line item in the draft Working Capital Statements and draft Net Debt Statements in respect of which Purchaser does not deliver a Notice of Disagreement in accordance with Clause 7.1.3(a), shall be final and binding on Seller (and each other member of Seller's Group) and Purchaser (and each other member of Purchaser's Group) for all purposes, provided that

Seller shall, by notice to Purchaser within 10 (ten) Business Days after the delivery of a Notice of Disagreement, be entitled to place such other line items in dispute as are directly related to the line items placed in dispute by Purchaser under the Notice of Disagreement; and

(ii) Seller and Purchaser shall attempt in good faith to reach agreement in respect of those line items in the relevant draft Working Capital Statement(s) and/or draft Net Debt Statement(s) in respect of which Purchaser has thus given Notice of Disagreement or Seller has provided notice to Purchaser pursuant to Clause 7.1.3(c)(i), provided that if Seller and Purchaser do not reach such agreement within 20 (twenty) Business Days after the later of the delivery of the Notice of Disagreement, or Seller's notice pursuant to Clause 7.1.3(c)(i),

24

Confidential

Execution version

Seller or Purchaser may by notice to the other require that those line items in the relevant draft Working Capital Statement(s) and/or draft Net Debt Statement(s) (A) that have been properly specified in Purchaser's Notice of Disagreement in accordance with Clause 7.1.3(a) or Seller's notice pursuant to Clause 7.1.3(c)(i), or (B) that have not become final and binding pursuant to Clause 7.1.3(c)(i), and in each case have not subsequently been agreed upon between Seller and Purchaser within the aforesaid 20 (twenty) Business Days (each a "**Disputed Item**"), be referred to the Reporting Accountants in the terms of Schedule 13 (Part 4).

- 7.1.4 In order to enable the preparation and determination of the Working Capital Statements and Net Debt Statements, Purchaser shall procure the keeping up-to-date and, subject to reasonable notice, making available to Seller's Representatives during normal office hours of all books and records relating to the Group to the extent relevant to the preparation and determination of the Working Capital Statements and the Net Debt Statements, and co-operate with them with regard to the preparation and determination of the Working Capital Statements and Net Debt Statements. Purchaser's Group shall, in so far as it is reasonable to do so, make available the employees of the Group to assist Seller in the performance of Seller's duties and exercise of Seller's rights under this Clause 7.1.
- 7.1.5 In order to enable Purchaser to evaluate the draft Working Capital Statements and Net Debt Statements, Seller shall provide Purchaser's Representatives with reasonable detailed support for the underlying information and basis of preparation of the Working Capital Statements and Net Debt Statements and co-operate with Purchaser's Representatives with regard to the review and evaluation thereof.

2 Adjustments to Purchase Price and repayment of Intra-Group Indebtedness

7.2.1 Working Capital

In respect of each of the Company (including all other Group Companies other than the US Company and its Subsidiary) and the US Company (including its Subsidiary) and each relevant Business:

- (a) if its Working Capital Adjustment reflected in the relevant Working Capital Statement is less than its Estimated Working Capital Adjustment, Seller, on behalf of the Share Seller, US Seller or Business Seller, as the case may be, shall repay to Purchaser, acting on behalf of the relevant Share Purchaser or Business Purchaser, an amount equal to such deficit; or
- (b) if its Working Capital Adjustment reflected in the relevant Working Capital Statement exceeds its Estimated Working Capital Adjustment, Purchaser, on behalf of the relevant Share Purchaser or Business Purchaser, as the case may be, shall pay Seller, acting on behalf of the Share Seller, US Seller or Business Seller, an amount equal to such excess.

25

Confidential

Execution version

7.2.2 Net Debt

In respect of each of the Company (including all other Group Companies other than the US Company and its Subsidiary) and the US Company (including its Subsidiary):

- (a) to the extent that its Net Debt reflected in the relevant Net Debt Statement exceeds the relevant Estimated Net Debt, Seller shall have a payment obligation to Purchaser in an amount equal to such excess; or
- (b) to the extent that its Net Debt reflected in the relevant Net Debt Statement is less than the relevant Estimated Net Debt, Purchaser shall have a payment obligation to Seller in an amount equal to such deficit.

7.2.3 Intra-Group Indebtedness

In respect of each Group Company:

- (a) to the extent that the amount of any Intra-Group Payable is less than the amount of the Estimated Intra-Group Payable, Seller shall have an obligation to pay, on behalf of the relevant member of the Seller's Group, to Purchaser, acting on behalf of the relevant Group Company, an amount equal to such deficit;
- (b) to the extent that the amount of any Intra-Group Payable exceeds the amount of the Estimated Intra-Group Payable, Purchaser shall have an obligation to pay, on behalf of the relevant Group Company, to Seller, acting on behalf of the relevant member of the Seller's Group, an amount equal to such excess;
- (c) to the extent that the amount of any Intra-Group Receivable is less than the amount of the Estimated Intra-Group Receivable, Purchaser shall have an obligation to pay, on behalf of the relevant Group Company, to Seller, acting on behalf of the relevant member of the Seller's Group, an amount equal to such deficit; and
- (*d*) to the extent that the amount of any Intra-Group Receivable exceeds the amount of the Estimated Intra-Group Receivable, Seller shall have an obligation to pay, on behalf of the relevant member of the Seller's Group, to Purchaser, acting on behalf of the relevant Group Company, an amount equal to such excess.

7.2.4 Interest

Any payment to be made in accordance with this Clause 7.2 shall include interest thereon calculated from the day after the Effective Time to the day of payment, both days inclusive, at the Interest Rate.

7.2.5 Payment

- (a) The due date for any payment to be made under this Clause 7.2, shall be the 5th (fifth) Business Day after the relevant Net Debt Statement or the relevant Working Capital Statement, as the case may be, has been finally determined in accordance with Clause 7.1.3.
- (b) All payments (including interest payments) made under this Clause 7.2 (excluding Clause 7.2.3) shall be made on account of the Purchase Price and the allocation of the Purchase Price shall be adjusted in accordance with Schedule 11 (Part 1).
- (c) To the extent legally permissible, the payment obligations and payments to be made

26

Confidential

Execution version

pursuant to this Clause 7.2, shall be aggregated and discharged by way of set-off.

8 POST-CLOSING OBLIGATIONS

8.1 Indemnity against Group Liabilities

Subject to Closing and Clause 8.2, and save to the extent otherwise provided in this Agreement, Purchaser shall indemnify, defend and hold harmless Seller and, as an irrevocable third party stipulation ('derdenbeding'), each member of Seller's Group against:

- 8.1.1 all Group Liabilities;
- 8.1.2 any Losses which any member of Seller's Group may suffer by reason of it taking any reasonable action to avoid, resist or defend any Liability referred to in Clause 8.1.1;

it being agreed that Clause 11, shall apply mutatis mutandis, to any claim by a member of Seller's Group under this Clause 8.1.

8.2 Indemnity against Excluded Liabilities

Seller shall indemnify and hold harmless Purchaser and, as an irrevocable third party stipulation ('derdenbeding'), each member of Purchaser's Group (including each Group Company, each Business Purchaser and each Share Purchaser) against:

- 8.2.1 all Excluded Liabilities; and
- 8.2.2 subject to Clause 11, any Losses that any member of Purchaser's Group may suffer by reason of it taking any reasonable action to avoid, resist or defend any Excluded Liability.

8.3 Release of Guarantees

8.3.1 Purchaser shall procure, with effect from Closing or as soon as practicable thereafter, the release of Seller and any other member of Seller's Group (excluding the Group Companies) from any (joint and/or several) Guarantees given by, assumed by or binding upon Seller or any other member of Seller's Group in relation to any of the Business Liabilities). or any of the Liabilities of the Group Companies (other than any Excluded Liabilities or any Liabilities (including any Liability for Tax) for which Seller is liable under an indemnity set out in this Agreement), provided that Seller shall procure that the relevant member of Seller's Group shall reimburse Purchaser (or the relevant member of Purchaser's Group) for any transaction costs initially incurred by Purchaser in excess of USD 500,000 (five hundred thousand US dollars) in the aggregate in procuring the release of any Guarantees that are not listed in Schedule 4. Subject to Closing, pending such release Purchaser shall indemnify, defend and hold harmless Seller and, as an irrevocable third party stipulation ('derdenbeding'), the other members of Seller's Group (excluding

the Group Companies) against all amounts paid by any of them pursuant to any such Guarantees (other than with respect to any Excluded Liabilities or any Liabilities (including any Liability for Tax) for which Seller is liable under an indemnity set out in this Agreement).

8.3.2 Seller shall procure, with effect from Closing or as soon as practicable thereafter, the

27

Confidential

Execution version

release of each Group Company from any (joint and/or several) Guarantees given by, assumed by or binding upon the Group Company in relation to any Liability of Seller or any other member of Seller's Group (excluding the Group Companies). Subject to Closing, pending such release, Seller shall indemnify, defend and hold harmless Purchaser and, as an irrevocable third party stipulation ('derdenbeding'), the other members of Purchaser's Group against all amounts paid by any of them pursuant to any such Guarantees.

8.4 Trade receivables/payables; Post-Closing receipts

- 8.4.1 Each Party shall procure that any indebtedness (other than indebtedness that is part of Intra-Group Receivables or Intra-Group Payables) incurred in the ordinary course of trading prior to Closing between the Group and Seller's Group and outstanding at Closing, shall be settled in cash in accordance with the usual terms and conditions of trading between such entities or, if there are no such terms, within 30 (thirty) days of invoice.
- 8.4.2 If at any time after the Effective Time, any member of Seller's Group receives any monies in respect of any Business Receivable or receivable of a Group Company, then Seller shall procure that the relevant member of Seller's Group shall pay the amount received to the relevant Business Purchaser or the relevant Group Company, as the case may be, as soon as reasonably practicable.
- 8.4.3 If at any time after the Effective Time, any member of Purchaser's Group receives any monies in respect of any right, interest, claim or other asset or Liability of any Business Seller which is not included in the Business Assets or Business Liabilities, then Purchaser shall procure that the relevant member of Purchaser's Group shall pay the amount received to the relevant Business Seller as soon as reasonably practicable.

8.5 Use of names

- 8.5.1 Purchaser shall not, and shall procure that no member of Purchaser's Group shall, after Closing, use in any way whatsoever, by means of trading names, domain names, registered or unregistered trade marks, logos or otherwise, the name AkzoNobel or Imperial Chemical Industries or ICI, or any abbreviation thereof or any name or lettering which is confusingly similar to the same. Notwithstanding the foregoing provisions of this Clause 8.5.1, Purchaser shall, as soon as practicable after the Closing, but in any event within 6 (six) months after the Closing Date, procure that the names AkzoNobel, Imperial Chemical Industries and ICI, or any abbreviation thereof, is removed from all products, business stationery and other assets held by any member of the Group or acquired by Purchaser pursuant to this Agreement, and, within 6 (six) months after the Closing Date, procure that such name or abbreviation is removed from all premises occupied by Purchaser or any other member of Purchaser's Group.
- 8.5.2 Except as otherwise provided in the IP Licence or in any Intra-Group Retained Agreement, Seller shall not, and shall procure that no member of Seller's Group shall, after Closing, use in any way whatsoever any Acquired Intellectual Property.

8.6 Retention of records

8.6.1 Purchaser shall retain for a period of 10 (ten) years from Closing, or such longer period as

28

Confidential

Execution version

may be prescribed by applicable Law, at least one copy of all material books, records and other written information relating to the Group that are at the Properties at Closing or that are held by or on behalf of any member of Purchaser's Group pursuant to the transactions contemplated by this Agreement and, to the extent reasonably required by Seller, shall allow Seller, upon reasonable notice, access during normal office hours to such books, records and other information, including the right to inspect and take copies (at Seller's expense); provided that, as between the Parties, Purchaser may dispose of such books, records or other information prior to the termination of such 10-year or longer period if it gives at least 90 days' prior written notice to Seller of its intention to do so, in which case Seller shall be given an opportunity (at Seller's expense), to remove and retain all or any part of such books and records as it may elect.

8.6.2 Seller's Group shall retain for a period of 10 (ten) years from Closing, or such longer period as may be prescribed by applicable Law, any material books, records or other written information relating to any Group Company or any Business that are not at the Properties at Closing or held by or on behalf of any member of Purchaser's Group pursuant to the transactions contemplated by this Agreement and, to the extent reasonably required by Purchaser, shall allow Purchaser, upon reasonable notice, access during normal office hours to such books, records and information, including the right to inspect and take copies (at Purchaser's expense); provided that, as between the Parties, Seller may dispose of such books, records or other information prior to the termination of such 10-year or longer period if it gives at least 90 days' prior written notice to Purchaser of its intention to do so, in which case Purchaser shall be given an opportunity (at Purchaser's expense), to remove and retain all or any part of such books and records as it may elect.

8.6.3 Without limiting the generality of the foregoing, in the case of records relating to Taxation, and, subject to Paragraph 1 of Schedule 12 in relation to VAT records, Seller and Purchaser shall retain all records that may be required for the conduct of any audit or other proceedings until the expiration of the applicable statutory limitation period or, if later, a final judgment in or settlement of any ongoing audit or other proceeding with any applicable Taxing Authority and, upon reasonable notice, shall provide each other access to all books and records relating to the Group as may be reasonably required to exercise their rights under Paragraph 9 of Schedule 14. In particular, Purchaser agrees to retain records of National Starch LLC in such format as required in Internal Revenue Service Revenue Procedure 98-25.

8.7 Insurance

8.7.1 Termination of coverage

Subject to Clause 8.7.2, as of the Closing Date all coverage with respect to the Group Companies and Businesses under the Insurance Policies in respect of events, occurrences or accidents occurring on or after the Closing Date shall be cancelled and terminated, excluding those Insurance Policies in respect of which one or more Group Companies are the sole policy holders or named insureds (the "Continuing Policies").

8.7.2 Availability of Seller's Group's rights of recovery under certain Insurance Policies

29

Confidential

Execution version

Subject to Closing, for a period of 90 (ninety) days after the Closing Date, Seller shall cause the relevant members of Seller's Group to make available (to the extent permitted under the relevant Insurance Policies) to the Group Companies and Business Purchasers (on behalf of and in relation to the Operations) Seller's Group's rights of recovery under the Insurance Policies solely in respect of third party general and product liability, auto liability or workers' compensation / employers' liability, claims filed in relation to events, occurrences or accidents occurring prior to the Closing Date, to the extent that such claims are not reflected in a Working Capital Statement or a Net Debt Statement (if then final and binding pursuant to Clause 7.1.3). Seller and Purchaser shall co-operate in connection with the handling and administration of these claims. Subject to Clause 8.7.3, following this 90 (ninety) day period, the Group Companies and Business Purchasers shall not be entitled to any rights of recoveries under any of the Insurance Policies other than the Continuing Policies for claims made in respect of events, occurrences or accidents occurring prior to the Closing Date that have not been reported by a Group Company or other member of Purchaser's Group.

8.7.3 Historical occurrence-based Insurance Policies

Subject to Closing, upon Purchaser's request and at Purchaser's sole cost and expense, Seller shall cause the relevant members of Seller's Group to make available to Purchaser's Group (to the extent permitted under and subject to the terms and limits of cover applicable under the relevant Insurance Policies and on a non-exclusive basis) any rights of recovery which any member of Seller's Group may have under any occurrence-based Insurance Policies with third-party insurers that at any time have covered Liabilities arising from or with respect to the Operations, including all such Insurance Policies in respect of which a predecessor of any Affiliate of Seller was the named insured.

8.8 Tax Indemnity

The provisions of Schedule 14 (Part 1) shall apply in respect of Taxes.

8.9 Intellectual Property Matters

The provisions of Schedule 14 (Part 2) shall apply in respect of Intellectual Property matters.

8.10 Hive-out Matters

- 8.10.1 Seller shall not assert any right to terminate any NPG Contract pursuant to Section 7.5(b) of the Master NPG Agreement or any similar provision of any other Hive-out Agreement.
- 8.10.2 Seller and Purchaser shall not, and shall cause their respective Affiliates not to, assert under any circumstance any claim against each other for any breach of, or any default under, Section 7.3 of the Master NPG Agreement or any similar provision of any other Hive-out Agreement.
- 8.10.3 As and to the extent requested by Purchaser, Seller shall, subject to being reimbursed by Purchaser for all reasonable out-of-pocket expenses incurred by Seller, promptly exercise, or cause to be exercised, any rights Seller or any of its Affiliates may have to require Henkel KGaA or its Affiliates to hold the benefit of any NPG Contract in trust, and Seller

30

Confidential

Execution version

shall hold or cause to be held the benefit of such NPG Contract in trust for Purchaser or such member of Purchaser's Group as Purchaser may specify. Seller shall promptly pay over to Purchaser or such member any monetary benefit received by Seller or any of its Affiliates under any such NPG Contract.

8.11 Payment of Employee bonuses related to the Transaction

Seller covenants that (i) the gross amount of all bonuses or change of control amounts contractually agreed to by a member of Seller's Group prior to the Closing Date in excess of normal salary and benefits, to be paid to any Employees in connection with or as a result of the Transaction (including the employee's share of any related employment Taxes required to be withheld from such bonuses and paid to a Tax Authority), and (ii) the employer's share of any related employment Taxes attributable thereto, shall, in each case, be paid by Seller (or reimbursed by Seller to Purchaser taking into account any net financial benefit in accordance with Clause 10.8 if such bonuses and/or Taxes are paid by a member of Purchaser's Group) when due (whether prior to, on or after Closing).

8.12 National Starch & Chemical GmbH

Purchaser shall not, and shall procure that no member of Purchaser's Group shall, in the period from Closing until 1 January 2013 (i) make a direct transfer of the shares in National Starch & Chemical GmbH to any party other than Deutsche ICI GmbH or (ii) merge (including an upstream merger and a downstream merger) Deutsche ICI GmbH and National Starch & Chemical GmbH.

9 WARRANTIES

9.1 Seller's Warranties

- 9.1.1 Subject to the remaining provisions of this Clause 9 and to Clauses 10 and 11, Seller represents and warrants to Purchaser that Seller's Warranties are true and correct as at Signing.
- 9.1.2 Subject to the remaining provisions of this Clause 9 and to Clauses 10 and 11, Seller further represents and warrants to Purchaser that Seller's Warranties will also be true and correct at Closing on the Closing Date as if they had been repeated at such Closing (except to the extent any Seller's Warranty speaks as of a specific date, in which case such Seller's Warranty will be true and correct at such date).
- 9.1.3 Each Seller's Warranty applies only to the subject expressly referred to therein. Without detracting from the generality of the foregoing, the only Seller's Warranties given:
 - (a) in respect of the Properties are those contained in Paragraphs 5, 7 and 9 of Schedule 15 (Part 1) and all other Seller's Warranties shall be deemed not to be given in respect of the Properties;
 - (*b*) in respect of Intellectual Property are those contained in Paragraphs 6, 7 and 9 of Schedule 15 (Part 1) and all other Seller's Warranties shall be deemed not to be given in respect of Intellectual Property;

31

Confidential

Execution version

- (c) in respect of employment and pension matters are those contained in Paragraphs 7, 8 and 9 of Schedule 15 (Part 1) and all other Seller's Warranties shall be deemed not to be given in respect of such matters;
- (*d*) in respect of the Environment, Environmental Laws and Environmental Permits, are those contained in Paragraphs 9 and 10 of Schedule 15 (Part 1) and all other Seller's Warranties shall be deemed not to be given in respect of the Environment;
- (e) in respect of anti-trust, fair trading, dumping, state and consumer protection or similar matters are those contained in Paragraphs 7, 9 and 11 of Schedule 15 (Part 1) and all other Seller's Warranties shall be deemed not to be given in respect of such matters; and
- (f) in respect of Tax matters are the Tax Warranties and all other Seller's Warranties shall be deemed not to be given in respect of such matters.
- 9.1.4 Purchaser acknowledges and agrees that Seller makes no representation or warranty as to the accuracy of any forecasts, estimates, projections, statements of intent or statements of opinion howsoever provided to Purchaser or any of its Representatives at or prior to Signing. Purchaser acknowledges that no representations or warranties, express or implied, have been given or are given other than the Seller's Warranties.
- 9.1.5 Any Seller's Warranty qualified by the expression "so far as Seller is aware" or any similar expression shall be deemed to refer to the knowledge of any individual whose name is set out in Schedule 15 (Part 3), who shall be deemed to have knowledge of such matters as such individual would have discovered, had such individual made reasonable enquiries within the Group and Seller's Group.
- 9.1.6 The applicability of article 7:17 of the Netherlands Civil Code is hereby excluded.

9.2 Disclosure

- 9.2.1 Each of Seller's Warranties is limited by, and Seller shall not be in breach of or liable for any such Seller's Warranty in respect of, the matters disclosed in the Disclosure Letter or in the Data Room Information, provided that a matter disclosed through the Data Room Information shall be deemed to limit any of Seller's Warranties only to the extent that such matter as it appears in the Data Room Information would be reasonably inferred to qualify such Seller's Warranty.
- 9.2.2 Seller's Warranties are given by Seller with the knowledge that Purchaser is relying thereon in connection with the Transaction, in acknowledgement and acceptance of the limitations contained herein applicable thereto. In the event of a claim for breach of Seller's Warranties, Seller shall not in any way argue that Purchaser should have further investigated the Group or the Operations, or, except as otherwise provided in

Confidential

Execution version

9.3 Liability for breach

- 9.3.1 Except as set out in Clauses 4.3.3, 6.2.3 and 6.4 and except in the event of fraud on the part of Seller, Share Seller, the US Seller or any Business Seller, in the event of any breach by Seller of any covenant, Seller's Warranty or other provision of or any claim under any indemnity included in this Agreement, Purchaser shall not have the right to terminate or rescind this Agreement and as its sole and exclusive remedy and subject to any other limitations of liability set out in this Agreement, shall have the right, after the Closing Date, to be compensated for Losses suffered or incurred by Purchaser as a result of such breach, it being agreed that (a) Seller shall not be liable for any Losses based on any multiplier used in Purchaser's valuation of the Group, and (b) a Loss suffered by the relevant Group Company or Business Purchaser in respect of such breach shall, subject to any other limitations of liability set out in this Agreement, be deemed to be a Loss suffered or incurred by Purchaser.
- 9.3.2 If Seller is liable for a breach of any Seller's Warranty any qualification contained in any Seller's Warranty relating to "materiality" or "Group Material Adverse Effect" shall be disregarded for purposes of quantifying the Loss suffered or incurred by Purchaser or any other member of Purchaser's Group as a result of such breach.
- 9.3.3 Except as set out in Clauses 4.3.3, 6.2.3 and 6.4 and except in the event of fraud on the part of Purchaser, US Purchaser or any Business Purchaser, in the event of any breach by Purchaser of any covenant, Purchaser's Warranty or other provision of or any claim under any indemnity included in this Agreement, Seller shall not have the right to terminate or rescind this Agreement and as its sole and exclusive remedy and subject to any other limitations of liability set out in this Agreement, shall have the right, after the Closing Date, to be compensated for Losses suffered or incurred by Seller as a result of such breach, it being agreed that a Loss suffered by the relevant member of Seller's Group in respect of such breach shall, subject to any other limitations of liability set out in this Agreement, be deemed to be a Loss suffered or incurred by Seller.

9.4 Purchaser's Warranties

- 9.4.1 Purchaser represents and warrants to Seller that, as at Signing:
 - (a) the statements set out in Schedule 15 (Part 4) are true and correct; and
 - (b) none of the persons referred to in Schedule (Part 5) has any actual knowledge, irrespective of the source, of any breach of Seller's Warranties.
- 9.4.2 Purchaser further represents and warrants to Seller that the statements set out in Schedule 15 (Part 4) will also be true and correct at Closing on the Closing Date as if they had been repeated at such Closing.

33

Confidential

Execution version

10 LIMITATION OF LIABILITY

10.1 Time limitation

No Party shall be liable in respect of any claim for breach of any Warranty under this Agreement unless a notice of the claim is given to such Party specifying the matters set out in Clause 11.1:

- 10.1.1 in the case of any claim under the Tax Warranties, within 90 (ninety) days after the later of (i) the expiry of the statutory limitation period (including any extensions thereof) applicable in the relevant jurisdiction for the Tax matter giving rise to such claims and (ii) the date on which an additional assessment is levied in accordance with the relevant applicable Law permitting such additional assessment;
- 10.1.2 in the case of any claim under Paragraph 10 of Schedule 15 (Part 1) (Environment), within 5 (five) years after the Closing Date; and
- 10.1.3 in the case of any other claim for a breach of any Seller's Warranty (other than any claim under Paragraph 1(a), 1(b), 1(c), 1(d), 2(a), 5(a)(ii), 5(a)(iii) or 5(d) of Schedule 15 (Part 1)) within eighteen (18) months after the Closing Date, provided that to the extent the Seller's Warranty under Paragraph 23 (*disclosure of information*) of Schedule 15 (Part 1) relates to any of Seller's Warranties referred to in Clause 10.1.1 or Clause 10.1.2, such Seller's Warranty under Paragraph 23 (*disclosure of information*) Schedule 15 (Part 1) shall survive for the period referred to in Clause 10.1.1 or Clause 10.1.2, as applicable.

10.2 Minimum claims

Subject to any other limitations set out in this Clause 10, Seller shall not be liable under this Agreement in respect of any individual claim or series of related claims in respect of a breach of its Warranties, unless the liability agreed or determined in respect of any such claim or series of claims exceeds USD 1,500,000 (one million five hundred thousand US dollars).

10.3 Aggregate minimum claims

Subject to any other limitations set out in this Clause 10, Seller shall not be liable under this Agreement in respect of any claim in respect of a breach of its Warranties except to the extent that such claim, together with the aggregate amount of all claims for which it would otherwise be liable under this Agreement, exceeds USD 15,000,000 (fifteen million US dollars).

10.4 Maximum liability

- 10.4.1 Subject to Clause 10.4.3, the aggregate liability of Seller in respect of all claims in respect of a breach of Seller's Warranties shall not exceed 15% (fifteen percent) of the Purchase Price.
- 10.4.2 Without detracting from Clause 10.4.1, but subject to Clause 10.4.3, the aggregate liability of Seller under this Agreement shall not exceed 100% (one hundred percent) of the Purchase Price.
- 10.4.3 The limitations in this Clause 10.4 shall not apply to any Excluded Liabilities or in the case of fraud on the part of Seller, Share Seller, the US Seller or any Business Seller, and no such liability nor any Liability for Taxes for which Seller is liable under an indemnity set out

34

Confidential

Execution version

in this Agreement shall be included in determining the aggregate liability of Seller for purposes of Clauses 10.4.1 and 10.4.2.

10.5 Provisions

Seller shall not be liable under this Agreement in respect of any claim if and to the extent that any allowance, provision or reserve which is made in a Working Capital Statement or Net Debt Statement, specifically takes into account the matter giving rise to the claim.

10.6 Matters arising after Signing / Effective Time

Seller shall not be liable under this Agreement in respect of any matter, act, omission or circumstance (or any combination thereof), including the aggravation of a matter or circumstance, to the extent that the same would not have occurred but for or was contributed to by:

- 10.6.1 any matter or thing done or omitted to be done as required by this Agreement or otherwise at the request or with the express written approval of an executive officer of Purchaser;
- 10.6.2 any act, omission or transaction of Purchaser or any other member of Purchaser's Group, or their respective directors, officers, employees or agents or successors in title, after Signing, excluding acts, omission or transactions undertaken in bona fide reliance on any written information received from Seller that was not correct at or as at the time when given;
- 10.6.3 any act, omission or transaction of any of the Group Companies, or their respective directors, officers, employees or agents or successors in title, after the Closing, excluding acts, omissions or transactions undertaken in bona fide reliance on any written information received from Seller that was not correct at the time when given;
- 10.6.4 the passing or coming into force of, or any change in, any Law or administrative practice of any Governmental Authority on or after the Closing Date:
- 10.6.5 any change on or after the Closing Date of any generally accepted interpretation or application of any Law;
- 10.6.6 any change in any accounting or Taxation policy, basis or practice of Purchaser or any other member of Purchaser's Group introduced or having effect after Signing; or
- 10.6.7 any change in any accounting or Taxation policy, basis or practice of any of the Group Companies introduced or having effect after the Closing.

10.7 Insurance

Neither Party shall be liable in respect of any claims made by Purchaser or any of its Affiliates to the extent that the Losses in respect of which a claim is made are actually recovered pursuant to a policy of insurance (net of all deductibles, allocated loss adjustment expenses, retention amounts (including therein any captive insurer retention amounts or reinsurance amounts), third party administration fees, other charges and reasonable recovery expenses provided for by the terms of any such policy of insurance). Each Party shall use commercially reasonable efforts to recover any such Losses pursuant to any such policy.

Execution version

10.8 Net financial benefit

Neither Party shall be liable under this Agreement in respect of any claims to the extent of any corresponding Tax savings by the other Party or any of its Affiliates arising in respect of such Losses or the facts giving rise to such Losses where the amount (if any) by which any Taxation for which the other Party or such Party's Affiliate would otherwise have been accountable or liable to be assessed is actually reduced or extinguished as a result of the matter giving rise to such liability (after taking into account the amount of any increased Taxes incurred by such Party as a result of the accrual or receipt of any payments for such claims under this Agreement, including any incurred Taxes resulting from compensating such Party for any such increases).

10.9 Mitigation of Losses

Each Party shall procure that all commercially reasonable steps are taken and all commercially reasonable assistance is given to avoid or mitigate any Losses which, in the absence of mitigation, might give rise to a liability in respect of any claim under this Agreement, it being agreed that any liability of a Party under this Agreement for Losses suffered or incurred by the other Party shall include liability in respect of the reasonable costs and expenses suffered or incurred by the other Party in taking such mitigation steps; provided, however, that with respect to (a) Excluded Liabilities and any Liabilities for which Seller is liable under an indemnity set out in this Agreement, all such steps and all such assistance shall be at Seller's sole cost and expense (including for all of Purchaser's reasonable internal costs) and (b) Group Liabilities all such steps and all such assistance shall be at Purchaser's sole cost and expense (including for all of Seller's reasonable internal costs).

10.10 Right to recover

- 10.10.1 Neither Party shall be required to pay over any amounts relating to any actual Liability unless and until such actual Liability is due and payable, or any Losses relating to any contingent Liability unless and until such contingent Liability becomes an actual Liability and is due and payable, provided that this Clause 10.10.1 shall not operate to exclude liability in relation to a claim made in respect of an actual or contingent Liability within the relevant time limit specified in Clause 10.1 and specifying the matters set out in Clause 11.2.
- 10.10.2 Neither Party shall be liable in respect of any Losses claimed under this Agreement to the extent that such Losses are covered by a right of recovery against a third party and the relevant member of Purchaser's Group or Seller's Group, as applicable, fails to use commercially reasonable efforts to make such recovery. If, before a Party pays an amount in discharge of any claim under this Agreement, the other Party has recovered (directly or indirectly, by payment, discount, credit, relief, insurance or otherwise) from a third party a sum which compensates such other Party (in whole or in part) in respect of the Loss which is the subject matter of the claim, then such payment obligations shall be commensurately reduced after taking into account costs and expenses reasonably incurred in obtaining such recovery and limited to the amount actually recovered in respect of the claim. If a Party (the "Payor Party") has paid an amount in discharge of any claim under this Agreement and the other Party (the "Payee Party") subsequently recovers (directly or

36

Confidential

Execution version

indirectly, whether by payment, discount, credit, relief, insurance or otherwise) from a third party a sum which compensates the Payee Party (in whole or in part) in respect of the Loss which is the subject matter of the claim, then the Payee Party shall forthwith pay or procure to be paid to the Payor Party the amount recovered, less any costs and expenses reasonably incurred in obtaining such recovery and limited to the amount actually paid by the Payor Party in respect of the claim.

10.10.3 Double claims

Neither Party shall be entitled to recover from the other Party under this Agreement more than once in respect of the same Losses suffered. Subject to any other limitations set out in this Agreement, in the event that any matter, act, omission or circumstance (or any combination thereof) giving rise to a breach of a Warranty is also the subject of an indemnity under this Agreement, the claim for such breach of Warranty shall be limited to a claim under said indemnity.

10.11 Tax indemnity and Tax warranty claims

Notwithstanding anything to the contrary in the Agreement and except for Clause 10.1 and Clause 10.8, this Clause 10 shall not apply with respect to any claim under the Tax Warranties or the Tax Indemnity.

11 CLAIMS

11.1 Notification of potential claims

11.1.1 If Purchaser becomes aware of any matter or circumstance that is reasonably likely to give rise to a claim against Seller under this Agreement, Purchaser shall within 20 (twenty) Business Days deliver a notice to Seller setting out such information as is available to Purchaser or any other member of Purchaser's Group as is reasonably necessary to enable Seller to assess the merits of the claim, to act to preserve evidence and to make such provision as Seller may consider necessary, provided that Purchaser shall continue to have the right to make any such claim after such 20 (twenty) Business Day period, except to the extent that Seller is actually prejudiced by Purchaser's failure to comply with the terms of this Clause 11.1.1.

- 11.1.2 If Seller becomes aware of any matter or circumstance that is reasonably likely to give rise to a claim against Purchaser under this Agreement, Seller shall within 20 (twenty) Business Days deliver a notice to Purchaser setting out such information as is available to Seller or any other member of Seller's Group as is reasonably necessary to enable Purchaser to assess the merits of the claim, to act to preserve evidence and to make such provision as Purchaser may consider necessary, provided that Seller shall continue to have the right to make any such claim after such 20 (twenty) Business Day period, except to the extent that Purchaser is actually prejudiced by Seller's failure to comply with the terms of this Clause 11.1.2.
- 11.1.3 The applicability of Sections 6:89 and 7:23 of the Netherlands Civil Code is hereby

37

Confidential

Execution version

excluded.

11.2 Notification of claims

Without detracting from Clause 11.1, notices of claims under this Agreement shall be given by each Party within the time limits specified in Clause 10.1, specifying reasonably adequate information of the legal and factual basis of the claim and the evidence on which such Party relies and, if practicable, an estimate of the amount of Losses which are, or are to be, the subject of the claim (including any Losses which are contingent on the occurrence of any future event), provided that a deficiency within such notice shall not preclude the Party providing such notice from, subject to any other applicable limitations of liability, recovering from the other Party with respect to such claim, except to the extent that the other Party is actually prejudiced by such deficiency.

11.3 Commencement of proceedings

Without detracting from Clause 11.1.1 or 11.1.2, any claim notified to Seller or Purchaser, as applicable, pursuant to Clause 11.1.1 or 11.1.2 shall (if it has not been previously satisfied, settled or withdrawn) be deemed to be irrevocably withdrawn 12 (twelve) months after the notice is given pursuant to Clause 11.1.1 or 11.1.2, or, in the case of any contingent liability, 12 (twelve) months after such contingent liability becomes an actual liability and is due and payable, unless legal proceedings in respect of it (i) have been formally commenced and (ii) are being and continue to be pursued with reasonable diligence.

11.4 Investigation

In connection with any matter or circumstance notified pursuant to Clause 11.1 or 11.2:

- 11.4.1 to the extent reasonably required by the other Party, the Party delivering notice shall allow the other Party and its financial, accounting, legal and other advisors to investigate the matter or circumstance alleged to give rise to such claim and whether and to what extent any amount is or may be payable in respect of such claim; and
- 11.4.2 the Party delivering notice shall disclose to the other Party all material information of which it is aware that reasonably relates to the claim and shall procure that all of its Affiliates shall give, subject to their being paid reasonable costs and expenses, all such information and assistance, including access to premises and personnel upon reasonable notice and during normal office hours and the right to examine and copy or photograph any assets, accounts, documents and records, in each case to the extent reasonably necessary and reasonably requested in connection with the claim.

11.5 Procedure for third party claims

If the claim set out in the notice delivered by the claiming Party (the "Claiming Party") to the other Party (the "Other Party") pursuant to Clause 11.1 is a result of or in connection with a claim by or liability to a third party then:

11.5.1 no admissions in relation to such third party claim shall be made by or on behalf of the Claiming Party or any of its Affiliates, and the claim shall not be compromised, disposed of or settled without the prior written consent of the Other Party;

38

Confidential

Execution version

11.5.2 if such third party claim or liability is solely for money damages or, where Seller is the Other Party, will have no continuing effect in any material respect on the Group or the Operations, the Other Party shall be entitled at its own expense by notice to the Claiming Party, and the Claiming Party and its Affiliates shall duly and fully co-operate to allow the Other Party, to assume the defence with counsel reasonably acceptable to the Claiming Party at its sole cost and expense and to take such action as it deems reasonably necessary to avoid, dispute, deny, defend, resist, appeal, compromise or contest such claim or liability (including making counterclaims or other claims against third parties) in the name of and on behalf of the Claiming Party or its Affiliates concerned, and to control the conduct of any related proceedings, negotiations or appeals, provided that the Other Party shall not settle or compromise any claim or demand without the consent of the Claiming Party, such consent not to

be unreasonably withheld, it being agreed that it shall be deemed reasonable for the Claiming Party to withhold its consent for any settlement that includes anything other than a monetary settlement and a complete release of the Claiming Party;

- 11.5.3 where the Other Party has issued a notice pursuant to Clause 11.5.2, the Claiming Party shall, and shall procure that its Affiliates shall, give, subject to their being paid reasonable costs and expenses, all such information and assistance including access to premises and personnel, and the right to examine and copy or photograph any assets, accounts, documents and records, in each case to the extent the Other Party may reasonably request for the purpose referred to in Clause 11.5.2, including instructing such professional or legal advisors as the Other Party may nominate to act on behalf of the Claiming Party or its Affiliates concerned, but in accordance with the Other Party's instructions, it being agreed that the Other Party shall keep the Claiming Party informed of all relevant matters relating to the claim and shall forward or procure to be forwarded to the Claiming Party copies of all material external correspondence (other than such correspondence as is subject to legal professional privilege) relating to the claim, provided that the Claiming Party shall have the right to employ separate counsel, the reasonable costs of which shall be for the Other Party's expense, and solely to control the Claiming Party's defence of such asserted liability if in the reasonable written opinion of counsel to the Claiming Party, a conflict or potential conflict exists between the Other Party and the Claiming Party that would make such separate representation necessary under the applicable canons of ethics, provided that in such event, the Claiming Party shall not settle or compromise any claim or demand without the consent of the Other Party, such consent not to be unreasonably withheld; and
- 11.5.4 until such time as the Other Party may issue a notice pursuant to Clause 11.5.2 and without detracting from the foregoing provisions of this Clause 11.5, the Claiming Party shall procure that (a) it and its relevant Affiliates take into consideration the reasonable interests of the Other Party when taking any action to avoid, dispute, deny, defend, resist, appeal, or contest the claim or liability (including in respect of the making of counterclaims or other claims against third parties), and (b) the Claiming Party and its relevant Affiliates and their legal advisors co-operate with the Other Party's legal advisors in allowing the Other Party's legal advisors to oversee such envisaged actions and to the extent

39

Confidential

Execution version

reasonably practicable shall consult with the Other Party's legal advisors and take into consideration their comments prior to taking such envisaged actions.

11.6 Tax indemnity and Tax warranty claims

Notwithstanding anything to the contrary in the Agreement, this Clause 11 shall not apply with respect to any claim under the Tax Warranties or the Tax Indemnity.

12 RESTRICTIONS

12.1 Restrictions on Seller

- 12.1.1 Seller undertakes with Purchaser to procure that no member of Seller's Group shall:
 - (a) during the Restricted Period undertake any Restricted Activity;
 - (*b*) during the Restricted Period canvass or solicit the custom, in relation to a Restricted Activity, of any person or company who has within 2 (two) years prior to the Closing Date been a customer of the Group; or
 - (c) during a period commencing on the Signing Date and ending 2 (two) years after the Closing Date, hire or seek to hire any Restricted Employee or any other person employed immediately prior to the Signing Date in Purchaser's Group with a base salary (or equivalent) in excess of USD 135,000 (one hundred thirty-five thousand US dollars) as employee, consultant or otherwise, whether or not such person would thereby commit a breach of his contract of service, provided that the placing of an advertisement of a post available to a member of the public generally and the recruitment of a person through an employment agency shall not constitute "seeking to hire" for purposes of this Clause 12, provided that no member of Seller's Group instructs or encourages such agency to approach any such person.
- 12.1.2 The restrictions in Clause 12.1.1(a) shall not operate to prohibit any member of Seller's Group from:
 - (a) exercising any right or fulfilling any obligation pursuant to this Agreement and any agreement entered into pursuant to this Agreement;
 - (b) acquiring the whole or part of any business if the annual turnover attributed to a Restricted Activity does not represent more than the lower of USD 250,000,000 (two-hundred fifty million US dollars) and the amount equal to 25% of the aggregate annual turnover of such acquired business for the immediately preceding financial year of such acquired business, provided that Seller shall use commercially reasonable efforts to procure that the divestment of the portion of any such acquired business engaged in such Restricted Activity be commenced within 12 (twelve) months, and consummated as promptly as reasonably practicable, after such acquisition, and provided further that Seller shall give Purchaser the opportunity to participate in the process with respect to the sale of that portion of such acquired business engaged in such Restricted Activity and shall consider in good faith any bona fide offer made by Purchaser with respect thereto, unless the purchaser to

whom the acquired business engaged in such Restricted Activity is sold initiated the transaction in respect of which Seller's acquisition took place;

- (c) manufacturing any products for its own use or for any member of Purchaser's Group;
- (d) continuing to operate the Seller's Retained Starch Activities; and
- (e) purchasing and incorporating Specialty Starch as a raw material or ingredient into products which are not themselves a Specialty Starch.

12.2 Restriction on Purchaser

Purchaser's Group shall not during a period commencing on the Signing Date and ending 2 (two) years after the Closing Date hire or seek to hire any person employed immediately prior to the Signing Date in Seller's Group with a base salary (or equivalent) in excess of USD 135,000 (one hundred thirty-five thousand US dollars), and with whom Purchaser has had contact in the course of the Transaction, provided that the placing of an advertisement of a post available to a member of the public generally and the recruitment of a person through an employment agency shall not constitute "seeking to hire" for purposes of this Clause 12.2, provided that no member of Purchaser's Group instructs or encourages such agency to approach any such person. The restriction in this Clause 12.2 shall not operate to prohibit any member of Purchaser's Group from exercising any right or fulfilling any obligation pursuant to this Agreement or any agreement entered into pursuant to this Agreement.

12.3 Specific performance

The Parties agree that irreparable damage is likely to occur if any of the provisions set out in Clause 12.1 or 12.2 were breached. Accordingly, each Party shall be entitled, subject to the discretion of the relevant competent judicial body, to an injunction or injunctions to prevent any breach or threatened breach of Clauses 12.1 or 12.2, without the requirement of posting bond, and to enforce specifically the provisions set out therein, such remedy being, subject to Clause 14.7, in addition to any other remedy available at Law.

13 CONFIDENTIALITY

13.1 Announcements

No public announcement or circular in connection with the existence or the subject matter of this Agreement shall be made or issued by or on behalf of any member of Seller's Group or Purchaser's Group without the prior written approval of Seller and Purchaser. This shall not affect any announcement or circular required by Law or the rules of any recognised stock exchange on which the securities of either Party or any Affiliate of such Party are listed, provided that the Party with an obligation to make an announcement or issue a circular shall consult with the other Party insofar as is reasonably practicable before complying with such an obligation.

13.2 Confidentiality undertaking

- 13.2.1 The Confidentiality Agreement shall cease to have any force or effect from Closing.
- 13.2.2 Subject to Clause 13.1 and Clause 13.2.3, each of the Parties shall treat as strictly

41

Confidential

Execution version

confidential and not disclose or use any information contained in or received or obtained as a result of entering into or pursuant to this Agreement (or any agreement entered into pursuant to this Agreement) that relates to:

- (a) the provisions of this Agreement or any agreement entered into pursuant to this Agreement;
- (b) the negotiations relating to this Agreement (or any such other agreement); or
- (c) the other Party to this Agreement or the business carried on by such other Party or such other Party's group of companies.
- 13.2.3 After the Closing Seller's Group shall treat as confidential and not disclose or use any confidential information to the extent related to the Group or the Operations.
- 13.2.4 Clauses 13.2.2 and 13.2.3 and the Confidentiality Agreement shall not prohibit disclosure or use of any information if and to the extent:
 - (a) the disclosure or use is required by Law or any recognised stock exchange on which securities of any Party or an Affiliate of such Party are listed;
 - (b) the disclosure or use is reasonably necessary to consummate the Transaction, including in connection with the arrangement of any financing by Purchaser;
 - (c) the disclosure or use is required for the prosecution or defence in any judicial proceedings of any claims arising out of this Agreement or any other agreement entered into under or pursuant to this Agreement or the disclosure is made to a Tax Authority in connection with the

Tax affairs of the disclosing Party;

- (*d*) the disclosure is made to professional advisors of any Party on terms consistent with the provisions of Clause 13.2.2 in respect of such information;
- (e) the information is or becomes publicly available (other than by breach of the Confidentiality Agreement or of this Agreement);
- (f) the other Party has given prior written approval to the disclosure or use; or
- (g) the information is independently developed after Closing, without reference to any other such information;

provided that prior to disclosure or use of any information pursuant to Clause 13.2.4(a)or (b), the Party concerned shall promptly notify the other Party of such requirement with a view to providing the other Party with the opportunity to contest such disclosure or use or otherwise to agree to the timing and content of such disclosure or use.

13.2.5 Clause 13.2.2 shall not prohibit disclosure or use by any member of Purchaser's Group after the Closing of any information received from Seller or its Representatives relating exclusively to the Operations, the Group Companies or the Businesses.

42

Confidential

Execution version

4 MISCELLANEOUS

14.1 Further assurances

- 14.1.1 From and after the Closing Date, each of the Parties shall from time to time execute such documents and perform such acts and things as the other Party may reasonably require to transfer the Shares, US Interests and Businesses to the Share Purchasers, Business Purchasers and Purchaser, and to give any Party the full benefit of this Agreement.
- 14.1.2 If, at any time after the Closing, any deeds, bills of sale, assignments, assurances or any other actions or things are reasonably necessary to vest, perfect or confirm of record or otherwise in the relevant member of Purchaser's Group, the right, title or interest in, to or under any of the Business Assets or the Shares or US Interests as sold under this Agreement, Seller shall, and shall cause the Share Seller, US Seller and each Business Seller to, execute and deliver all such deeds, bills of sale, assignments and assurances and to take and do all such other actions and things as may be reasonably necessary to vest, perfect or confirm or defend any and all such right, title and interest in, to and under the Business Assets or the Shares or US Interests in the Share Purchasers and Business Purchasers, as applicable.

14.2 Whole agreement

- 14.2.1 This Agreement and, until the Closing Date, the Confidentiality Agreement contain the whole agreement between the Parties relating to the subject matter of this Agreement, to the exclusion of any terms implied by Law which may be excluded by contract, and supersedes any previous written or oral agreement between the Parties in relation to the matters dealt with in this Agreement.
- 14.2.2 Each Party acknowledges that it has not been induced to enter this Agreement by any representation, warranty or undertaking not expressly set out in this Agreement.

14.3 No assignment

Except as otherwise expressly provided in this Agreement, neither Party may, unless with the prior written consent of the other Party, assign, grant any security interest over or otherwise transfer, in whole or in part, any of its rights and obligations under this Agreement.

14.4 Waiver

No waiver of any provision of this Agreement shall be effective unless in writing and signed by or on behalf of the waiving Party.

14.5 Variation

No variation of this Agreement shall be effective unless in writing and signed by or on behalf of each of the Parties.

14.6 Third party rights

Save as expressly otherwise stated, this Agreement does not contain a stipulation in favour of a third party ('derdenbeding').

43

Confidential

14.7 Rescission

Without prejudice to Clauses 4.3.3, 6.2.3 and 6.4.1 and except in the event of fraud on the part of Seller, Share Seller, the US Seller or any Business Seller, or Purchaser, Share Purchaser, the US Purchaser or any Business Purchaser, each Party waives its right to rescind (`ontbinden') this Agreement on the basis of section 6:265 of the Netherlands Civil Code. The mistaken party shall bear the risk of any mistake (`dwaling') in making this Agreement.

14.8 Method of payment

- 14.8.1 Wherever in this Agreement provision is made for a payment to be made or procured by Seller to Purchaser, Seller shall arrange that such payment shall be made by Seller for itself and on behalf of the relevant member of Seller's Group to Purchaser for itself and on behalf of the relevant member of Purchaser's Group.
- 14.8.2 Wherever in this Agreement provision is made for a payment to be made or procured by Purchaser to Seller, Purchaser shall arrange that such payment shall be made by Purchaser for itself and on behalf of the relevant member of Purchaser's Group to Seller for itself and on behalf of the relevant member of Seller's Group.
- 14.8.3 Any such payments shall be effected by crediting for same day value the account specified by Seller or Purchaser, as the case may be, on behalf of the Party entitled to the payment (reasonably in advance and in sufficient detail to enable payment by telegraphic or other electronic means to be effected) on or before the due date for payment.
- 14.8.4 Payment of a sum in accordance with this Clause 14.8 shall be a good discharge to the payer (and those on whose behalf such payment is made) of its obligation to make such payment and the payer (and those on whose behalf such payment is made) shall not be obliged to see to the application of the payment as between those on whose behalf the payment is received.

14.9 Costs and transfer Taxes

Unless this Agreement specifically provides otherwise, all costs which a Party has incurred or must incur in preparing, concluding or performing this Agreement are for its own account. Except in relation to VAT, to which Clause 3.4 and Schedule 12 shall apply, all stamp, transfer, registration, sales, real property transfer and other similar Taxes, duties, fees and charges and all notarial fees payable in connection with the sale, transfer or purchase of the Shares, the US Interests, the Businesses and any Business Asset or Business Liability under this Agreement shall be paid by Purchaser or if paid by Seller, reimbursed by Purchaser to Seller upon delivery to Purchaser of proof of Seller's payment of such Taxes, duties, fees, charges and notarial fees.

14.10 Interest

If any Party defaults in the payment when due of any sum payable under this Agreement, the liability of that Party shall be increased to include interest on such sum from the date when such payment is due until the date of actual payment (as well after as before judgement) at the Interest Rate.

44

Confidential

Execution version

14.11 Notices

- 14.11.1 Any notice in connection with this Agreement (a "Notice") shall be:
 - (a) in writing;
 - (b) in English; and
 - (c) delivered by hand, registered post or by courier using an internationally recognised courier company.
- 14.11.2 A Notice to Seller shall be sent to Seller at the following address, or such other person or address as Seller may notify to Purchaser from time to time in accordance with this Clause 14.11:

Akzo Nobel N.V.

Strawinskylaan 2555, 1077 ZZ Amsterdam, the Netherlands

Attention: General Counsel

14.11.3 A Notice to Purchaser shall be sent to Purchaser at the following address, or such other person or address as Purchaser may notify to Seller from time to time in accordance with this Clause 14.11:

Corn Products International, Inc

5 Westbrook Corporate Center

Westchester, Illinois 60154, United States of America

Attention: General Counsel

As a matter of courtesy, a copy of the notice (which copy shall not constitute notice) shall be sent for information purposes to:

Sidley Austin LLP

One South Dearborn

Chicago, Illinois 60603 U.S.A. Attention: John M. O'Hare

Failure to send such copy shall not affect the validity of the notice.

- 14.11.4 A Notice shall be effective upon receipt and shall be deemed to have been received at the time of delivery, if delivered by hand, registered post or courier.
- 14.11.5 Each of the Parties further agrees that service of any process, summons, notice or document by any national or international, as applicable, express courier service to such Party at its respective address set out in this Clause 14.11 shall be effective service of process for any action, suit or proceeding required in order to enforce Clause 14.15 or any arbitral decision thereunder.

14.12 Invalidity

If any provision in this Agreement shall be held to be illegal, invalid or unenforceable, in whole or in part, under any Law:

14.12.1 such provision or part shall to that extent be deemed not to form part of this Agreement but the legality, validity or enforceability of the remainder of this Agreement shall not be

45

Confidential

Execution version

affected:

14.12.2 Seller and Purchaser shall use reasonable good faith efforts to agree on a replacement provision that is legal, valid and enforceable to achieve so far as possible the intended effect of the illegal, invalid or unenforceable provision.

14.13 Counterparts

This Agreement may be entered into in any number of counterparts, all of which taken together shall constitute one and the same instrument. Seller and Purchaser may enter into this Agreement by signing any such counterpart.

14.14 Notary

Seller is aware that the Notary is a civil law notary working at De Brauw Blackstone Westbroek N.V., the law firm acting as Seller's Lawyers. With reference to the provisions of the Code of Conduct ('Verordening Beroeps- en Gedragsregels') of the Royal Notarial Regulatory Body ('Koninklijke Notariële Beroepsorganisatie'), Purchaser acknowledges and agrees that De Brauw Blackstone Westbroek N.V. may assist and act on behalf of Seller in connection with this Agreement including any disputes arising in relation to this Agreement.

14.15 Dispute resolution

With the exception of disputes with regard to the determination of the Working Capital Statements and the Net Debt Statements pursuant to Clause 7.1.3 which are to be negotiated and resolved as set out in Clause 7.1.3, any dispute arising out of or in connection with this Agreement, including any question regarding its existence, validity or termination, that the Parties do not resolve through direct negotiation shall be referred to and finally resolved by arbitration under the rules of the ICC International Court of Arbitration from time to time, which rules are deemed to be incorporated by reference into this Clause. The number of arbitrators shall be three. The seat, or legal place, of arbitration shall be, and any hearings required to be conducted with respect to the arbitral proceedings shall take place in, London, United Kingdom. The language to be used in the arbitral proceedings shall be English.

14.16 Governing law

This Agreement (including, for the avoidance of doubt, Clause 14.15) and the documents to be entered into pursuant to it, save as expressly otherwise provided therein, shall be governed by and construed in accordance with the Law of the Netherlands, without regard to any rule or provision regarding choice of law or conflict of law, provided that the United Nations Convention on Contracts for the International Sale of Goods shall not apply to this Agreement and the documents to be entered into pursuant to it.

[SIGNATURE PAGE FOLLOWS:]

46

Confidential

Execution version

AGREED AND SIGNED:

Akzo Nobel N.V.

/s/ Floris A.J. van Hall

Name: Floris A.J. van Hall

Name: Ben J. Schoordijk

Title: Attorney-in-fact

on 19 June 2010 for and on behalf of Purchaser:

Corn Products International, Inc.

/s/ Cheryl K. Beebe Name: Cheryl K. Beebe Title: Chief Financial Officer

On 19 June 2010 for and on behalf of Seller:

47

Confidential

Execution version

Schedule 1 Definitions

(Clause 1.1)

- "Accounting Principles" has the meaning defined in Paragraph 3 of Schedule 13 (Part 3);
- "Accounts" means the audited special-purpose combined statement of assets and liabilities of the Group as of December 31, 2009 and 2008, and the related special-purpose combined statements of operating activities and identifiable cash flows for the years then ended, set out in Schedule 16;
- "Accounts Date" means 31 December 2009;
- "Acquired Intellectual Property" means the Group Intellectual Property and the Business Intellectual Property;
- "Acquired Registered Intellectual Property" means the Registered Group Intellectual Property and any Registered Intellectual Property that is included in the Business Intellectual Property;
- "Affiliate" means, with respect to any person, any other person that, directly or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such person, it being understood that the Group Companies constitute Affiliates of Seller prior to the Closing (or any applicable Deferred Closing) and, unless subject to a Deferred Closing that has not yet occurred, the Group Companies constitute Affiliates of Purchaser from and after the Closing, and for purposes of this definition, "control" shall mean, as to any person, the power to direct or cause the direction of the management and policies of such person, whether through the ownership of voting securities, by contract or otherwise and the terms "controlled by," "under common control with" and "controlling" shall have correlative meanings;
- "Agreed Terms" means, in relation to a document, such document in the terms agreed between Seller and Purchaser and signed for identification by Seller's Lawyers and Purchaser's Lawyers with such alterations as may be agreed in writing between Seller and Purchaser from time to time;
- "Agreement" means this international share and business sale agreement and the schedules thereto;
- "Allocated Value" means, in relation to the Shares, the US Interests and to each Business, the amount specified in, or calculated in accordance with, Paragraph 1 of Schedule 11 (Part 1) in respect of the Shares, US Interests or such Business;
- "Anti-trust Approvals" means, in respect of any filings, notifications, applications or requests required or necessary under any statutory provisions in connection with the conclusion or performance of this Agreement, including those set out in Clause 4.1.1 or 4.1.2, the relevant Governmental Authority having stated in writing that the subject matter of the filing, notification, application or request is permitted, or that it shall not conduct any further investigation, or in the case of the HSR Act, that the waiting period applicable to the Transaction has expired or been terminated, and "Anti-trust Approval" means any one of them or the relevant one of them, as the

48

Confidential

Execution version

- "Automatic Transfer Employees" means all those persons who are immediately prior to Closing employed in the Operations by any member of Seller's Group (other than the Group Companies) and will automatically transfer with the Businesses by operation of Transfer Regulations;
- "Balance Sheet" means the balance sheet of the Group as of December 31, 2009 forming part of the Accounts;
- "Base Working Capital" means, in relation to the Company (including all other Group Companies other than the US Company and its Subsidiary), the US Company (including its Subsidiary) and each relevant Business, the amount specified in Schedule 11 (Part 2) in respect of the Company, the US Company or such Business and, in aggregate, USD 184,000,000 (one hundred eighty-four million US dollars);
- "Businesses" means the unincorporated businesses conducting part of the Operations and carried on by the Business Sellers, which businesses are sold under Clause 2.1.1(c)-(l), particulars of which are contained or referred to in Schedule 2 (Part 2), and "Business" means any one of them or the relevant one of them, as the context requires;
- "Business Assets" means all the rights and assets sold under Clause 2.1.1(b) through 2.1.1(l), and "Business Asset" means any one of them or the relevant one of them, as the context requires;
- "Business Books and Records" means the books and records of Seller's Group (excluding the Group Companies) to the extent relating to the Businesses, Properties or Operations, including sales, advertising and marketing materials and customer lists;
- "Business Contracts" means all Contracts entered into at or prior to Closing, by or on behalf of any of the Business Sellers primarily in relation the Operations, to the extent they remain executory or in force, but excluding:
- (a) employment and other Contracts with Employees;
- (b) Contracts of insurance relating to the Operations; and
- (c) Contracts (other than the Intra-Group Retained Agreements) between any of the Business Sellers on the one hand and any member of Seller's Group (excluding the Group Companies);
- and "Business Contract" means any of them or the relevant one of them, as the context requires;
- "Business Day" means a day which is not a Saturday, a Sunday, or a public holiday in the Netherlands or New York;
- "Business Intellectual Property" means any Intellectual Property that is owned by a Business Seller and primarily used by any Business Seller in the Operations;
- "Business Inventory" means, in relation to each Business, all raw materials, consumables, work in progress, part-processed stocks, finished goods, goods for resale and stock in transit,

49

Confidential

Execution version

wherever located, of the relevant Business Seller and physically allocated or separated for use primarily in any part of the Operations, excluding, for the avoidance of doubt, any such inventory owned by any of the Group Companies;

- "Business Leased Properties" means the Properties set out in Paragraph 1(1) of Schedule 3 (Part 2), and "Business Leased Property" means any one of them or the relevant one of them, as the context requires;
- "Business Liabilities" means all Liabilities (other than any Excluded Liabilities, any amount in respect of VAT or for which Purchaser or any member of Purchaser's Group may be or become liable to pay to Seller or any member of Seller's Group pursuant to Schedule 12, any Liability for Tax taken into account in the computation of any amount payable to or by Seller pursuant to Clause 6.5.5, or any Liabilities (including any Liability for Tax) for which Seller is liable under an indemnity set out in this Agreement), any Intra-Group Indebtedness and any Third-Party Indebtedness) incurred by the relevant Business Seller to the extent related to the Operations and existing at Closing or arising, accruing or assessed after Closing whether known or unknown, reported or unreported, in respect of any period or in consequence of any transaction carried out in carrying on the Operations prior to Closing, and "Business Liability" means any one of them or the relevant one of them, as the context requires;
- "Business Owned Properties" means the Properties set out in Paragraph 1(1) of Schedule 3 (Part 1), and "Business Owned Property" means any one of them or the relevant one of them, as the context requires;
- "Business Permits" means, in relation to each Business, all licences, permits, consents, approvals, authorizations, registrations, franchises, exemptions and orders of any Governmental Authority at Closing and relating primarily to the Operations;
- "Business Properties" means the Business Owned Properties and the Business Leased Properties, and "Business Property" means any one of them or the relevant one of them, as the context requires;
- "Business Purchaser" means, in relation to each Business or specified Business Asset referred to in Schedule 2 (Part 2) column (2):
- (a) the relevant person or company identified in Schedule 2 (Part 2) column (3); or

(b) the relevant person or company nominated by Purchaser in accordance with Clause 2.3.1;

and "Business Purchasers" means all such persons and companies;

"Business Receivables" means all amounts receivable, accrued or prepaid by or owed to the relevant Business Seller in relation to the Operations in respect of trading debtors, excluding any cash in hand, cash in transit, cash at bank and any current investments held by or on behalf of the Business Seller:

"Business Seller" means, in relation to each Business or specified Business Asset referred to in Schedule 2 (Part 2) column (2), the relevant person or company identified in Schedule 2 (Part 2) column (1), and "Business Sellers" means all such persons and companies;

50

Confidential

Execution version

"Cash Balances" means any cash in hand, cash in transit, cash at bank and any current investments held at the Effective Time by or on behalf of any of the Group Companies, as derived from the relevant Net Debt Statement and included in line items under the heading "Cash Balances" in Paragraph 3 of Schedule 13 (Part 3);

"Claiming Party" has the meaning set out in Clause 11.5;

"Claims" means all rights and claims arising at any time whether before or after Closing to the extent related to any Business Asset or any Business Liability, but excluding any rights or claims under the Insurance Policies that are not Continuing Policies, and "Claim" means any one of them or the relevant one of them, as the context requires;

"Closing" means, in respect of the sale of the Shares, the US Interests and each particular sale of Business, the performance of the actions set out in Clause 6.2.3 to the extent relating to the Shares, US Interests or Business, as the case may be, provided that a reference to circumstances or events at or after Closing, or to periods before or after Closing, shall be a reference to each Closing but then only in relation to the Company and/or the US Company, if applicable, or to the Business or Group Company which is the subject of such Closing;

"Closing Date" means the date on which Closing has taken place in respect of the Shares and the US Interests;

"Closing Payment Amount" has the meaning set out in Clause 6.2.1;

"Code" means the U.S. Internal Revenue Code of 1986, as amended;

"Combined Tax Group" has the meaning set out in Schedule 14 (Part 1);

"Committed Financing Facility" means the bridge facility referred to in the Commitment Letter;

"Commitment Letter" means the financing commitment letter attached hereto as Schedule 19;

"Company" means Brunob II, B.V., a private limited liability company incorporated under the laws of the Netherlands, with corporate seat in Arnhem, the Netherlands, and registered address at Velperweg 76, Arnhem, the Netherlands;

"Company Assets" means all of the assets of the Group Companies;

"Competition Laws" means all statutes, rules, regulations, orders, decrees, administrative guidance and judicial doctrines and other Laws that are designed or intended to regulate mergers or other business combinations (requiring pre-closing or post-closing authorizations, consents or judgments) or that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade;

"Conditions Precedent" means the conditions set out in Clause 4.1, and "Condition Precedent" means any one of them or the relevant one of them, as the context requires;

"Confidentiality Agreement" means the confidentiality agreement dated 22 May 2009 between Seller and Purchaser;

"Constitutional Documents" means, with respect to any company, such company's deed or

51

Confidential

Execution version

memorandum of incorporation, charter, articles of association, bylaws and other organizational documents, as applicable;

"Continuing Policies" has the meaning set out in Clause 8.7.1;

"Contracts" means all contracts, agreements, deeds, leases, licenses, and other legally binding undertakings and commitments, whether written or oral, and "Contract" means any of them or the relevant one of them, as the context requires;

"Customer Field" means:

- (a) non-pharmaceutical products for application to human hair or skin, including products that are regulated as over-the-counter drugs under applicable law, other than wound care products;
- (b) non-ingestible products for oral care, but excluding chewing gum;
- (c) products for modifying the properties including rheology, and the performance including cleaning, fabric softness and hand feel in fabric softness and other laundry products for the consumer, industrial and institutional markets and products for the controlled delivery of fragrances and other functional ingredients in detergents for consumer, industrial and institutional markets, and for detergent building in detergents for consumer, industrial and institutional markets;
- (*d*) products for home cleaning and for hard surface cleaning in consumer, industrial and institutional markets, and encapsulation products for fragrances and active ingredients in animal litter, but not including products to improve absorbency in animal litter;
- (e) linerboard and paperboard coating products, which for the avoidance of doubt shall not include products for use at the wet end and size press of the papermaking process;
- (f) products for use in water treatment applications and systems;
- (g) products for use in oil and gas exploration, drilling, stimulation and production including refinery operations and products;
- (h) products for use in agricultural applications and in the agrochemical and agricultural industries including seed and plant treatments (which, for the avoidance of doubt, shall not include products for use in food processing), other than (i) biodegradable products for use in protective preformed sheets for use in crop fields and (ii) ingredients for animal feed and for use in aquaculture;
- (i) products for use in mining applications and in the mining industry;
- (i) products for use in asphalt applications and the asphalt industry; and
- (k) any such other products as may be agreed from time to time by Seller and Purchaser;
- "Data Room" means the electronic data room containing documents and information relating to the Group, made available by Seller, the contents of which are listed in Appendix 1 to Schedule 15 (Part 2);

"Data Room Information" means (a) that information and those documents in the Data Room

52

Confidential

Execution version

which Purchaser was able to print as of the day prior to the Signing Date, (b) that information contained in written summaries of documents in the so-called "clean room" portion of the Data Room which Seller permitted to be shared with Representatives of Purchaser (other than counsel to Purchaser), (c) that information contained in the written correspondence, including attachments where applicable, between Representatives of Purchaser and Seller attached to the Disclosure Letter, and (d) that information contained in the public registries searched by Purchaser or any Representative of Purchaser prior to Signing;

"Deferred Closing" has the meaning set out in Clause 6.5.4;

"Deferred Condition" has the meaning set out in Clause 6.5;

"Deferred Jurisdiction" has the meaning set out in Clause 6.5;

"Disclosure Letter" means the letter from Seller to Purchaser, attached as Schedule 15 (Part 2);

"Disputed Item" has the meaning set out in Clause 7.1.3(c)(ii);

"Effective Time" means the end of the day on the day immediately preceding the Closing Date;

"Employees" means all those persons who are immediately prior to Closing employed by a Group Company, together with the Automatic Transfer Employees;

"Employment Costs" has the meaning set out in Schedule 6;

"Employment Liabilities" has the meaning set out in Schedule 6;

"Encumbrance" means any charge, pledge, mortgage, lien, option, equity, power of sale, hypothecation, usufruct, retention of title, right of preemption, right of first refusal or other similar third party rights or security interest of any kind or an agreement to create any of the foregoing;

"Environment," "Environmental Authority," "Environmental Law" and "Environmental Permit" have the meanings set out in Paragraph 10(a) of Schedule 15 (Part 1);

"ERISA" means the United States Employee Retirement Income Security Act of 1974, as amended;

"ERISA Affiliate" means any trade or business (whether or not incorporated) which, together with Seller or any of its Affiliates, would be treated as a single employer under section 414 of the Code or section 4001 of ERISA;

"Estimated Cash Balances" means Seller's good faith best estimate of the aggregate of the Cash Balances as of the Effective Time, as set out in the Estimated Net Debt Statements;

"Estimated Intra-Group Indebtedness" means, in relation to each Group Company, the aggregate of the Estimated Intra-Group Payables less the Estimated Intra-Group Receivables of said Group Company;

"Estimated Intra-Group Payables" means, in relation to each Group Company, Seller's good faith best estimate of the Intra-Group Payables of said Group Company as of the Effective Time,

53

Confidential

Execution version

as set out in the relevant Estimated Net Debt Statement;

"Estimated Intra-Group Receivables" means, in relation to each Group Company, Seller's good faith best estimate of the Intra-Group Receivables of said Group Company as of the Effective Time, as set out in the relevant Estimated Net Debt Statement;

"Estimated Net Debt" means:

- (a) in relation to the Company (including all other Group Companies other than the US Company and its Subsidiary) the aggregate (which may be a positive or a negative number) of the following:
 - (i) Estimated Intra-Group Indebtedness of each Group Company other than the US Company and its Subsidiary; plus
 - (ii) Estimated Third Party Indebtedness of each Group Company other than the US Company and its Subsidiary; less
 - (iii) Estimated Cash Balances of each Group Company other than the US Company and its Subsidiary;
- (b) in relation to the US Company (including its Subsidiary) the aggregate (which may be a positive or a negative number) of the following:
 - (i) Estimated Intra-Group Indebtedness of the US Company and its Subsidiary; plus
 - (ii) Estimated Third Party Indebtedness of the US Company and its Subsidiary; less
 - (iii) Estimated Cash Balances of the US Company and its Subsidiary; and
- (c) in relation to the Group, the sum of all amounts calculated under items (a) and (b);

"Estimated Net Debt Statement" means, in relation to each of the Company (including all other Group Companies other than the US Company and its Subsidiary) and the US Company (including its Subsidiary), the statement of Estimated Net Debt to be prepared in accordance with Clause 5.4;

"Estimated Purchase Price" has the meaning set out in Clause 3.2;

"Estimated Third Party Indebtedness" means Seller's good faith best estimate of the Third-Party Indebtedness as of the Effective Time, as set out in the Estimated Net Debt Statements;

"Estimated Working Capital" means, in relation to the Company (including all other Group Companies other than the US Company and its Subsidiary), the US Company (including its Subsidiary) and each relevant Business, Seller's good faith best estimate of its Working Capital as of the Effective Time, as set out in the Estimated Working Capital Statement;

"Estimated Working Capital Adjustment" means, in relation to the Company (including all other Group Companies other than the US Company and its Subsidiary), the US Company (including its Subsidiary) and each relevant Business, the amount by which its Estimated Working Capital is greater than its Base Working Capital (in which case such amount shall be expressed as a

Confidential

positive figure) or by which it is less than its Base Working Capital (in which case such amount shall be expressed as a negative figure), and, in relation to the Group, the sum of all such amounts;

"Estimated Working Capital Statement" means, in relation to the Company (including all other Group Companies other than the US Company and its Subsidiary), the US Company (including its Subsidiary) and each relevant Business, the statement of Estimated Working Capital to be prepared in accordance with Clause 5.4:

"EU Regulation" has the meaning set out in Clause 4.1.1;

"Excess Cash Amount" has the meaning set out in Clause 6.3.2;

"Excluded Liabilities" means:

(a) any Liability of any past or present member of Seller's Group or its predecessors, to the extent such Liability (i) is a Liability (whether known or unknown) of a Group Company at Closing or is a Liability (whether known or unknown) of a Business Seller at Closing and does not relate to the Operations, or, without detracting from the generality of the foregoing, (ii) relates to any of the Seller's Retained Starch Activities, or (iii) relates to the adhesives, electronic and engineering materials or specialty polymers businesses formerly owned and operated by past or present members of Seller's Group or their predecessors;

(b) any Liability that relates to those matters set out in section $[\cdot]$ of the Data Room;

but excluding:

(c) any Liability (including any Liability for Tax) for which Seller is liable under an indemnity set out in this Agreement other than pursuant to Clause 8.2;

"Final DBO Adjustment Amount" has the meaning set out in Paragraph 2.4 of Schedule 7 (Part 1);

"Financial Statements" has the meaning set out in Clause 5.5.1;

"Financing" has the meaning set out in Clause 5.6.1;

"GAAP" means United States generally accepted accounting principles;

"Goodwill" means the goodwill of the Business Sellers in relation to the Operations, together with the goodwill attaching to any registered or unregistered trademarks, together with the exclusive right (so far as the Business Sellers can grant the same) for the Business Purchasers to represent themselves as carrying on the Operations and the business of the Group under the name "National Starch" or any other similar name in succession to the Business Sellers;

"Governmental Authority" means, to the extent it has jurisdiction, any supranational governmental commission, council, directorate, court, trade agency, regulatory body or other authority, or any national government, any legislature, any political subdivision of a national government or of any state, county, province or local jurisdiction therein, or any agency or instrumentality of any such government or political subdivision;

55

Confidential

Execution version

"Group" means the Group Companies and the Businesses, taken as a whole;

"Group Companies" means the Company, the US Company and the Subsidiaries, and "Group Company" means any one of them or the relevant one of them, as the context requires;

"Group Contracts" means all Contracts to which any Group Company is a party and all Business Contracts, and "Group Contract" means any one of them or the relevant one of them, as the context requires;

"Group Intellectual Property" means any Intellectual Property owned by any Group Company;

"Group Liabilities" means all Liabilities (whether known or unknown) of any member of Seller's Group, suffered or incurred after Closing, to the extent that any such Liability (a) is a Liability of a Group Company at Closing, (b) is a Business Liability at Closing, or (c) otherwise relates to the Operations (including the assets used therein) as conducted at any time prior to Closing, by any past or present member of Seller's Group or their predecessors, but not to the extent that any such Liability is an Excluded Liability or a Liability (including a Liability for Tax) for which Seller is liable under an indemnity set out in this Agreement.

"Group Material Adverse Effect" means any fact, circumstance or change (including any loss of, or adverse change in the relationship with, Employees, customers or suppliers of the Operations that Purchaser proves is not proximately caused by the pendency or announcement of the Transaction or any other transactions or actions contemplated by this Agreement) that, individually or when taken together with all other such facts, circumstances, changes that exist at the date of determination, has or is reasonably likely to have a material adverse effect on the Operations taken as a whole; provided, however, that facts, circumstances or changes or effects resulting from, relating to or arising out of the following shall not be taken into account when determining whether a Group Material Adverse Effect has occurred or is reasonably likely to occur: (i) general market or economic

conditions provided that such conditions do not have a materially disproportionate impact on the Operations, as compared to other similar companies in industries in which the Group operates; (ii) general market or economic conditions in the industries in which the Group conducts the Operations provided that such conditions do not have a materially disproportionate impact on the Operations, as compared to other similar companies in industries in which the Group operates; (iii) currency exchange rate fluctuations; (iv) political conditions (including changes arising out of acts of terrorism, sabotage, armed hostilities or war), crop diseases or weather conditions or other force majeure events; (v) changes in Law or changes in the accounting principles pursuant to which the Group is required to change its prior accounting policies or practices; and (vi) any loss of, or adverse change in the relationship with, Employees, customers or suppliers of the Operations proximately caused by the pendency or announcement of the Transaction or any other transactions or actions contemplated by this Agreement;

"Guarantee" means any guarantee, indemnity, surety, letter of comfort or other assurance, security, right of set-off, obligation to contribute ('bijdrageplicht') or undertaking, given by a person to secure or support the obligations (actual or contingent) of any other person (including any right of recourse or similar liability under applicable Law), whether given directly, by way of

56

Confidential

Execution version

counter-indemnity or otherwise;

"Hazardous Substances" has the meaning set out in Paragraph 10(a) of Schedule 15 (Part 1);

"Hive-out Agreements" means the Master NPG Agreement and the Transaction Documents (as defined in the Master NPG Agreement);

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvement Act of 1976, as amended;

"IFRS" means International Financial Reporting Standards, as adopted by the European Union and as amended from time to time;

"Insurance Policies" means all insurance policies in effect at Signing in which a Group Company or Business Seller (in relation to a Business) is an insured party;

"Intellectual Property" means Registered Intellectual Property and Unregistered Intellectual Property;

"Interest Rate" means 3% (three percent) per annum accrued on a daily basis;

"Interim Holdco" has the meaning set out in Clause 6.5.2;

"Intra-Group Indebtedness" means, in relation to each Group Company, the aggregate amount of the Intra-Group Payables minus the aggregate amount of the Intra-Group Receivables of said Group Company, which aggregate amount may be a positive or negative number;

"Intra-Group Payables" means, in relation to each Group Company, all amounts owed by said Group Company to a member of the Seller's Group (other than the Group Companies) in respect of loans and other indebtedness, including dividends, Intra-Group Service Fees owed by such Group Company and amounts due in respect of Taxation paid by another member of the Seller's Group on its behalf (together in each case with accrued interest thereon), at the Effective Time (but excluding any Intra-Group Trading Payables and any item to be included in calculating the Cash Balances of said Group Company) or the Third-Party Indebtedness of said Group Company, as derived from the relevant Net Debt Statement and included in line items under the heading "Intra-Group Payables" in Schedule 13 (Part 2);

"Intra-Group Receivables" means, in relation to each Group Company, all amounts owed by a member of the Seller's Group (other than the Group Companies) to said Group Company in respect of loans and other indebtedness, including Intra-Group Service Fees owed to such Group Company at the Effective Time and including amounts due in respect of Taxation payable to or recoverable (by credit or otherwise) by another member of the Seller's Group but which amounts are attributable to the activities of such Group Company (together, in each case, with accrued interest thereon), at the Effective Time (but excluding any Intra-Group Trading Receivables and any item to be included in calculating the Cash Balances of said Group Company) or the Third-Party Indebtedness of said Group Company, as derived from the relevant Net Debt Statement and included in line items under the heading "Intra-Group Receivables" in Paragraph 3 of Schedule 13 (Part 3);

"Intra-Group Retained Agreements" means the agreements listed in Schedule 5;

57

Confidential

Execution version

"Intra-Group Service Fees" means, in relation to each relevant Group Company, any amount owed by or to any member of Seller's Group in respect of royalties, management services and research (including, in each case, such part of such amount as relates to VAT);

"Intra-Group Trading Payables" means the aggregate amount that members of Seller's Group (excluding the Group Companies) are entitled to claim from any Group Company for payment of amounts owed in respect of any trading debt between such members of Seller's Group and the Group Companies, including amounts pursuant to Intra-Group Retained Agreements, as at the Effective Time, as are described in Schedule 13 (Part 5);

"Intra-Group Trading Receivables" means the aggregate amount that the Group Companies are entitled to claim from Seller's Group (other than the Group Companies) for payment of amounts owed in respect of any trading debt between such members of Seller's Group and the Group Companies, including amounts pursuant to Intra-Group Retained Agreements, as at the Effective Time, as are described in Schedule 13 (Part 5);

"IP Licence" means the licence attached hereto as Schedule 10;

"Know-how" means confidential and proprietary industrial and commercial information and techniques in any form including drawings, formulae, test results, reports, project reports and testing procedures, instructions and training manuals, tables of operating conditions, market forecasts, list and particulars of customers and suppliers;

"Law" means any applicable statute, law, ordinance, rule, regulation, order, judgment or decree of any Governmental Authority;

"Liabilities" means all liabilities, duties and obligations of every description, whether deriving from Contract, common law, statute or otherwise, whether present or future, actual or contingent, ascertained or unascertained or disputed and whether owed or incurred severally or jointly or as principal or surety, and "Liability" means any one of them or the relevant one of them, as the context requires;

"Local Transfer Documents" and "Local Transfer Document" have the meanings set out in Clause 2.5.1;

"Losses" means all damage, losses, costs (including reasonable legal costs and reasonable experts' and consultants' fees), charges, expenses, claims and demands:

"Master NPG Agreement" means that certain Master Intra-Group Sale and Purchase Agreement relating to the NPG Business, dated April 1, 2008, between Seller and the Company;

"Material Contract" has the meaning set out in Paragraph 7 of Schedule 15 (Part 1);

"Member States" means the member states of the European Union from time to time, and "Member State" means any of them or the relevant one of them, as the context requires;

"Moveable Assets" means, in relation to each Business, all tangible property (including plant and machinery, vehicles, testing and other equipment, and spare parts for any such machinery and equipment, and furniture), excluding Business Property and Business Inventory, of the relevant

58

Confidential

Execution version

Business Seller and used primarily in the Operations at Closing excluding, for the avoidance of doubt, any such tangible property owned by any of the Group Companies;

"Net Debt" means:

- (a) in relation to the Company (including all other Group Companies other than the US Company and its Subsidiary) the aggregate (which may be a positive or a negative number) of the following:
 - (i) Intra-Group Indebtedness of each Group Company other than the US Company and its Subsidiary; plus
 - (ii) Third Party Indebtedness of each Group Company other than the US Company and its Subsidiary; less
 - (iii) Cash Balances of each Group Company other than the US Company and its Subsidiary;
- (b) in relation to the US Company (including its Subsidiary) the aggregate (which may be a positive or a negative number) of the following:
 - (i) Intra-Group Indebtedness of the US Company and its Subsidiary; plus
 - (ii) Third Party Indebtedness of the US Company and its Subsidiary; less
 - (iii) Cash Balances of the US Company and its Subsidiary; and
- (c) in relation to the Group, the sum of all amounts calculated under items (a) and (b).

"Net Debt Statement" means, in relation to each of the Company and the US Company, the statement of Net Debt prepared in accordance with Clause 7.1:

"Non-Automatic Transfer Employees" means all those persons who are immediately prior to Closing employed primarily in the Operations by any member of Seller's Group (other than the Group Companies) but will not transfer with the Businesses by operation of the Transfer Regulations;

"Non-Cleared Assets" has the meaning set out in Clause 6.5.6;

"Non-EU Country" means any country that is not a Member State;

"Non-Material Business" has the meaning set out in Clause 6.5.5;

"Non-Material Group Company" has the meaning set out in Clause 6.5.2;

"Notary" means civil law notary Professor Martin van Olffen, or any other civil law notary of De Brauw Blackstone Westbroek N.V., or such notary's substitute;

"Notice" has the meaning set out in Clause 14.11.1;

"Notice of Disagreement" has the meaning set out in Clause 7.1.3(a);

"NPG Contracts" means the Contracts (as defined in the Master NPG Agreement) included in the NPG Business Assets (as defined in the Master NPG Agreement);

"Offer Value" means USD 1,300,000,000 (one billion three hundred million US dollars);

59

Confidential

Execution version

"Operations" means the business conducted by the Group Companies, combined with the business activities conducted by the Business Sellers to the extent such activities comprise the research and development into and production, marketing, sale and distribution of specialty starches on a worldwide basis, principally focused on the food industry and also serving papermaking markets and the supply of industrial starch applications, but excluding any other line of business conducted by any Group Company or Business Seller and Seller's Retained Starch Activities;

"Other Party" has the meaning set out in Clause 11.5;

"Outside Date" has the meaning set out in Clause 4.2.2(c);

"Parties" means Seller and Purchaser, and "Party" means any one of them or the relevant one of them, as the context requires;

"Payee Party" has the meaning set out in Clause 10.10.2;

"Payor Party" has the meaning set out in Clause 10.10.2;

"Permitted Encumbrances" means (i) security interests arising in the ordinary course of business or by operation of Law and not attributable to the borrowing of money that do not affect the usage of the underlying asset, (ii) security interests arising under sales contracts with title retention provisions and equipment leases with third parties and entered into in the ordinary course of business, (iii) security interests for Taxes and other governmental charges which are not due and payable or which may be paid without penalty, and (iv) any encumbrances disclosed in section [·] of the Disclosure Letter;

"Properties" has the meaning set out in Paragraph 5(a)(i) of Schedule 15 (Part 1), and "Property" means any one of them or the relevant one of them, as the context requires;

"Purchase Price" has the meaning set out in Clause 3.1;

"Purchaser" has the meaning set out in the preamble of this Agreement;

"Purchaser's Group" means Purchaser and its subsidiaries from time to time, including, after Closing, the Group Companies;

"Purchaser's Lawyers" means Sidley Austin LLP at One South Dearborn Street, Chicago, Illinois 60603, United States of America

"Purchaser's Warranties" means the warranties given by Purchaser to Seller pursuant to Clause 9.4 and Schedule 15 (Part 4), and "Purchaser's Warranty" means any one of them or the relevant one of them, as the context requires;

"Registered Group Intellectual Property" means any Registered Intellectual Property that is owned by a Group Company;

"Registered Intellectual Property" means all of the following items: (i) patents, including reissued patents, re-examined patents, patent applications of all types, and any patents which may be filed or granted therefor and all rights to claim priority in whole or in part on the basis of

60

Confidential

Execution version

any of the aforementioned patents or applications; (ii) registered trademarks, trade names or service marks, applications to register trademarks, trade names or service marks, registered trade dress, applications to register trade dress, intent to use applications or other applications or registrations related

to trade identity and trademarks; (iii) registered copyrights and applications for copyright registration in any country; (iv) mask work registrations and applications to register mask works in any country; (v) domain name registrations and applications to register domain names in any country; and (vi) any other intellectual property that is the subject of an application, certificate, issuance or registration filed with, issued by, or recorded by, any state, government, or other legal authority in any country;

- "Regulations S-K and S-X" means Regulations S-K and S-X promulgated by the SEC under the Securities Act;
- "Relevant Participant" has the meaning set out in Schedule 7 (Part 1);
- "Relevant Period" has the meaning set out in Paragraph 10(a) of Schedule 15 (Part 1);
- "Reporting Accountants" means E&Y or, if the Parties otherwise agree, another firm of accountants of international repute, with offices in the Netherlands and the United States of America;
- "Representative" means any manager, officer, employee, legal, financial or other professional advisor, accountant or other agent, of the Party concerned;
- "Restricted Activity" means any activity that is substantially the same as the Operations at Signing, wherever conducted;
- "Restricted Employee" means any person who immediately prior to the Signing Date is employed by a Group Company or otherwise employed primarily in the Operations by any member of Seller's Group and who:
 - (*d*) has access to trade secrets or other similar confidential information of the Group; or
 - (e) is a Senior Employee;
- "Restricted Period" means, in relation to any action referred to in Clause 12.1.1, (a) where such action takes place in any Member State, a period of 3 (three) years commencing on the Closing Date, and (b) where such action takes place other than in any Member State, a period of 5 (five) years commencing on the Closing Date or such shorter period of time recognised by applicable Law as being binding on Seller, and the other members of Seller's Group;
- "SEC" means the United States Securities and Exchange Commission;
- "Securities Act" means the United States Securities Act of 1933, as amended;
- "Seller" has the meaning set out in the preamble of this Agreement;
- "Seller Plan" has the meaning set out in Schedule 7;
- "Seller's Group" means Seller and its subsidiaries from time to time, excluding after Closing the Group Companies;

61

Confidential

Execution version

- "Seller's Lawyers" means De Brauw Blackstone Westbroek N.V. at Claude Debussylaan 80, 1082 MD Amsterdam, the Netherlands;
- "Seller's Retained Starch Activities" means the following activities of the Seller's Group (excluding the Group Companies and, in relation to the Businesses, the Business Sellers) relating to starch (which for the purposes of this definition includes Specialty Starches) and which, for the avoidance of doubt, are not reflected in the Accounts:
- (a) the research, development, production, marketing, sale and distribution of starch and products that contain starch, in each case for use or application in the Customer Field, conducted by Seller's Surface Chemistry business unit;
- (b) the research, development, production, marketing, sale and distribution of cellulose derivatives and modified cellulose derivatives, in each case which may contain starch or starch derivatives, conducted by Seller's Functional Chemicals business unit;
- (c) the research, development, production, marketing, sale and distribution of coatings, and adhesives, in each case which may contain starch, conducted by Seller's Performance Coatings and Deco business units;
- (d) the research, development, marketing, sale and distribution of redispersible powders which may contain starch, and the research, development, marketing, sale and distribution of starch, in each case, for use or application in dry-mix construction and building materials (including adhesives), air-entraining agents, binders, cement set modifiers, defoamers, hydrophobes, rheology modifiers and shrinkage reducers, conducted by Seller's Functional Chemicals business unit;
- (e) the research, development, production, marketing, sale and distribution of products and systems, in each case which may contain starch, and the research, development, marketing, sale and distribution of starch, in each case, for use or application in the pulp and paper industries, and the production of starch for application in linerboard and paperboard coatings products (including, for use in the wet and size press of the papermaking process), all as conducted by Seller's Pulp and Paper business unit; and
- (f) the sale and distribution of starch by ICI Pakistan Limited;

provided that, for purposes of this definition, anything that consists entirely of starch shall be deemed not to "contain starch."

"Seller's Warranties" means the warranties given by Seller to Purchaser pursuant to Clause 9.1 and Schedule 15 (Part 1), and "Seller's Warranty" means any one of them or the relevant one of them, as the context requires;

"Senior Employee" means the Global Leadership Team and the Regional Leadership Teams of the Group, comprising the individuals occupying the positions listed in Schedule 20 as of the Signing Date or at any time between the Signing Date and the Closing Date;

"Share Purchasers" means Corn Products Netherlands S.a.r.l. of Luxembourg, the purchaser of the Shares, and Corn Products International, Inc., the purchaser of the US Interests, or in each case such other members of Purchaser's Group nominated in accordance with Clause 2.3.1, and

62

Confidential

Execution version

"Share Purchaser" means any one of them or the relevant one of them, as the context requires;

"Share Seller" means Carelaa B.V., a private limited liability company incorporated under the laws of the Netherlands, with corporate seat in Arnhem, the Netherlands, and registered address at Velperweg 76, Arnhem, the Netherlands;

"Shares" means the entire issued share capital of the Company;

"Signing" means the signing of this Agreement by the Parties;

"Signing Date" means the day on which the second Party signing this Agreement has signed it;

"Specialty Starches" means (a) starch from any source the structure of which has been modified through physical, enzymatic or chemical processes and (b) starch derived from specialty hybrids of any plant, which shall mean all hybrids produced by traditional breeding and/or genetic engineering techniques, and that yield starches with distinctive structural features, physical characteristics and/or performance benefits in use, in each case whether in solid or liquid form;

"Subsidiaries" means the companies identified in Schedule 2 below the Company or the US Company, as applicable, and "Subsidiary" means any one of them or the relevant one of them, as the context requires;

"Tax" or "Taxation" has the meaning set out in Schedule 14 (Part 1);

"**Tax Authority**" has the meaning set out in Schedule 14 (Part 1);

"Tax Indemnity" means the indemnities relating to Taxation set out in Schedule 14 (Part 1);

"Tax Return" has the meaning set out in Schedule 14 (Part 1);

"Tax Sharing Arrangement" has the meaning set out in Schedule 14 (Part 1);

"**Tax Warranties**" means Seller's Warranties contained in Paragraph 15 of Schedule 15 (Part 1) and, insofar as they relate to Tax matters, Paragraphs 2, 5, 7, 8, 9, 12 and 17 of Schedule 15 (Part 1), and "**Tax Warranty**" means any one of them or the relevant one of them, as the context requires;

"Third Party Consents" means all consents, licences, approvals, permits, authorisations or waivers required from third parties in respect of the assignment or transfer to Purchaser or the relevant Share Purchaser or Business Purchaser of (a) any of the Shares or other rights and assets transferred, directly or indirectly, under Clause 2.1.1, (b) any of the US Interests or other rights and assets transferred, directly or indirectly, under clause 2.1.1, or (c) any of the Business Liabilities, or otherwise reasonably necessary to permit Purchaser to conduct the Operations after the Closing, and "Third Party Consent" means any one of them or the relevant one of them, as the context requires;

"Third Party Indebtedness" means the aggregate amount (which may be a positive or negative number), at the Effective Time, of all outstanding borrowings, including bank loans, overdrafts, bonds and other loans, and finance leases, together in each case with accrued interest thereon (i) owed by any Group Company to any third party (expressed as a positive number) and (ii) owed

63

Confidential

Execution version

by any third party to any Group Company (expressed as a negative number), as derived from the relevant Net Debt Statement and included in line items under the heading "Third-Party Indebtedness" in Paragraph 3 of Schedule 13 and, for the purposes of this definition, third party shall exclude any member of the Seller's Group;

- "Transaction" means the acquisition of the Group by the Purchaser (and relevant other member(s) of the Purchaser's Group);
- "Transfer Regulations" means EU Acquired Rights Directive 2001 as implemented in the relevant Member States or any equivalent legislation or regulation dealing with the same subject matter in any other jurisdiction, as amended or enacted from time to time;
- "Transferring Employee" shall mean any Automatic Transfer Employee (other than any person who exercises his or her right not to transfer under applicable Law) and any Non-Automatic Transfer Employee who accepts Purchaser's offer of employment;
- "Transitional Services Agreement" means the agreement attached hereto in Agreed Terms as Schedule 17, regarding a 12 (twelve) month transitional arrangement in respect of certain human resources, benefits and IT matters (e.g. server space and certain network services), a six-month transitional arrangement in respect of certain cash management matters and such other transitional arrangements as may be mutually agreed by the parties in accordance with the provisions thereof;
- "UK Business" means that part of the Business of the Group conducted by the relevant Business Seller in the UK;
- "UK Defined Benefit Plans" means the [ICI Pension Fund] and the [ICI Speciality Chemicals Pension Fund];
- "UK Employees" means all those Automatic Transfer Employees working in the UK Business;
- "US Company" means National Starch LLC, a Delaware limited liability company;
- "US Interests" means all of the outstanding membership interests of the US Company;
- "US Seller" means Akzo Nobel Chemicals Inc., a Delaware corporation (Federal ID Number 13-3429437);
- "Unregistered Intellectual Property" means all intellectual property other than Registered Intellectual Property, including: (i) proprietary and confidential inventions (whether or not patentable): (ii) Know-how; (iii) unregistered or common law trademarks and service marks, trade names, trade dress and trade styles, product configurations, and logos and like protections; (iv) copyrights, moral rights (to the extent assignable under applicable Law) and like protections throughout the world, works of authorship and derivative works thereof, whether published or unpublished and whether or not the same also constitutes a trade secret, websites and website content, written materials in any form, marketing and sales materials, and brochures; (v) industrial designs throughout the world; and (vi) databases, data structures and data collections and all rights therein throughout the world;

64

Confidential

Execution version

"USD" means US Dollars;

- "VAT" means, within the European Union, such Tax as may be levied in accordance with (but subject to derogation from) the Directive 2006/112/EC, and outside the European Union, any Tax levied by reference to added value, or sales and/or consumption;
- "Warranties" means the Seller's Warranties or the Purchaser's Warranties, as the context requires, and "Warranty" means any one of them or the relevant one of them, as the context requires;
- "Working Capital" means the aggregate of Working Capital Inventory, plus Working Capital Receivables, less Working Capital Payables, but excluding, for the avoidance of doubt, all amounts in respect of (i) Cash Balances, (ii) Intra-Group Receivables, (iii) Intra-Group Payables, and (iv) Third Party Indebtedness;
- "Working Capital Adjustment" means, in relation to the Company (including all other Group Companies other than the US Company and its Subsidiary), the US Company (including its Subsidiary) and each relevant Business, the amount by which its Working Capital is greater than its Base Working Capital (in which case such amount shall be expressed as a positive figure) or by which its Working Capital is less than its Base Working Capital (in which case such amount shall be expressed as a negative figure), and, in relation to the Group, the sum of all such amounts;
- "Working Capital Inventory" means (i) in relation to each relevant Business, its Business Inventory as at the Effective Time and (ii) in relation to each Group Company, all raw materials, consumables, work in progress, part-processed stocks, finished goods, goods for resale and stock in transit, wherever located, of said Group Company as at the Effective Time, as included in line items under the heading "Working Capital Inventory" in Paragraph 3 of Schedule 13;
- "Working Capital Payables" means, in relation to each Group Company and each Business, the aggregate of all amounts owing, accrued or deferred by, as the case may be, said Group Company or the relevant Business Seller in respect of the said Business, in respect of trading creditors, as at the Effective Time, as included in line items under the heading "Working Capital Payables" in Paragraph 3 of Schedule 13;
- "Working Capital Receivables" means (i) in relation to each relevant Business, its Business Receivables as at the Effective Time and (ii) in relation to each Group Company, the aggregate of all amounts receivable, accrued or prepaid by or owed to said Group Company in respect of trading debtors as at the Effective Time, excluding (a) any item to be treated as part of the Cash Balances of said Group Company, and (b) debts owed to said Group Company which are not included in any Working Capital Statement, as included in line items under the heading "Working Capital Receivables" in Paragraph 3 of Schedule 13; and
- **"Working Capital Statement**" means, in relation to the Company (including all other Group Companies other than the US Company and its Subsidiary), the US Company (including its Subsidiary) and each relevant Business, the statement of Working Capital prepared in accordance with Clause 7.1.

Confidential

Execution version

Schedule 6 Employees

(Clause 2.2.1)

1 Definitions

In this Schedule 6:

"Employment Costs" means:

- (a) the amounts payable or paid to or in respect of the employment of the relevant Employee (including salary, wages, Tax and social security contributions, employer's pension contributions, bonus, insurance premiums, payments or allowances or any other consideration for employment); and
- (b) the costs of providing any non-cash benefits, which the employer is required to provide, by Law or Contract or customarily provides in connection with such employment (including other employee benefit provisions);

"Employment Liabilities" means any and all Losses, excluding Employment Costs, directly arising out of or directly connected with employment or the employment relationship, or the initiation (or failure to initiate) or the termination of employment, or of the employment relationship (including all Losses in connection with any claim, award, judgement or agreement for redundancy pay, or damages or compensation for unfair or wrongful dismissal or breach of contract or discrimination).

"Non-Transferring Employee" has the meaning set out in Paragraph 3(d).

2 Non-Automatic Transfer Employees and Automatic Transfer Employees

- (a) Subject to Closing, in respect of any Non-Automatic Transfer Employees, as soon as reasonably practicable after Signing, but in any event not less than 5 (five) Business Days prior to the Closing Date set in Clause 6.1.1 for Closing, a member of the Purchaser's Group shall make an offer to each Non-Automatic Transfer Employee (other than those under notice of termination of employment by Seller or any Seller Company for any reason) to employ such Non-Automatic Transfer Employee under a new contract of employment commencing, subject to Closing, with effect from the date of Closing. The offer to be made shall be such that:
 - (i) the provisions of the new contract shall be, as to the capacity and place in which the Employee will be employed, and as to the other terms and conditions of his or her employment, as favourable in overall terms as the corresponding provisions of his or her contract of employment (including retirement benefits) as existing immediately prior to the Closing;
 - (ii) it provides that such Non-Automatic Transfer Employee's period of continuous service with Seller's Group shall be counted as continuous service with Purchaser's Group; and
 - (iii) it provides that it shall be a condition for the acceptance of such offer that the Non-Automatic Transfer Employee submits a letter of resignation, with effect

66

Confidential

Execution version

from the date of Closing, as an employee of the relevant member of Seller's Group.

Notwithstanding the foregoing, if Purchaser so requests in writing and if Seller agrees thereto in writing, Purchaser may elect not to make offers of employment to certain Non-Automatic Transfer Employees selected by Purchaser, in which event such Non-Automatic Transfer Employees shall be treated as a Non-Transferring Employee in accordance with Paragraph 3(d) (and Purchaser's request in writing shall be deemed to constitute the notice of Purchaser referred to in Paragraph 3(d)). In addition to the notice and termination provisions of Paragraph 3(d), the relevant members of Seller's Group shall use commercially reasonable endeavours to avoid or mitigate the quantum of Employment Costs and Employment Liabilities incurred in respect thereof.

- (b) Subject to Closing and Paragraph 3(d):
 - (i) In respect of all Automatic Transfer Employees, Purchaser's Group shall, with effect on and from the date of the Closing:

- (A) employ, for at least one (1) year after the Closing on terms as to the capacity and place in which such Automatic Transfer Employee will be employed and as to other contractual terms and conditions of his or her employment which are as favourable in overall terms as the corresponding provisions of his or her contract of employment (including retirement benefits) immediately prior to the date of Closing (except in relation to any terms, conditions or provisions excluded by the Transfer Regulations); and
- (B) count his or her period of continuous service with Seller's Group as continuous service with Purchaser's Group.
- (c) Each Party shall comply, or shall procure that its applicable Affiliates comply, in all respects with the obligations imposed on a transferor and transferee by the applicable Transfer Regulations.
- (d) If, following the sale and transfer of the Group, it is established that the contract of employment or employment (as applicable) of an Automatic Transfer Employee who was identified by Seller as such did not transfer in accordance with the Transfer Regulations (a "Remaining Transfer Employee"), then a member of the Purchaser's Group:
 - (i) in consultation with Seller, shall as soon as reasonably possible but in any event within 5 (five) Business Days of being so requested by Seller, make to each such Remaining Transfer Employee an offer in writing to employ him or her under a new contract of employment to take effect immediately; and
 - (ii) shall procure that the offer to be made will be such that (A) the provisions of the new contract shall be, subject to Paragraph 3(c), for at least one (1) year

67

Confidential

Execution version

after the Closing, as to the capacity and place in which the Remaining Transfer Employee will be employed and as to the other contractual terms and conditions of his or her employment, as favourable in overall terms as the corresponding provisions of such Remaining Transfer Employee's contract of employment (including retirement benefits) as existing immediately prior to the Closing, and (B) it provides that such Remaining Transfer Employee's period of continuous service with Seller's Group shall be counted as continuous service with Purchaser's Group.

(e) Seller shall procure that, as of acceptance of the offer made pursuant to Paragraph 2(d)(i), the relevant member of Seller's Group shall terminate the employment of the Remaining Transfer Employee concerned.

3 Further provisions regarding Employees

- (a) Each Party shall where and to the extent required by Law, contract or custom comply with its obligations to inform and consult and to assist the other Party in informing and consulting with employees, trade unions, works councils or other employee representatives regarding the Transaction, the transfer of Automatic Transfer Employees and/or the offers of employment to be made to Non-Automatic Transfer Employees pursuant to this Part 1, and shall fulfil any obligations to notify any statutory or other authority whatsoever in regard thereto.
- (b) All Employment Costs with respect to the Employees relating to any act or omission of the Seller's Group with respect to the period up to the Effective Time shall be borne by the Seller's Group.
- (c) Purchaser shall procure that, in respect of terms and conditions of employment other than terms and conditions relating to redundancy and retirement benefits, those Employees who were employees of Imperial Chemical Industries PLC (or any its subsidiaries) at the time of the acquisition by Seller of Imperial Chemical Industries PLC (the "ICI employees"), continue to enjoy, with effect from the Closing Date until 2 January 2012, terms and conditions that are as favourable in overall terms as those that apply as at the Effective Time, subject to any changes necessary to facilitate the integration of the combined workforce and harmonisation of their terms and conditions, and save that Purchaser and each member of Purchaser's Group shall be authorised to depart from the foregoing to the extent approved by the relevant employees and/or any relevant trade unions, works councils or other employee representative bodies.
- (d) Subject to Closing, Purchaser shall be responsible for and shall fully indemnify and keep indemnified Seller and, as an irrevocable third party stipulation ('derdenbeding'), the relevant members of Seller's Group from and against any and all Employment Costs incurred on or after the date of Closing, and Employment Liabilities which become due on or after the date of Closing, in respect of any Employee, provided that the foregoing indemnity shall not apply in respect of (i) any Automatic Transfer Employee who rejects the transfer to Purchaser's Group as a

68

Confidential

Execution version

result of the Transaction and as a result thereof remains employed by Seller's Group, or (ii) any Non-Automatic Transfer Employee or Remaining Transfer Employee who does not accept Purchaser's offer of employment made in accordance with Paragraph 2(a) or

Paragraph 2(d) (as applicable) and as a result thereof remains employed by Seller's Group (each a "Non-Transferring Employee"), unless within 20 (twenty) Business Days after the later of Closing and Purchaser giving notice to Seller of such rejection or non-acceptance, Seller or another member of Seller's Group has notified Purchaser that the relevant member of Seller's Group intends to terminate the employment of the Non-Transferring Employee and, within 20 (twenty) Business Days after expiry of the aforesaid 20 (twenty) Business Days, initiates such termination taking into account any applicable notice period under the contract of employment or applicable Law.

- (e) Without detracting from the generality of Paragraph 3(d), subject to Closing, Purchaser shall be responsible for and shall fully indemnify and keep indemnified Seller and, as an irrevocable third party stipulation (`derdenbeding´), the relevant members of Seller's Group from and against any and all Employment Costs and Employment Liabilities arising out of or in connection with:
 - (i) any Employment Liabilities which Seller or any of its Affiliates incurs as a result of Purchaser's or any of its Affiliates' failure to comply with its obligations under Paragraph 3(a) or its obligations under Paragraph 2(c) (save where such Employment Liabilities are attributable to the failure of Seller or its Affiliates to comply with its obligations under the Transfer Regulations or under Paragraph 3(a));
 - (ii) any change in the working conditions of any Automatic Transfer Employee or Non-Automatic Transfer Employee occurring after the date of Closing;
 - (iii) the change of employer occurring by virtue of any applicable Law and/or this Agreement; and
 - (iv) any act or omission of any member of Purchaser's Group in relation to any Automatic Transfer Employee or Non-Automatic Transfer Employee after the date of Closing.
- (f) Subject to Closing, Seller shall be responsible for and shall indemnify and keep indemnified Purchaser and, as an irrevocable third party stipulation ('derdenbeding'), each of Purchaser's Affiliates with respect to any and all Employment Costs and Employment Liabilities which Purchaser or any of its Affiliates incurs arising out of, in connection with or as a result of Seller's or any of its Affiliates' failure to comply with its obligations under Paragraph 3(a) or its obligations under Paragraph 2(c) (save where such Employment Liabilities are attributable to the failure of Purchaser or its Affiliates to comply with its obligations under the Transfer Regulations or under Paragraph 3(a)).
- (g) Seller and Purchaser shall give each other such assistance as either may reasonably require in contesting any claim by any Employee or Undisclosed Employee resulting

69

Confidential

Execution version

from or in connection with this Agreement.

70

Confidential

Execution version

Schedule 7 Benefit Arrangements

Schedule 7 (Part 1) General

1. Definitions for this Schedule

"Actuarial Assumptions" means the assumptions used to measure the DBO or equivalent accrued benefit liability in Seller's 31 December 2009 financial reports except that (i) the discount rate shall be adjusted by the change in the Pension Discount Rate Index from 31 December 2009 to the Closing Date; and (ii) if any relevant Seller Plan has been closed or frozen then the salary increase assumption shall be nil provided that steps have been taken in relation to that relevant Seller Plan which would allow a plan curtailment for US GAAP purposes;

"Closing Membership Data" means the Seller Transferring Plan membership data at Closing Date. Membership data shall be rolled forward from 1 January 2010 or the latest date of available data;

"DBO" means the Defined Benefit Obligation as defined by International Financial Reporting Standard IAS19;

"ERISA" means the United States Employee Retirement Income Security Act of 1974, as amended;

"Final DBO Adjustment Amount" means the amount equal to half of the sum of the amount of the Final Excess DBO multiplied by 0.66;

"Non-US Plan" means each Seller Plan which is not a US Plan;

"Past Service Liability" means the DBO for all Seller Transferring Plans measured at Closing Date using the Actuarial Assumptions and the Closing Membership Data;

"**Pension Discount Rate Index**" means the Citigroup pension liability discount rate for US Seller Plans and the relevant iBoxx AA 10+ year bond yield index for all other Seller Plans or an equivalent published discount rate index if agreed by Seller and Buyer;

"Relevant Participants" means (i) all those individuals who immediately prior to Closing are Transferring Employees, and (ii) in respect of any Seller Transferring Plan all beneficiaries, pensioners, deferred pensioners and dependents (including, if applicable, alternate payees and survivors) to the extent such persons have an interest in or the right to receive any benefit under any Seller Transferring Plan;

"Retirement Benefits" means all retirement benefits, survivors' and termination benefits including post retirement healthcare or other welfare benefits that are payable in the form of

71

Confidential

Execution version

lump sums, periodic payments or annuities;

"Seller Plan" means all plans or schemes (whether funded or unfunded) or insurance contracts or other arrangements in effect at Closing under which Seller (or another member of Seller's Group) offer Retirement Benefits to Relevant Participants;

"Seller Retained Plans" means all Seller Plans that are not Seller Transferring Plans, as listed in Part 2 of this Schedule 7. Notwithstanding the foregoing, "Seller Retained Plans" shall include all Seller Plans that are not Seller Transferring Plans, even if such plans do not appear in Part 2 of this Schedule 7:

"Seller Transferring Plan" means any plan, scheme, insurance contract or other arrangement (whether funded or unfunded) under which Seller (or any member of the Seller's Group) offers, prior to or on the Closing Date, Retirement Benefits solely to Relevant Participants that is listed in Part 3 of this Schedule 7;

"Target DBO" means USD 122,000,000 (one hundred twenty two million US dollars);

"US Plans" means all Seller Plans that are maintained in or established under the laws of the United States of America, including each "employee pension benefit plan" (as such term is defined in Section 3(2) of ERISA) and each "employee welfare benefit plan" (as such term is defined in Section 3(1) of ERISA) and "US Plan" means any one of them or the relevant one of them, as the context requires.

2. General

- 2.1 Purchaser, or a relevant other member of Purchaser's Group shall be responsible for the continued operation and shall accept the transfer of all Retirement Benefits obligations of the Seller Transferring Plans with effect from the Closing Date. Seller and its Affiliates and Purchaser, or a relevant other member of Purchaser's Group, shall jointly undertake all action necessary to effectuate the transfer, within 60 days following the Closing Date, of all trusts, insurance Contracts and other assets held by Seller relating to the Seller Transferring Plans to Purchaser or its applicable Affiliate effective as of the Closing.
- 2.2 All Relevant Participants who are participants in a Seller Retained Plan shall cease to be an active participant in such plan and shall cease to accrue Retirements Benefits under such plan with effect on and from the Closing Date. Seller, or the relevant Affiliate of Seller, shall retain all liabilities for the Seller Retained Plans, and neither Purchaser nor any of its Affiliates shall have any liability or obligation with respect to such Seller Retained Plans.
- 2.3 Subject to other provisions within this Schedule 7 and this Agreement, for services rendered on or after the Closing Date, Purchaser's Group shall offer Retirement Benefits under pension plans which are no less favourable in overall terms than either (i) the Retirement Benefits provided to similarly situated employees of Purchaser and its Affiliates

72

Confidential

Execution version

- or (ii) the Retirement Benefits provided to Participants under either the Seller Retained Plans or the Seller Transferring Plans as applicable, and effective as of Closing. For the avoidance of doubt the replacement Retirement Benefits need not be defined benefits.
- 2.4 Seller and Purchaser shall make such arrangements as are required by Law and otherwise use reasonable endeavours to make such arrangements as are permitted by Law so that the accrued Retirement Benefits of Relevant Participants and the related assets of any Seller Retained Plan determined with reference to such Retirement Benefits are transferred from the relevant Seller Retained Plan to Purchaser Group Plans and the applicable Purchaser Group Plans shall be responsible for paying out such accrued benefits.

3. Non-US Plans

- 3.1 Seller shall retain responsibility and liability for and shall indemnify and keep indemnified Purchaser and, as an irrevocable third party stipulation ('derdenbeding'), each of Purchaser's Affiliates with respect to any and all Liabilities and Losses in relation to:
 - (a) any Seller Retained Plan, including for the avoidance of doubt, any Liabilities and Losses in relation to any Seller Plan contributed to by or on behalf of any Automatic Transfer Employee (including any Automatic Transfer Employee who does not transfer in accordance with applicable Law); and
 - (b) any contribution notice or financial support direction arising under the Pensions Act 2004 or any payment, debt or other Liability arising under Section 75 or Section 75A of the Pensions Act 1995 in respect of the UK Defined Benefit Plans and any equivalent debt, penalty, tax, fine, demand or other Liability arising in any other relevant jurisdiction in respect of a Seller Retained Plan.
- 3.2 From the Closing Date, Purchaser and its Affiliates shall offer Retirement Benefits to the Automatic Transfer Employees in accordance with the Transfer Regulations and applicable Law.

4. United States of America

- 4.1 Purchaser's Group shall comply with all required legal provisions related to the US Plans, including collective bargaining agreements and all relevant US regulations effective upon the Closing Date. Purchaser shall procure that the relevant members of Purchaser's Group shall assume full responsibility for the liability of all US Plans which are Seller Transferring Plans including ongoing funding (including any amounts first due after the Closing relating to obligations that arose before the Closing), administrative and legal requirements.
- 4.2 The Parties acknowledge that actions required to be taken pursuant to this Paragraph 4 may be subject to fiduciary duties or standards of conduct under ERISA or other Law, and no Party shall be deemed to be in violation of this Agreement if it fails to comply with any provisions hereof based upon its good faith determination that to do so would violate such a fiduciary duty or standard.
- 4.3 Except to the extent otherwise provided in this Agreement, (a) neither Purchaser nor any of its Affiliates shall assume any obligations arising under any US Plans, and (b) Seller and its Affiliates shall retain liability for, and shall indemnify Purchaser and its Affiliates with respect to, all liabilities that arise under the Seller Retained Plans forming part of the US

73

Confidential

Execution version

Plans.

4.4 Regardless of anything else contained herein, the Parties do not intend for this Agreement to amend any employee benefit plans or arrangements or create any rights or obligations except between the Parties. No employee, or beneficiary or dependent thereof, or any other person not a party to this Agreement, shall be entitled to assert any claim hereunder.

5. Excess DBO

- 5.1 A purchase price adjustment is to be calculated for all Seller Transferring Plans listed in Part 3 of this Schedule 7 and the accrued Retirement Benefits of Relevant Participants and the related assets of any Seller Retained Plan which are transferred under paragraph 2.4 of this schedule within 60 days of the Closing Date.
- 5.2 As soon as practicable following the Closing Date, Seller shall provide Purchaser with all data reasonably requested by Purchaser to enable Purchaser's actuaries to calculate and agree with Seller's actuaries the Past Service Liabilities. In any event, no later than 60 (sixty) days after the receipt of data (or such other date mutually agreed to in writing by Purchaser and Seller), Purchaser and Seller shall cause their respective actuaries to jointly determine the "Excess DBO" as follows:
 - (a) the US Dollar value of Past Service Liabilities under the Seller Transferring Plans less
 - (b) the US Dollar market value of assets at Closing Date associated with such Seller Transferring Plans, if any, (such amount, which shall be determined without regard to any action taken by Purchaser or any of its Affiliates on or after the Closing Date, including any employee terminations or plan terminations and shall include for the avoidance of doubt the anticipated amounts of the Canadian Pension Transfer described in Clause 7 if such transfer has not yet been made) less
 - (c) the Target DBO.
- 5.3 US Dollar values shall be converted from other currencies using the mid market currency exchange rates on the Closing Date as published in the London edition of the Financial Times.
- 5.4 If the Excess DBO cannot be agreed upon between Seller's actuaries and Purchaser's actuaries within such 60 (sixty) day period (or such later date as agreed to by Purchaser and Seller), then Purchaser and Seller may, within 5 (five) days after the expiration of such 60 (sixty) day period (or such mutually agreed later date), agree on a third-party independent actuary (or if not agreed, such actuary shall be nominated by the President of the Institute of Actuaries, for the time being), who shall determine the Excess DBO, whose expenses shall be shared equally by Purchaser and Seller, and whose determination shall be final, conclusive and binding on all parties.
- 5.5 The Excess DBO as finally determined by Seller's and Purchaser's actuaries or such independent actuary, as applicable, shall be referred to as the "Final Excess DBO."

Confidential

Execution version

2009 financial reports.

- 5.7 If the Final Excess DBO exceeds zero, Seller shall pay Purchaser the Final DBO Adjustment Amount.
- 5.8 Payment under this Clause 5 shall be made, in cash, no later than 14 Business Days following the final determination of the Final Excess DBO.

6. Canadian Plan

If the transfer of assets and liabilities from the Pension Plan for Salaried Employees of Nacan Products Limited and Associated and Subsidiary Companies to the Pension Plan for Salaried Employees of National Starch ULC and its Associates and Subsidiary Companies (the "Canadian Pension Transfer") which has an effective date as of April 1, 2008 (the "Canadian Pension Transfer Effective Date") has not occurred before Closing then:

- (i) if the Ontario Superintendent of Financial Services approves the Canadian Pension Transfer [in an amount that is, as of the Canadian Pension Transfer Effective Date, less than the amount calculated as of the Canadian Pension Transfer Effective Date and as specified in the asset transfer report filed with the Ontario Superintendent of Financial Services in 2009 requesting approval for the Canadian Pension Transfer. Seller shall pay to Purchaser an amount equal to the difference between such amounts within 14 Business Days following the date on which the Ontario Superintendent of Financial Services approves the Canadian Pension Transfer; and
- (ii) if the Ontario Superintendent of Financial Services approves the Canadian Pension Transfer in an amount that is, as of the Canadian Pension Transfer Effective Date, more than the amount calculated as of the Canadian Pension Transfer Effective Date and as specified in the asset transfer report filed with the Ontario Superintendent of Financial Services in 2009 requesting approval for the Canadian Pension Transfer, Purchaser shall pay to Seller an amount equal to the difference between such amounts within 14 Business Days following the date on which the Ontario Superintendent of Financial Services approves the Canadian Pension Transfer,

provided that no payment will be made under this clause if the amount of such payment is Can \$1 million or less.

To the extent that any member of Seller's Group has any co-management or custodial rights with respect to the Nacan Plan (including with respect to the investment of the assets thereof), Seller shall grant Purchaser such rights to the full extent permitted by Law during the period from and after the Closing through and including the date on which the Ontario Superintendent consents to a transfer of assets from the Nacan Plan to the National Starch Plan.

7. US Plan Funding

7.1 If the Closing Date occurs on or after September 1, 2010, Seller shall, prior to September 15, 2010, make any and all cash contributions to the National Starch LLC Pension Plan for the 2009 plan year as necessary so that the "adjusted funding target attainment percentage" as defined under section 436(j) of the Code ("AFTAP"), for such 2009 plan year shall be at least 80%, as certified pursuant to regulations promulgated under section

75

Confidential

Execution version

436 of the Code (the "AFTAP Certification"). If the Closing Date occurs prior to September 1, 2010, Purchaser shall be responsible for any and all cash contributions necessary to bring the AFTAP for the National Starch LLC Pension Plan for the 2009 plan year to at least 80%, as reported on the AFTAP Certification. In such case, Seller agrees to provide to Purchaser on the Closing Date actuarial data, including target liability and actuarial value of assets, to determine values to be reported on such AFTAP Certification.

- 7.2 If a contribution under clause 8.1 is required and made by Seller, at the Closing Purchaser shall pay to Seller an amount equal to the product of the amount of such contribution multiplied by 0.65.
- 7.3 Seller shall be responsible to effect minimum required contributions for the National Starch LLC Pension Plan for the 2009 plan year, either by making cash contributions on or prior to September 15, 2010, or by making a valid election to apply the funding carryover balance, as permitted under sections 430 and 436 of the Code. Seller shall be responsible to effect all quarterly required contributions under sections 412 and 430 of the Code for the National Starch LLC Pension Plan for the 2010 plan year prior to the Closing Date.

Confidential

Execution version

Schedule 8 Closing

Schedule 8 (Part 1) Closing obligations

(Clause 6)

4 General Obligations

(a) Seller's obligations

At Closing, Seller shall deliver or make available to Purchaser the following:

- (i) evidence reasonably satisfactory to Purchaser that Seller is authorised to sign this Agreement and that the Share Seller, US Seller or the relevant Business Seller, as the case may be, in respect of each Local Transfer Document, is authorised to sign such Local Transfer Document;
- (ii) a certificate signed by a duly authorised officer of Seller, dated the Closing Date, to the effect that, to Seller's knowledge, no change, event or circumstance has occurred between the Signing Date and the Closing Date that has a Group Material Adverse Effect; and
- (iii) either (1) a certificate signed by a duly authorised officer of Seller, dated the Closing Date, to the effect that, to Seller's knowledge, there is no breach of Seller's Warranties disregarding all qualifications contained therein relating to "materiality" or "Group Material Adverse Effect," as of the Closing Date as though made on such date (except to the extent any Seller's Warranty speaks as of a specific date, in which case there is no breach of such Seller's Warranty as of such specified date), except where the breach of any of Seller's Warranties does not, individually or in the aggregate, have a Group Material Adverse Effect or (2) a certificate signed by a duly authorised officer of Seller, dated the Closing Date, to Seller's knowledge, there is no breach of Seller's Warranties disregarding all qualifications contained therein relating to "materiality" or "Group Material Adverse Effect," as of the Closing Date as though made on such date (except to the extent any Seller's Warranty speaks as of a specific date (in which case there is no breach of such Seller's Warranty as of such specified date), except where the breach of any of Seller's Warranties does not, individually or in the aggregate, have a Group Material Adverse Effect except as set out in such certificate and an undertaking from Seller to indemnify and hold harmless Purchaser and, as an irrevocable third party stipulation ('derdenbeding'), each Group Company, each Business Purchaser and each Share Purchaser against any Losses that they may suffer arising out of the breaches set out in such certificate, subject to the limitations set out in Clauses 10.1 through 10.3 and Clauses 10.5 through 10.10, but without regard to the limitations set out in Clause 10.4.

(b) Purchaser's obligations

At Closing, Purchaser shall deliver or make available to Seller:

77

Confidential

Execution version

- (i) evidence reasonably satisfactory to Seller that Purchaser is authorised to sign this Agreement and that Purchaser and / or the relevant Share Purchaser, and / or the relevant Business Purchaser, as the case may be, in respect of each Local Transfer Document, is authorised to sign such Local Transfer Document; and
- (ii) evidence of the due satisfaction of the Conditions Precedent set out in Clause 4.1.1 and 4.1.2.

5 Transfer of the Shares, Membership Interests and Businesses

(a) General transfer obligations

At Closing, Seller shall procure that the Share Seller, US Seller and Business Sellers, and Purchaser shall procure that the Share Purchasers and Business Purchasers, shall execute and/or deliver and/or make available the Local Transfer Documents and take such steps as are required to transfer the Shares, the US Interests and Businesses, provided that:

- (i) the Local Transfer Agreement in respect of the UK Business shall be executed after performance of the actions set out in Paragraph 2(b)(i); and
- (ii) in respect of the Local Transfer Document relating to India, in the event that, notwithstanding Seller's reasonable efforts to procure the approval by the non-executive directors of the relevant Business Seller for the execution and performance of the Local Transfer Document, such approval has not been obtained, this Agreement shall cease to apply in respect of the Business in India and the (Estimated) Purchase Price in respect thereof shall be excluded from the aggregate (Estimated) Purchase Price and, to the extent already received by Seller, forthwith repaid to Purchaser.

(b) Specific transfer obligations

For the purposes of compliance with Paragraph 5(a), Seller shall procure that the Share Seller, US Seller and the Business Sellers, and Purchaser shall procure that the Share Purchasers and Business Purchasers, shall do the following, in relation to the Company, the US Company and the Businesses incorporated or located in the jurisdictions listed below:

(i) United States

The US Seller shall take the actions required by the US Company's Limited Liability Company Agreement to admit the applicable Share Purchaser as the sole member of the US Company and deliver the certificate(s) representing the US Interests to the applicable Share Purchaser, together with an appropriate instrument of transfer duly executed in blank by or on behalf of the US Seller.

(ii) Netherlands

After completion of the actions described in Paragraphs 2(a)(i) and 2(b)(i), Share Seller shall transfer the Shares to the applicable Share Purchaser, such Share

78

Confidential

Execution version

Purchaser shall accept the transfer, and such Share Seller shall procure that the Company acknowledges the transfer, the foregoing to be effected by execution by the Share Seller, the Share Purchaser and the Company, before one of the civil law notaries of Seller's Lawyers, of a notarial deed of transfer in Agreed Terms.

6 Further obligations in addition to transfer

Seller's Group shall deliver or make available to the Share Purchasers or Business Purchasers, as the case may be, the following:

- (a) the written resignations in Agreed Terms (and legalised by a notary where required) of each of those directors and other officers of a Group Company who is not an Employee from such office;
- (b) evidence that all persons referred to in Paragraph 6(a) above holding share(s) in any Group Company under a nominee-type arrangement or any arrangement having a similar effect have transferred such share(s) to such other persons as Purchaser may specify, to take effect on Closing;
- (c) in each case where the said information is not in the possession of the relevant Group Company, the corporate books and records, including the shareholders' register and share certificates in respect of the Subsidiaries, and all other books and records required to be kept by Law, all to the extent under the control of any Group Company;
- (d) in each case where the tangible Business Books and Records are not at the Properties at Closing, such Business Books and Records in the possession or under the control of Seller (or relevant other member of Seller's Group) (save for books, records and other information which a member of Seller's Group is required by Law to retain or in respect of which a member of Seller's Group obtains a direction to retain, in which case Seller's Group shall deliver or make available to the Share Purchasers or Business Purchasers, as the case may be, copies thereof);
- (e) if applicable, evidence as to the withdrawal of signing authority of the persons referred to in Paragraph 6(a) in respect of any of the bank accounts of any of the Group Companies;
- (f) evidence as to the acceptance by shareholders or the directors of each of the relevant Group Companies of the resignations referred to in Paragraph 3(a) and of the appointment of such persons to take effect on Closing (within the maximum number permitted by the Constitutional Documents of the Group Company concerned) as Purchaser may nominate as directors and (if relevant) secretary, Purchaser to notify Seller of such nomination by no later than 20 (twenty) Business Days after the Signing Date;
- (g) a statement in accordance with U.S. Treasury Regulation §§ 1.1445-2(b)(2) certifying the non-foreign status of US Seller.

79

Confidential

Execution version

7 IP Licence agreements

The relevant members of Seller's Group shall and the relevant members of Purchaser's Group shall enter into separate licence agreements pursuant to the IP Licence.

8 Transitional Services Agreement

Seller and Purchaser shall enter into the Transitional Services Agreement.

80

Confidential

Execution version

Schedule 8 (Part 2) Transfer of Contracts

(Clause 2.1.1(g))

1 Obligation to obtain Third Party Consents

- (a) In relation to any Contract which is not assignable without a Third Party Consent, this Agreement shall not be construed as an assignment or an attempted assignment and Seller and Purchaser shall procure that the relevant Business Seller and the relevant Business Purchaser each use reasonable efforts both before and after Closing to obtain all necessary Third Party Consents as soon as possible and shall keep each other informed of progress in obtaining such Third Party Consents. Seller shall procure that the relevant Business Seller shall deliver to the relevant Business Purchaser, on Closing or, if later, as soon as possible after receipt, any Third Party Consent and the necessary transfer instrument duly executed by the appropriate parties.
- (b) In connection with the obtaining of any Third Party Consent referred to in Paragraph 1(a), Purchaser shall procure that the relevant Business Purchaser supplies to the relevant Business Seller such information and references regarding the financial position of the relevant Business Purchaser as may be reasonably requested by the relevant Business Seller or any relevant third party and shall enter into such undertakings or procure such guarantees in favour of any relevant third party as are required by the terms of any such Group Contract or may be reasonably requested in respect of any liabilities or obligations to which the relevant Business Purchaser will become subject or which the relevant Business Purchaser will incur on assignment.

2 Obligations until Third Party Consents are obtained / where Third Party Consents are refused

In respect of any Contract, from Closing until the relevant Third Party Consent has been obtained as contemplated by Paragraph 1(a) or where the Third Party Consent has been refused:

- (a) Seller and Purchaser shall procure that the relevant Business Seller and the relevant Business Purchaser shall make such other arrangements between themselves to provide to the relevant Business Purchaser the benefits of the Contract, including the enforcement, at the cost and for the account of the relevant Business Purchaser, of all rights of the relevant Business Seller against any other party thereto;
- (b) to the extent that the relevant Business Purchaser is lawfully able to do so, Purchaser shall procure that the relevant Business Purchaser shall perform the relevant Business Seller's obligations under the Contract in the name of the Business Seller as agent or sub-contractor and shall indemnify the relevant

81

Confidential

Execution version

Business Seller in respect thereof. To the extent that the relevant Business Purchaser is not lawfully able to do so, Seller shall procure that the relevant Business Seller, (subject to being indemnified by the relevant Business Purchaser for any Losses the relevant Business Seller may incur in connection therewith) at the relevant Business Seller's cost do all such things as the relevant Business Purchaser may reasonably require to enable due performance of the Contract.

82

Confidential

Execution version

Schedule 12 VAT

(Clause 3.4)

3 VAT: Records

- (a) Seller shall have the right to obtain (or to procure that the relevant member of Seller's Group obtains) a direction from the relevant Tax Authority for the retention and preservation by it of any VAT records relating to its period of ownership of the relevant part of the Group and where such directions are obtained Seller's Group shall preserve any such records in such a manner and for such period as may be required by the applicable Law governing VAT and will allow Purchaser and its Representatives, upon Purchaser giving reasonable notice to Seller, reasonable access to, and permit Purchaser and its Representatives to take copies of, such records where reasonably required by Purchaser or its Representatives for Taxation purposes.
- (b) If no such direction is obtained and any documents are required by Law to be preserved by Purchaser (or relevant other member of Purchaser's Group), Seller shall, as soon as reasonably practicable, deliver or procure delivery of such documents to Purchaser (or relevant other member of Purchaser's Group).

4 VAT: Going Concern - EU Member States

- (a) Subject to Paragraph 4, Seller's Group shall and Purchaser's Group shall use all reasonable efforts (including, for the avoidance of doubt, but subject to Paragraph 2(c), the making of an election and/or application in respect of VAT to any relevant Tax Authority) to secure that the sale of the Businesses so far as carried on in the EU is treated as neither a supply of goods nor a supply of services for the purposes of the Law governing VAT in the relevant member state. Purchaser shall procure that each Business Purchaser is at the Closing a taxable person for VAT purposes in the relevant member state and agrees that the relevant Business Purchaser intends to use the assets acquired in carrying on the same kind of business, whether or not as part of its existing business, as the relevant Business Seller.
- (b) Seller shall have the right to obtain (or to procure that the relevant other member of Seller's Group obtains) a ruling from the relevant Tax Authority as to whether the sale of a Business so far as carried on in the relevant member state should be treated as neither a supply of goods nor a supply of services for the purposes of the Law governing VAT in that member state and to charge (or not to charge) VAT to the relevant Business Purchaser(s) in accordance with that ruling. Seller (and relevant other member(s) of Seller's Group) shall not be obliged to challenge that ruling. If Purchaser (or relevant other member(s) of Purchaser's Group) wishes to challenge any ruling, it may do so at its own cost, but any such challenge shall not affect the date on which VAT must be paid to the relevant Business Seller under Paragraph 5.

83

Confidential

Execution version

- (c) In the event that in order to secure that the sale of a Business so far as carried on in the EU is treated as neither a supply of goods nor a supply of services for the purposes of the Law governing VAT in the relevant member state, it is necessary for the relevant Business Purchaser to make an election for the purposes of VAT in respect of such Business or any asset comprised in the sale of such Business then the relevant Business Purchaser shall only be obliged to make any such election:
 - (i) to the extent that it is entitled so to do; and
 - (ii) provided that Seller (or relevant other member of Seller's Group) notifies Purchaser (or relevant other member of Purchaser's Group) in writing of the need for such an election no later than 10 (ten) Business Days prior to the last date upon which such election can be made under the applicable Law governing VAT in the relevant member state.

5 VAT: Going Concern - non-EU Jurisdictions

- (a) Subject to Paragraph 4, to the extent that any state outside the European Union provides for relief or exemption from VAT on the transfer of a business in a company or treats such a transaction as being non-taxable for VAT purposes, Seller's Group shall and Purchaser's Group shall use all reasonable efforts (including, for the avoidance of doubt, but subject to Paragraph 3(c), the making of an election and/or application in respect of VAT to any relevant Tax Authority) to secure such treatment as regards the sale of a Business so far as carried on in the relevant state under this Agreement. To the extent that the intention as to the use of the assets acquired as set out in this sentence is a condition as to the availability of a relief or exemption from VAT in respect of the sale of a business or for such transaction to be treated as being non-taxable for VAT purposes as described above in this Paragraph 3(a), Purchaser agrees that the relevant Business Purchaser intends to use the assets acquired in carrying on the same kind of business, whether or not as part of its existing business, as the relevant Business Seller.
- (b) Seller shall have the right to obtain (or to procure that the relevant other member of Seller's Group obtains) a ruling from the relevant Tax Authority as to whether the sale of a Business so far as carried on in the relevant state is eligible for a relief or exemption or is otherwise non-taxable for the purposes of the Law governing VAT in that state and to charge (or not to charge) VAT to the relevant Business Purchaser in accordance with that ruling. Seller (and relevant other member(s) of Seller's Group) shall not be obliged to challenge that ruling. If Purchaser (or relevant other member(s) of Purchaser's Group) wishes to challenge any ruling, it may do so at its own cost, but any such challenge shall not affect the date on which VAT must be paid to the relevant Business Seller under Paragraph 5.
- (c) In the event that in order to secure that the sale of a Business so far as carried on in a non-EU member state is treated as being non-taxable for VAT purposes, it is necessary for the relevant Business Purchaser in respect of such Business or any

Execution version

asset comprised in the sale of such Business to make an election for the purposes of VAT then the relevant Business Purchaser shall only be obliged to make any such election:

- (i) to the extent that it is entitled so to do; and
- (ii) provided that Seller (or relevant other member of Seller's Group) notifies Purchaser (or relevant other member of Purchaser's Group) in writing of the need for such an election no later than 10 (ten) Business Days prior to the last date upon which such election can be made under the applicable Law governing VAT in the relevant non-EU member state.

6 VAT: election

If:

- (a) a sale, transfer or supply of one or more assets in accordance with Paragraphs 4(a) or 5(a) would result in a VAT disadvantage for Seller (or the relevant other member of Seller's Group), including a repayment or recharge of input VAT credited or deducted; and
- (b) Seller (or relevant other member of Seller's Group) can make an election as to whether or not the sale, transfer or supply of such asset is subject to VAT; and
- (c) Purchaser (or relevant other member of Purchaser's Group) would be entitled to recover (by way of credit or repayment) the full amount of VAT charged to Purchaser (or relevant other member of Purchaser's Group) in respect of such sale, transfer or supply; and
- (d) Purchaser (or relevant other member of Purchaser's Group) would not otherwise suffer any Tax disadvantage as a result of such VAT charge and/or the making of any elections as described in this Paragraph 4,

then:

- (e) Seller (or relevant other member of Seller's Group) shall subject to prior written approval of Purchaser, such approval not to be unreasonably withheld (provided that it shall be reasonable for Purchaser so to withhold its approval unless Purchaser, acting reasonably, considers both that (x) both of the conditions set out in paragraph 4(c) and (d) above will be satisfied and (y) it (or other relevant member of Purchaser's Group) will be entitled to and can make the election described at paragraph 4(f) below prior to the expiry of the time period for the making of such election), be entitled to elect for the sale, transfer or supply of the asset to be subject to VAT; and
- (f) subject to the making of such election as is described in Paragraph 4(e) above by Seller (or relevant other member of Seller's Group), Purchaser (or relevant other member of Purchaser's Group) shall, to the extent that it is entitled to do so, make any election that may be necessary in order to give effect to the election made by Seller (or relevant other member of Seller's Group), provided that Seller (or relevant

85

Confidential

Execution version

other member of Seller's Group) notifies Purchaser (or relevant other member of Purchaser's Group) in writing no later than 10 (ten) Business Days prior to the last date upon which such election can be made under the applicable Law governing VAT.

7 VAT: time, manner and currency of payment

- (a) In the event that it is determined by a relevant Tax Authority (whether or not in response to an application for a ruling, by assessment or otherwise) or by a court of competent jurisdiction, that VAT is properly chargeable in respect of the sale of a Business or the Shares under this Agreement, Purchaser shall procure that any amounts of VAT which Purchaser (or relevant other member of Purchaser's Group) is obliged to pay to Seller (or relevant other member of Seller's Group) in terms of this Schedule 12, shall be paid to Seller (or relevant other member(s) of Seller's Group) or as Seller may direct in accordance with Paragraph 5(b) or 5(c) (as the case may be) below. Such amounts shall be paid in the currency in which the VAT in question must be accounted for to the relevant Tax Authority.
- (b) Subject to Paragraph 5(c) below, any VAT payable in any jurisdiction in respect of the transfer of any of the Businesses or the Shares shall be paid by Purchaser (or relevant other member of Purchaser's Group) at the later of (i) Closing or (ii) ten (10) Business Days after the receipt of notification from Seller that VAT is so properly chargeable pursuant to a determination as described in Paragraph 5(a) above in respect of the relevant sale provided that Seller has first delivered to Purchaser (x) a valid VAT invoice or equivalent document which is necessary to enable Purchaser (or relevant other member of Purchaser's Group) to recover or obtain credit for such VAT, and (y) a copy of the relevant determination as described in Paragraph 5(a) above, in each case in respect of the relevant sale.
 - In the event that Seller (or relevant other member of Seller's Group) had previously delivered to Purchaser (or relevant other member of Purchaser's Group) an invoice without VAT in relation to the transfer of the Shares or relevant Business, Seller shall, or shall procure that the relevant other member of Seller's Group shall, issue a credit invoice to Purchaser (or relevant other member of Purchaser's Group).
- (c) Neither Purchaser nor relevant other member of Purchaser's Group shall be obliged to pay any VAT arising pursuant to Paragraph 4 above unless and until it has recovered an amount equal to such VAT (whether by way of repayment, credit or otherwise) from the

relevant Tax Authority.

In the event that Purchaser (or relevant other member of Purchaser's Group) has paid an amount of VAT pursuant to this Paragraph 5 to Seller (or relevant other member of Seller's Group) and it is subsequently determined by the relevant Tax Authority or court of competent jurisdiction that all or part of such VAT was not properly chargeable then, if so requested by Purchaser, Seller shall, or shall procure that the relevant other member of Seller's Group shall, use all reasonable

86

Confidential

Execution version

efforts to recover such VAT (by repayment, credit or otherwise) together with interest thereon and shall account to Purchaser, or relevant other member of Purchaser's Group as directed by Purchaser for an amount equal to the amount of VAT so recovered and interest thereon. Seller shall or shall procure that the relevant other member of Seller's Group shall deliver to Purchaser (or relevant other member of Purchaser's Group) a credit invoice.

87

Confidential

Execution version

Schedule 13 (Part 4) Reporting Accountants

(Clause 1.1)

1 Engagement

The Reporting Accountants shall be engaged jointly by Seller and Purchaser on terms to be agreed between Seller and Purchaser.

2 Procedure for determination

Except to the extent that Seller and Purchaser agree otherwise, the Reporting Accountants shall determine their own procedure, but:

- (a) apart from procedural matters and as expressly otherwise set out in this Agreement, shall determine only:
 - (i) whether, in respect of each Disputed Item, any of the reasons for an alteration to a draft Working Capital Statement or a draft Net Debt Statement, as the case may be, set out in the Notice of Disagreement or in Seller's notice pursuant to Clause 7.1.3(c)(i), is correct in whole or in part; and
 - (ii) if so, what alterations should be made to the relevant draft Working Capital Statement or draft Net Debt Statement, as the case may be, in order to correct the relevant Disputed Item;
- (b) shall apply the provisions of Schedule 13 (Part 3);
- (c) shall make their determination pursuant to Clause 7.1.3 as soon as is reasonably practicable;
- (d) the procedure of the Reporting Accountants shall:
 - (i) give Seller and Purchaser a reasonable opportunity to make written representations to them;
 - (ii) require that Seller supply Purchaser, and vice versa, with a copy of any written representations at the same time as they are made to the Reporting Accountants;
 - (iii) permit each Party and its Representatives to be present while oral submissions are being made to the Reporting Accountants; and
- (e) for the avoidance of doubt, the Reporting Accountants shall not be entitled to determine the scope of their own jurisdiction.

3 Determination

The determination of the Reporting Accountants pursuant to Clause 7.1.3 shall:

- (a) be made in writing and made available for collection by Seller and Purchaser at the offices of the Reporting Accountants at such time as they shall determine; and
- (b) unless otherwise agreed by Seller and Purchaser, include reasons for each relevant determination.

Confidential

Execution version

4 Binding advice

The Reporting Accountants shall act as experts and not as arbitrators and their determination of any matter falling within their jurisdiction shall be final and binding on Seller (and the relevant member(s) of Seller's Group) and Purchaser (and the relevant member of Purchaser's Group), save in the event of manifest error (when the relevant part of their determination shall be void and the matter shall be remitted to the Reporting Accountants for correction). In particular, without limitation, their determination shall be deemed to be incorporated into the relevant Working Capital Statement or Net Debt Statement, as the case may be. In resolving any matter specified in a Notice of Disagreement, the Reporting Accountants shall not assign a value to any item greater than the greatest value for such item claimed by either Party or less than the smallest value for such item claimed by either Party.

5 Expenses

The fees and expenses (including VAT) of the Reporting Accountants shall be borne proportionately by Seller and Purchaser, determined on the basis of the aggregate amount of the adjustments requested by Purchaser in respect of the Disputed Items, the aggregate amount of the adjustments proposed or requested by Seller (if any) and the aggregate amount of the adjustments as determined by the Reporting Accountants.

6 Co-operation

Seller and Purchaser shall co-operate with the Reporting Accountants and comply with their reasonable requests made in connection with the carrying out of their duties under this Agreement. In particular, without limitation, Purchaser shall keep up-to-date and, subject to reasonable notice, make available to Seller's Representatives, Seller's accountants and the Reporting Accountants all books and records relating to the Group during normal office hours during the period from the appointment of the Reporting Accountants to the making of the relevant determination.

7 Confidentiality

Each Party shall, and shall procure that its accountants and the other advisors and the Reporting Accountants shall, keep all information and documents provided to them pursuant to this Part 4 confidential and shall not use the same for any purpose, except for disclosure or use in connection with the preparation of the draft Working Capital Statements and the draft Net Debt Statements, the proceedings of the Reporting Accountants or another matter arising out of this Agreement, or in defending any claim or argument or alleged claim or argument relating to this Agreement or its subject matter.

89

Confidential

Execution version

Schedule 14 Tax and Intellectual Property

Schedule 14 (Part 1) Tax Indemnity

8 Definitions

For purposes of this Tax Indemnity and Paragraph 15 of Schedule 15 (Part 1) (Tax):

"Canadian Butterfly Transaction" means the transfer of the National Starch business by Nacan Products Ltd. to National Starch ULC pursuant to a Canadian "butterfly" tax-free reorganization and the transfer of the shares of National Starch ULC to the Company.

"CIT Fiscal Unity" means the fiscal unity (fiscale eenheid) established between the Seller and the relevant Dutch Group Companies pursuant to article 15 of the Dutch Corporate Income Tax Act 1969 (Wet op de vennootschapsbelasting 1969), as of the date of request to form a fiscal unity and until the termination in accordance with article 15(6) of the Dutch Corporate Income Tax Act 1969;

"Combined Tax Group" means (i) (a) any group of companies that with respect to any period ending on or before the Effective Time, files or has filed Tax Returns on a combined, consolidated or unitary basis with any Group Company or any predecessor of or successor to any Group Company (or another such predecessor or successor) or (b) without limitation of (i)(a), any VAT Fiscal Unity that, in each case, with respect to any period ending on or before the Effective Time, includes or has included any Group Company or any predecessor of or successor to any Group Company (or another such predecessor or successor) and (ii) (a) any group of companies that files or has filed Tax Returns on a combined, consolidated or unitary basis or (b) without limitation of (ii)(a), any VAT Fiscal Unity that, in each case, at any time after the Effective Time, includes any member of Seller's Group on the one hand, and any Group Company on the other hand, or any predecessor of or successor thereof (or another such predecessor or successor);

"Event" means any transaction, event, act or omission;

"Post-Closing Relief" means a Relief arising to a Group Company to the extent that it arises in respect of any Event occurring, or period commencing, after the Effective Time;

"**Project Everest**" means the sale of the NPG business by Seller and the relevant members of Seller's Group to the Company pursuant to the Master Intra-Group Sale and Purchase Agreement relating to the "NPG Business" between Seller and the Company, dated 1 April 2008;

"Relief" means any relief, including, for instance, relief in the form of a reduced capital gain as a result of a step-up in share or asset basis, allowance, amortisation, depreciation, credit, deduction, exemption, tax loss, set-off or other relief of a similar nature granted or available in relation to Tax and any repayment or right to repayment of Tax;

"Taxation" or "Tax" means (i) all forms of taxation whether direct or indirect and whether levied by reference to income, profits, gains, net wealth, asset values, turnover, added value, employment, transaction or transfer value, franchise, capital, import, excise, business and occupation or other reference and statutory, governmental, state, provincial, local

90

Confidential

Execution version

governmental or municipal impositions, duties, contributions, rates and levies (including social security contributions and any other payroll taxes), whenever and wherever imposed (whether imposed by way of a withholding or deduction including as a result of any failure to withhold and deduct for or on account of tax or otherwise) as well as all additional amounts, penalties, fines, charges, costs and interest relating thereto and (ii) any liability of any Group Company for the payment of amounts determined by reference to amounts described in clause (i) in connection with being a member of an affiliated, consolidated, combined or unitary group, or as a result of any obligation of any Group Company under any Tax Sharing Arrangement or Tax indemnification agreement;

"Tax Authority" means any taxing or other authority competent to impose any liability in respect of Tax or responsible for the administration and/or collection of Tax or enforcement of any Law in relation to Tax;

"Tax Liability" means (i) any obligation to make any payment of Tax, (ii) any liability imposed by a Tax Authority to or in respect of any Tax (iii) the use or set off, against income, profits or gains earned, accrued or received or against any liability to any Tax or against any liability imposed by a Tax Authority to or in respect of any Tax, of any Post-Closing Relief or any Relief reflected as an asset or noted as a reduction in the amount of a liability in the relevant Working Capital Statement or Net Debt Statement, in each case, in circumstances where but for such use or set off the relevant party would have had an actual liability to such Tax in respect of which Seller would have been liable under this Agreement; and in the case of (i) through (iii) above, includes any liability in respect of Tax whether or not it is directly or primarily the liability of or chargeable against or to any person (whether or not such person has or may have a right of recovery or reimbursement from any other person);

"Tax Refund" means a credit against, offset, refund or repayment of Tax;

"Tax Return" means any return, declaration, report or information return relating to Taxes, including any schedule or attachments thereto, and including any amendment thereof;

"Tax Sharing Arrangement" means any written or unwritten agreement or arrangement applicable to any Group Company providing for the allocation or payment of Tax liabilities or payment for Tax benefits between or among members of any Combined Tax Group and includes any group payment arrangement; and

"VAT Fiscal Unity" means any group of companies which includes any Group Company together with Seller or any Affiliate of Seller and

- (i) which, for the purposes of VAT in any jurisdiction, is regarded as a single taxable person; or
- (ii) in respect of which, in accordance with applicable Law, any Group Company is or can be made liable for VAT in respect of activities of Seller or any Affiliate of Seller, and includes
 - (*A*) in the Netherlands the fiscal unity (fiscale eenheid) which includes the Seller and any Group Company pursuant to article 7(4) of the Dutch Value Added Tax Act 1968 (Wet op de omzetbelasting 1968); and

91

Confidential

Execution version

(*B*) in the United Kingdom any VAT group for the purposes of sections 43 to 43D of the United Kingdom Value Added Tax Act 1994 which includes any Group Company together with Seller or any Affiliate of Seller.

In this definition any reference to any Affiliate of Seller does not include a Group Company. Solely for purposes of this Tax Indemnity and Paragraph 15 of Schedule 15 (Part 1) (Tax), "Closing Date" shall mean, with respect to any Group Company or Business, the date on which the Closing occurs for such Group Company or Business.

9 Seller's Tax Covenant

- 9.1 In respect of each Business and each Group Company, Seller shall indemnify, defend and hold harmless Purchaser and, as an irrevocable third party stipulation ('derdenbeding'), each member of Purchaser's Group, against any and all Losses incurred by Purchaser or any member of Purchaser's Group (which, for the avoidance of doubt, shall include costs and expenses reasonably incurred by Purchaser or any member of Purchaser's Group, including legal and advisor fees other than costs and expenses referred to in Paragraph 8), arising in connection with, or related to:
 - (i) any Tax Liability of any Group Company in respect of any Event, or in respect of or with reference to any income, profits or gains which were earned, accrued or received, on or prior to the Effective Time;
 - (ii) any Tax Liability with respect to any Business in respect of any Event, or in respect of or with reference to any income, profits or gains which were earned, accrued or received, on or prior to the Effective Time;
 - (iii) any Tax Liability of any Group Company, or for which such Group Company may otherwise be liable, and any Tax Liability with respect to any Business, or for which a Business Purchaser may otherwise be liable, in respect of any taxable period ending on or prior to the Effective Time or, with respect to any taxable period that commences on or prior to the Effective Time and ends after the Effective Time, the portion of such period ending on and including the Effective Time;
 - (iv) any Tax Liability of any Group Company as a result of having been a member of a Combined Tax Group, of a CIT Fiscal Unity or of a VAT Fiscal Unity in respect of any Event or period, or in respect of or with reference to any income, profit or gains which were earned, accrued or received, on or before, or after, the Effective Time;
 - (v) any Tax Liability of any Group Company, or with respect to any Business, in respect of any Event or period, or in respect of or with reference to any income, profits, gains which were earned, accrued or received, on, before or after the Effective Time, in each case arising out of or related directly or indirectly to:
 - (A) Project Everest;
 - (B) the Canadian Butterfly Transaction;

92

Confidential

Execution version

- (*C*) the settlement, distribution, transfer, cancellation, set-off or other disposition of the Intra-Group Indebtedness pursuant to Clauses 6.3 and 7.2.3;
- (D) dividends paid by Grupo ICI Mexico, S.A. de C.V. and its subsidiaries that were not supported by CUFIN;
- (*E*) the failure of the US Company or any Subsidiary of US Company to be disregarded as an entity separate from the holder of its membership interests with respect to any period ending on or before the Effective Time;
- (F) any Tax Liability of any Group Company, the US Seller, any Business Seller, Seller, or any Affiliate of Seller, in respect of any Event, or in respect of or with reference to any income, profits, or gains which were earned, accrued or received after the Effective Time as a result of the performance by either Party of its obligations pursuant to Clauses 6.2.3 and 6.5.9 of this Agreement; or
- (*G*) any Tax Liability of any Group Company or with respect to any Business in respect of any Event, or in respect of or with reference to any income, profits, or gains which were earned, accrued or received outside the ordinary course of business of such Group Company or Business (as the case may be), in the period commencing immediately after the Effective Time and ending on the Closing Date.
- **9.2** Each of the covenants contained in Paragraph 2.1 shall be construed as a separate and independent covenant and shall not be limited or restricted by reference to or inference from the terms of any other covenant, any of the Tax Warranties or any other term of this Agreement.
- **9.3** Seller agrees that in the event of a breach of the warranty contained in Paragraph 15(l) of Schedule 15 (Part 1) (relating to stamp duty and other documentary or registration tax), Seller shall pay to Purchaser by way of liquidated damages an amount equal to any unpaid stamp duty or other documentary or registration tax together with any interest and penalties payable to any Tax Authority in respect thereof.
- 9.4 The covenants contained in Paragraph 2.1 shall not cover any Tax Liability to the extent that
 - (a) a specific provision or reserve in respect of such Tax Liability has been made, (or such Tax Liability has been specifically accrued or noted as a reduction in the value of an asset) in the relevant Working Capital Statement or Net Debt Statement; or
 - (b) such Tax Liability arises or is increased as a result of any change, after the Effective Time, in Tax rates or the passing or coming into force of or any change in Law or generally accepted accounting practice, or in a regulation, directive or requirement, or any withdrawal of any extra-statutory concession by a Tax Authority, save where such passing or change is announced on or before the Signing Date; or

Confidential

Execution version

- (c) such Tax Liability would not have arisen but for, or is increased by, a voluntary transaction, action or omission carried out or effected by or on behalf of any member of the Purchaser's Group, at any time after the Closing Date with knowledge that such transaction, action or omission would give rise to such Tax Liability, other than any such transaction, action or omission: (i) carried out or effected in the ordinary course of business of the relevant member of the Purchaser's Group as carried on from time to time; or (ii) carried out or effected pursuant to an obligation that was entered into on or before the Closing Date and which was binding on the relevant member of the Purchaser's Group; or (iii) carried out or effected to comply with any applicable Law; or (iv) carried out or effected in accordance with the terms of Paragraph 5 or 6; or (v) carried out or effected in reliance on any information or advice received from Seller that was not correct and complete; or (vi) which comprises a change in any accounting or Tax policy, basis or practice of any of the Group Companies as described in paragraph 2.4(e) below which falls within the exception to the exclusion set out at paragraph 2.4(e) below; or (vii) carried out or effected at the request or direction of Seller, or
- (d) notice of a claim in respect of such Tax Liability is not delivered by the Purchaser to Seller within 90 (ninety) days after the later of (i) the expiry of the statutory limitation period (including any extensions thereof) applicable in the relevant jurisdiction for the Tax matter giving rise to such claim and (ii) the date on which an additional assessment is levied in accordance with the relevant applicable law permitting such additional assessment; or
- (e) such Tax Liability arises or is increased as a result of any change in any accounting or Tax policy, basis or practice of any of the Group Companies introduced or having effect after the Closing Date, except to the extent such change is necessary in order to comply with applicable Law or applicable accounting requirements (including requirements that need to be met in order to ensure that transactions between the related parties are on fully arm's length terms) applicable to any Group Company as of, or in respect of any period ending on or before, the Closing Date; or
- (f) such Tax Liability would not have arisen but for a cessation after the Closing Date of trade carried on by the Group Companies on the Closing Date; or
- (g) the Tax Liability has been recovered by Purchaser from third parties (including, for the avoidance of doubt, from employees or under an insurance policy), or Purchaser has otherwise received compensation from third parties for such Tax Liability (including under the Warranties), in each case determined after subtracting costs and expenses (including legal and advisor fees) payable by Purchaser or any member of the Purchaser's Group in obtaining such recovery or compensation and after subtracting Taxes payable by Purchaser or any member of the Purchaser's Group as a result of such recovery or compensation, or
- (h) the Tax Liability arises as a result of Purchaser or any member of Purchaser's Group not complying with its obligations under Paragraph 9;

94

Confidential

Execution version

- (i) such Tax Liability would not have arisen but for the Purchaser or a member of the Purchaser's Group not complying with the post-closing obligation set out in clause 8.12 of the Agreement.
- 9.5 For the purpose of Paragraph 2.1, with respect to any taxable period that commences on or prior to the Effective Time and ends after the Effective Time, (i) the amount of Taxes levied by reference to income, profits or gains, which arises during the taxable period that commences on or prior to the Effective Time and ends on the Effective Time, shall be deemed equal to the amount which would be payable if the taxable period ended on the Effective Time and (ii) the amount of Taxes levied by reference to real or personal property shall be time apportioned between the period starting on the first day of the relevant taxable period and ending on (and including) the Effective Time and the period starting the day after the Effective Time and ending on the last day of the relevant taxable period and shall be based on the amount of real or personal property Taxes for the entire taxable period multiplied by a fraction the numerator of which is the number of calendar days in the relevant portion of the Tax period and the denominator of which is the number of calendar days in the entire taxable period.

10 Offsetting Relief

10.1 If, after Seller has made an indemnification payment pursuant to Paragraph 2.1, the Event, occurrence or payment of any Tax Liability gives rise to any Relief (other than any Relief reflected as an asset or noted as a reduction in the amount of a liability in the relevant Working Capital Statement or Net Debt Statement) for Purchaser or any Group Company, and a liability of Purchaser or such Group Company, as the case may be, to make an actual payment or increased payment of Tax has been satisfied or avoided, or a repayment of Tax has been obtained, in whole or in part by the use of that Relief (an "Offsetting Relief") with the amount satisfied, avoided or repaid pursuant to the use of an Offsetting Relief pursuant to this Paragraph 3.1 (referred to herein as "Cash Tax Savings"), then the Relevant Amount (as defined in Paragraph 3.3 below) in respect of any Offsetting Relief shall be paid by Purchaser to Seller within 20 (twenty) Business Days following the date on which Purchaser realises the Cash Tax Savings attributable to such Offsetting Relief.

10.2 In relation to any Offsetting Relief:

- (a) Purchaser shall only be obliged to account to the Seller in accordance with this Paragraph 3.2 for any Relevant Amount in respect of such Offsetting Relief to the extent that it is satisfied (acting reasonably) that such accounting to Seller will not (x) prejudice any of the entitlement of Purchaser or the relevant Group Company to such Offsetting Relief nor (y) result in the loss, nullification, cancellation or reduction of the Cash Tax Savings resulting from the utilisation of such Offsetting Relief; and
- (b) Purchaser shall not be obliged to account to Seller in accordance with this Paragraph 3.2 for any Relevant Amount in respect of such Offsetting Relief to the extent that Seller Affiliate of Seller has otherwise derived a benefit from such Offsetting Relief, including (i) where such Relief has been taken into account in computing any liability of Seller or any member of Seller's Group pursuant to Clause

95

Confidential

Execution version

10.8 or (ii) such Offsetting Relief has been used or offset by Purchaser or a Group Company in reducing or eliminating any Tax Liability in circumstances where but for such use or offset Purchaser or the relevant Group Company would have had a Tax Liability in respect of which a claim might have been brought under this Tax Indemnity or under the Tax Warranties.

- 10.3 The Relevant Amount in respect of an Offsetting Relief shall be an amount equal to the amount of any Cash Tax Savings obtained by Purchaser or a Group Company from the utilisation of the Offsetting Relief, less (in each case) the amount of all reasonable costs and expenses (including legal and advisor fees) reasonably incurred by Purchaser or the relevant Group Company in establishing the entitlement to such Offsetting Relief and/or in connection with the utilisation of such Offsetting Relief to obtain the Cash Tax Savings.
- **10.4** Notwithstanding the foregoing, the obligation of Purchaser to pay Seller any amount pursuant to Paragraph 3.1 on account of any Offsetting Relief shall expire 6 (six) years after the date on which the Event, occurrence or payment of Tax Liability giving rise to such Offsetting Relief occurs.

11 Tax Refund

- 11.1 If after the Closing Date Purchaser or the Group Companies actually receive a Tax Refund of a Tax that:
 - (a) a Group Company paid on or before the Effective Time or for which a specific reserve or provision in respect of such Tax was made (or specifically accrued or noted as reduction in the value of an asset) in the relevant Working Capital Statement or Net Debt Statement; and
 - (b) such Tax Refund has not been reflected as an asset or noted as a reduction in the amount of a liability in the relevant Working Capital Statement or Net Debt Statement; and
 - (c) such Tax Refund is not the result of the carryback or other utilisation of any Post-Closing Relief; and
 - (d) such Tax Refund does not comprise a credit or offset of a right to a refund or repayment of Tax against a Liability to Tax in circumstances where, but for such credit or offset, a claim might have been brought under this Tax Indemnity or under the Tax Warranties, then
 - (i) the Purchaser will promptly notify the Seller and will take such action as may reasonably be requested in writing by Seller at Seller's sole expense, to obtain such Tax Refund; and
 - (ii) the amount of the Tax Refund shall be paid by the Purchaser to the Seller within ten (10) Business Days of the receipt of the Tax Refund.

11.2 Notwithstanding the foregoing:

(a) if any Tax Refund for which payment is made to Seller is later reduced or disallowed, Seller shall promptly repay to Purchaser the amount of such reduction or

96

Confidential

Execution version

disallowance and shall indemnify and hold harmless Purchaser in connection with such reduction, disallowance or repayment;

- (b) the amount of any Tax Refund payable by Purchaser to Seller shall be determined after subtracting any costs and expenses (including legal and advisor fees) reasonably incurred by Purchaser or relevant Group Company, in obtaining such Tax Refund and any Taxes payable by Purchaser or any member of the Purchaser's Group as a result of the receipt of such Tax Refund; and
- (c) the obligation of Purchaser to pay Seller any Tax Refund shall expire 90 (ninety) days after the expiration of the applicable statute or limitations for collecting the Tax to which the Tax Refund relates.

12 Mitigation

Purchaser shall, at the direction of Seller and at Seller's sole expense, procure that the Group Companies take all such steps as Seller may reasonably request in writing to make all such claims and elections specified by Seller in respect of any period commencing before the Effective Time having the effect of reducing or eliminating any Tax Liability referred to under Paragraph 2.1; provided in each case (x) Purchaser shall not be required to make any claim or election if doing so would increase Taxes for which any member of Purchaser's Group is liable, (y) Seller shall reimburse Purchaser for any reasonable costs or reasonable expenses incurred by Purchaser in making any such claim or election, and (z) Seller shall indemnify and hold harmless Purchaser and each member of Purchaser's Group against any Loss incurred by any of them arising in connection with making any such claim or election.

13 Purchaser's Tax Covenant

- 13.1 Purchaser shall indemnify, defend and hold harmless Seller and, as an irrevocable third party stipulation ('derdenbeding'), each Business Seller against any Tax for which any director of the Seller or of another member of the Seller's Group (excluding the Group Companies) becomes liable as a result of the failure of the Purchaser or any Group Company to discharge a Tax that under applicable Law Purchaser or any Group Company is obligated to discharge following the Closing Date; provided, however, that Purchaser shall not be liable for or pay, and shall not indemnify or hold harmless Seller or any Business Seller from and against any Taxes for which Seller is liable under this Agreement (including under the Tax Warranties and this Tax Indemnity).
- 13.2 The covenant contained in paragraph 6.1 shall not cover any Tax to the extent that notice of a claim in respect of such Tax is not delivered by Seller to Purchaser within 90 (ninety) days after the later of (i) the expiry of the statutory limitation period (including any extensions thereof) applicable in the relevant jurisdiction for the Tax matter giving rise to such claim and (ii) the date on which an additional assessment is levied in accordance with the relevant applicable Law permitting such additional assessment.

97

Confidential

Execution version

14 Due date for payment

A payment to be made by the Seller to the Purchaser under Paragraph 2 or by the Purchaser to the Seller under Paragraph 6 shall be made within 20 (twenty) Business Days after the earliest of:

- (a) the date on which a compromise or settlement has been reached between the Parties with respect to a claim on the basis of Paragraph 2 or Paragraph 6;
- (b) the date on which a compromise or settlement has been reached with the Tax Authority in respect of Tax subject to a claim on the basis of Paragraph 2 or Paragraph 6; and
- (c) the date on which a final decision is made with respect to Tax subject to a claim on the basis of Paragraph 2 or Paragraph 6 against which no appeal is permitted;

provided, that in no event shall the due date for payment be later than five (5) Business Days before the date on which the Tax giving rise to such payment obligation is required to be paid to the applicable Governmental Authority.

15 Preparation of Tax Returns

- 15.1 The Seller shall, at its own cost, prepare or procure the preparation of (i) all Tax Returns in respect of the Group Companies to the extent required to be filed on or prior to the Closing Date, and (ii) all corporate income tax returns and VAT returns in respect of the Group Companies for periods during which the relevant Group Companies were members of a Combined Tax Group. All Tax Returns which Seller is required to file or cause to be filed in accordance with this Paragraph 8.1 shall be prepared and filed in a manner consistent with applicable Law and, subject thereto, past practice and, on such Tax Returns, no position shall be taken, election made or method adopted that is inconsistent with positions taken, elections made or methods used in preparing and filing similar Tax Returns in prior periods. The Purchaser shall, at its own expense, within 60 days after Closing complete income tax information packages consistent with historic practices to permit the preparation of Seller's federal and state U.S. income Tax Returns. The Purchaser shall provide the Seller such other information and render the Seller such assistance as may reasonably be requested in order to ensure the proper and timely completion and filing of such Tax Returns.
- **15.2** The Purchaser shall, at its own cost, prepare (or procure the preparation of), and duly and properly file or procure to be filed with each relevant Tax Authority, all Tax Returns in respect of the Group Companies to the extent required to be filed after the Closing Date and not covered by Paragraph 8.1. The Seller shall provide the Purchaser such information and render the Purchaser such assistance as may reasonably be requested in order to ensure the proper and timely completion and filing of such Tax Returns.

16 Tax Audits

16.1 Upon becoming aware of any pending audit, investigation, assessment or other material proceedings with respect to Tax matters of the Group Companies that may affect a Tax Liability for which the other Party may be liable under this Agreement, the Seller shall, or as the case may be, the Purchaser shall, within 20 (twenty) Business Days give notice of such audit, investigation, assessment or proceeding to the other Party; provided, that

Confidential

Execution version

failure of the notifying Party to comply with this provision shall not affect such Party's right to indemnification hereunder except to the extent such failure impairs the other Party's ability to contest any such Tax Liability. The notice shall set out such information as is available and as is reasonably necessary to enable such other Party to assess the merits of the claim, to act to preserve evidence and to make such provision as such other Party may consider necessary.

- **16.2** If the Seller so requests in writing, ultimately within 20 (twenty) Business Days of receiving notice of or giving notice to the Purchaser in accordance with Paragraph 9.1, the Seller shall have the right (but not the obligation), at its sole cost and expense, to control the conduct of any audits or other proceedings if and to the extent that these relate to:
 - (i) any accounting period for Tax purposes ending on or prior to the Effective Time; and
 - (ii) corporate income Tax or VAT matters of the Group Companies relating to periods during which the relevant Group Companies were members of a Combined Tax Group;

Notwithstanding the foregoing:

- (i) Seller shall have no right to represent any Group Company's interests in any Tax audit or administrative or court proceeding unless Seller shall have first notified Purchaser in writing that Seller agrees with Purchaser that Seller shall be liable for any Losses that result from Seller's conduct of such audit or proceeding and for any costs and expenses (including legal and advisor fees) reasonably incurred by any member of Purchaser's Group to facilitate Seller's conduct of such audit or proceeding;
- (ii) Purchaser and its representatives shall be permitted, at Purchaser's expense, to be present at, and participate in, any such audit or proceeding.
- (iii) Seller shall not be entitled to settle, either administratively or after the commencement of litigation, any claim for Taxes which could adversely affect a liability for Taxes of any member of Purchaser's Group to any extent, unless Seller has indemnified each such member of Purchaser's Group against the effects of any such settlement (including the imposition of Tax deficiencies, the reduction of asset basis or cost adjustments, the lengthening of any amortization or depreciation periods, the denial of amortization or depreciation deductions, or the reduction of loss or credit carryforwards), without the prior written consent of Purchaser, which consent may be withheld in the sole discretion of Purchaser;
- (iv) If Seller is not able or permitted to, or otherwise does not, control any such audit or proceeding, Purchaser may do so (and in doing so Purchaser shall not waive any right against Seller under this Agreement) and Seller shall pay, and indemnify Purchaser against, reasonable costs and expenses (including legal and advisor fees) incurred by Purchaser in doing so. Nothing herein shall be construed to impose on Purchaser any obligation to defend any Group

99

Confidential

Execution version

Company in any Tax audit or administrative or court proceeding.

- 16.3 The Purchaser shall control the conduct of any audits or other proceedings relating to Tax matters of the Group Companies except as provided in Paragraph 9.2, shall be entitled to settle and compromise any such audit or proceeding and shall have no liability to Seller relating to any such audit or proceeding, settlement or compromise (including relating to any decision not to conduct any such proceeding). To the extent that any such audits or other proceedings relate to a period that starts prior to, and ends after, the Effective Time or for which Seller would be obligated to indemnify Purchaser pursuant to Paragraph 9.2:
 - (i) Purchaser shall advise the Seller periodically of developments in the audit investigation or other proceedings and obtain the Seller's prior written approval (such approval not to be unreasonably withheld, conditioned or delayed) on critical decisions and on material written communications to be forwarded to any Tax Authority or competent court in relation to the audit or other proceedings if and to the extent they regard Taxes attributable to the period prior to the Effective Time; and
 - (ii) Seller and its representatives shall be permitted, at Seller's expense, to be present at, and participate in, any such audit or proceeding.
- **16.4** Seller and the Purchaser shall provide each other such information and render such assistance as may reasonably be requested in order to ensure the proper and adequate defence of any audit or other proceeding including audits or other proceedings of Seller or Purchaser Tax Returns insofar as pertaining to the Group Companies or the Businesses.

17 Fiscal Unity

17.1 Parties acknowledge that the Group Companies shall be separated from the CIT Fiscal Unity and VAT Fiscal Unity ultimately as from the Effective Time or as from an earlier date to be agreed upon between the Parties. As soon as reasonably practicable following the Effective Time, Seller shall provide Purchaser with (i) an opening balance sheet for Dutch corporate income tax purposes of the relevant Group Companies as at the Effective Time and (ii) explanatory notes thereto.

- 17.2 If for Tax purposes a revaluation of the assets and/or the liabilities of a Group Company is required, pursuant to the application of article 15ai of the Dutch Corporate Income Tax Act 1969 (Wet op de vennootschapsbelasting 1969) at the level of the CIT Fiscal Unity, and Seller or the member of the Seller's Group is liable to Tax resulting from such revaluation, then Purchaser will compensate Seller for any additional Taxes actually paid as a consequence of such revaluation to the extent and in an amount that the Group Company actually (i) receives a refund or reduction of Tax or (ii) benefits from a reduction in Tax payable resulting from such revaluation.
- 17.3 The amount mentioned in Paragraph 10.2 will be computed and paid for in the year during which a final assessment for the relevant year has been imposed on the relevant Group Company and such assessment has not been appealed. Purchaser may at any time set off the amount of compensation determined in accordance with this Paragraph 10.3 against

100

Confidential

Execution version

any claims, which it may have against Seller under this Agreement.

- **17.4** If, notwithstanding Paragraph 10.1, any Group Company is not separated on or before the Effective Time from a VAT Fiscal Unity then, in respect of the period from the day immediately after the Effective Time to the date of such ultimate separation (the "Post-Closing Period"):
 - (a) Purchaser shall or shall procure that the relevant Group Company shall pay to Seller or relevant member of Seller's Group as Seller may direct an amount (if any) equal to the excess of (i) the amount of output VAT (in respect of any supply, acquisition or importation) attributable to the activities of the relevant Group Company in respect of the Post-Closing Period which output VAT has been received by the relevant Group Company over (ii) any deductible input VAT attributable to the activities of the relevant Group Company in respect of the Post-Closing Period; and
 - (b) Seller shall or shall procure that the relevant member of Seller's Group shall pay to Purchaser or relevant member of the Purchaser's Group as Purchaser may direct an amount (if any) equal to the excess of (i) any deductible input VAT attributable to the activities of the relevant Group Company in respect of the Post-Closing Period over (ii) output VAT (in respect of any supply, acquisition or importation) attributable to the activities of the relevant Group Company in respect of the Post-Closing Period which output VAT has been received by the relevant Group Company.

The Seller and the Purchaser agree to use all reasonable efforts to separate any relevant Group Company from the relevant VAT Fiscal Unity as soon as possible following the Effective Time so that the length of any Post-Closing Period shall be minimised.

17.5 The Seller and the Group Companies shall jointly file a request to allocate loss carry forwards of the CIT Fiscal Unity to the extent any such loss carry forward was reflected as an asset in the relevant Working Capital Statement or as a reduction to a liability in the relevant Working Capital Statement or Net Debt Statement, to the extent possible, to the relevant Group Companies in accordance with article 15af of the Dutch Corporate Income Tax Act 1969 (Wet op de vennootschapsbelasting 1969).

18 Costs

Notwithstanding anything to the contrary in this Schedule, the covenants to pay included in Paragraph 2 and Paragraph 3 shall not extend to costs and expenses (including legal and advisor fees) which a Party has incurred in the preparation or filing of the relevant claim with the other Party.

19 No Double Claims

19.1 Neither Party shall be entitled to recover from the other Party under this Tax Indemnity more than once in respect of the same Losses suffered in respect of an Event. Notwithstanding anything to the contrary in this Agreement, in the event that any matter, act, omission or circumstance (or any combination thereof) giving rise to a breach of a Tax Warranty is also the subject of an indemnity under this Tax Indemnity, the claim for such breach of Tax Warranty shall be limited to a claim under this Tax Indemnity.

101

Confidential

Execution version

19.2 Notwithstanding anything to the contrary in this Agreement, where the same Relief, benefit or advantage gives rise to an entitlement of Seller under two or more provisions of this Agreement, Seller shall not be entitled to repayment, recovery of, exclusion or limitation of liability in respect of or credit for the same amount in respect of such Relief, benefit or advantage more than once.

20 Application to Tax Warranties

Paragraph 2.2 (independent covenants), Paragraph 2.4 (exclusions), Paragraph 3 (Offsetting Relief), Paragraph 5 (Mitigation), Paragraph 7 (Due Date for Payment), Paragraph 9 (Tax Audits), Paragraph 11 (Costs) and Paragraph 12 (No Double Claims) shall *mutatis mutandis* apply to the Tax Warranties.

Confidential

Execution version

Schedule 14 (Part 2) Intellectual Property Matters

- 1 Seller shall, at its expense, use its commercially reasonable efforts to complete the recordal in the relevant jurisdictions of all assignments of the Registered Group Intellectual Property to the relevant member of Purchaser's Group, within six (6) months after the Closing Date.
- 2 In the event that the Parties determine after Closing that the Acquired Registered Intellectual Property does not include, by inadvertence or otherwise, and/or improperly includes any Registered Intellectual Property, then the Parties shall take all such actions and execute and deliver all such documents or agreements as may be reasonably necessary to correct the foregoing.

103

Confidential

Execution version

Schedule 15 (Part 1) Seller's Warranties

(Clause 9.1)

3 Incorporation, authority, corporate action, existence

- (a) Each of Seller, the Share Seller, US Seller and each Business Seller is validly existing and is a company duly incorporated under the Law of its jurisdiction of incorporation.
- (b) Each of Seller, the Share Seller, US Seller and each Business Seller has the full power and authority to enter into and perform this Agreement and any other documents to be executed by such person pursuant to or in connection with this Agreement, which, when executed, will constitute valid and binding obligations on such person, in accordance with their respective terms.
- (c) Each of Seller, the Share Seller, US Seller and each Business Seller has taken or will have taken by Closing all corporate action required by it to authorise it to enter into and to perform this Agreement and any other documents to be executed by it pursuant to or in connection with this Agreement.
- (d) Each Group Company is validly existing and is a legal entity duly incorporated or organised, as the case may be, under the Law of its jurisdiction of incorporation or organisation, as the case may be.
- (e) Each Group Company and Business Seller is duly licensed or qualified to do business where the ownership, lease or operation of its assets and properties or the conduct of its business requires such license or qualification. To the extent the concept of good standing exists under applicable Law, each Group Company and Business Seller is in good standing as a foreign corporation in those jurisdictions where it owns, leases or operates its assets and properties relating to the Operations.
- (f) The execution, delivery and performance by Seller of this Agreement and the consummation of the transactions contemplated hereby, does not and will not (i) violate any provision of the Constitutional Documents of any member of Seller's Group; (ii) violate, or result in a breach of, or constitute a default under, any Law to which any member of Seller's Group or any of the Business Assets or Company Assets is subject; or (iii) result in the creation of any Encumbrances except for Permitted Encumbrances on the Business Assets or Company Assets.
- (g) No Group Company or Business Seller is insolvent under the Law of its jurisdiction of incorporation or unable to pay its debts as they fall due.
- (h) There are no proceedings in relation to any compromise or arrangement with creditors or any winding up, bankruptcy or other insolvency proceedings concerning any Group Company or Business Seller and, so far as Seller is aware, no events have occurred which, under applicable Law, would justify such proceedings.

104

Confidential

Execution version

(i) So far as Seller is aware, no event has occurred to give the right to enforce any security over any assets of any Group Company or Business Seller and no steps have been taken to enforce any such security.

4 Corporate information

- (a) The Shares, US Interests and Group Companies
 - (i) The Share Seller is the sole record, legal and beneficial owner of the Shares.
 - (ii) The Shares comprise the whole of the issued share capital of the Company and have been duly authorised, properly and validly issued and are each fully paid and non-assessable, and were issued in compliance with all applicable stock ownership or share capital Laws of the jurisdiction of organization of the Company.
 - (iii) The US Seller is the sole record, legal and beneficial owner of the US Interests.
 - (iv) The US Interests comprise all of the membership interests in the US Company and have been duly authorized, properly and validly issued and are fully paid and non-assessable and were issued in compliance with all limited liability company Laws of the jurisdiction of organisation of the US Company.
 - (v) In respect of each Subsidiary, the shareholders or other equity holders specified in respect of the Subsidiary in Schedule 2 (Part 1) are the sole record, legal and beneficial owners of the shares or other equity interests in the Subsidiary.
 - (vi) In respect of each Subsidiary, the specified shareholder or other equity holder in respect of the Subsidiary holds the whole of the issued share capital or other equity interests of the Subsidiary, the shares or other equity interests of which have been duly authorised, properly and validly issued, fully paid and non-assessable, and were issued in compliance with all applicable stock ownership, limited liability company or share capital Laws of the jurisdiction of organization of such Subsidiary as applicable.
 - (vii) Except for pre-emption rights held by a Group Company with respect to the capital of another Group Company, no person has the right, or has claimed to have the right, (whether exercisable now or in the future and whether contingent or not) to call for the conversion, issue, registration, sale or transfer, amortisation or repayment of any share capital or any other security giving rise to a right over, or an interest in, the capital of any Group Company under any option, warrant agreement or other arrangement (including conversion rights, rights of pre-emption, rights of first refusal or first offer, subscription rights, put or call rights or other similar rights).
 - (viii) There are no Encumbrances on any of the Shares, the US Interests or the shares or other equity interests of the Subsidiaries.

105

Confidential

Execution version

- (ix) Upon transfer of the Shares and the US Interests to the Share Purchasers in accordance with the terms of this Agreement, the Share Purchasers will receive good and valid title to such Shares and US Interests, as applicable, free and clear of all Encumbrances.
- (b) Constitutional Documents, corporate registers and minute books, officers and directors
 - (i) The Constitutional Documents of the Group Companies made available in section 5 of the Data Room are correct and complete copies of the Constitutional Documents of the Group Companies at Signing and such Constitutional Documents have not been amended, supplemented, modified or rescinded. So far as Seller is aware, as at Signing there have not been and are not any breaches by any Group Company of its Constitutional Documents.
 - (ii) So far as Seller is aware, as at Signing the registers and minute books required to be maintained by each Group Company under the Law of the jurisdiction of its incorporation or organization (i) are up-to-date, (ii) have been maintained in accordance with applicable Law and (iii) contain correct and complete records of all matters required to be dealt with in such books and records.
 - (iii) All such registers and minute books are in the possession, or under the control, of the relevant Group Company and at Closing will either be in the possession of the relevant Group Company or delivered to the Share Purchaser or US Purchaser as applicable.
 - (iv) All material filings, publications, registrations, fees and other formalities required by applicable Law to be delivered, made or paid by the Group Companies, and, in relation to the Businesses, by the Business Sellers, to company registries in each relevant jurisdiction have been delivered, made or paid on a timely basis.
 - (v) Section 5.1.2 of the Data Room contains a list of the officers, directors or persons holding similar positions of each of the Group Companies at Signing.

5 Accounts

The Accounts present fairly, in all material respects, the assets and liabilities of the Group as of 31 December 2008 and 31 December 2009, and the results of its operating activities and its identifiable cash flows for the years then ended, on the basis of accounting described in Notes 1 and 2 to the Accounts.

6 Guarantees

There is no outstanding Guarantee given:

- (a) by any Group Company or, in relation to any Business, by the Business Seller; or
- (b) for the benefit of any Group Company or, in relation to any Business, Business Seller;

106

Confidential

Execution version

in each case, with respect to any liability or contingent liability in excess of USD 2,000,000 (two million US dollars).

7 Assets

For purposes of this Paragraph 5:

"Property Encumbrance" means any charge, pledge, mortgage, lien, option, equity, power of sale, hypothecation, usufruct, retention of title, right of pre-emption, right of first refusal or other similar third party rights or security interest of any kind or an agreement to create any of the foregoing, and any easement, quasi-easement, license, right-of-way, right of re-entry, encroachment, covenant, restriction, defect in title, zoning, entitlement or other land use regulation or other similar encumbrance; and

"Permitted Property Encumbrances" means (i) the Encumbrances set out in the Accounts or Schedule 21, (ii) security interests for Taxes and other governmental charges which are not due and payable or which may be paid without penalty, (iii) zoning, entitlement or other land use regulations that do not adversely impact in any material respect the use of the Property for the business purposes conducted by the Group thereon at Closing, (iv) easements, quasi-easements, licenses, covenants, rights-of-way, rights of re-entry or other restrictions, including any other agreements, conditions or restrictions that would be shown by a current title report or other similar report or listing or, in the UK, any overriding interest, which do not underlie any buildings or improvements composing any portion of the Properties and which do not adversely impact in any material respect the use of the Properties for the business purposes conducted by the Group thereon as of the Closing Date and (v) any reasonable conditions that may be shown by a current survey or physical inspection of the Properties, which do not adversely impact in any material respect the use of such Properties for the business purposes conducted by the Group thereon at Closing.

(a) Properties

- (*i*) In respect of each Group Company and, in relation to a Business, each Business Seller, sections 12.14.4, 12.14.5, 12.14.6 and 12.14.7 of the Data Room contains a list, as at Signing, of all real property (the "**Properties**") owned, leased or in respect of which said Group Company or Business Seller, in relation to a Business, has any other interest.
- (ii) In respect of each Property thus listed as being owned by a Group Company, the Group Company named as the owner is the legal and beneficial owner of the whole Property and has good and valid title to the Property. The Property is registered in the competent real estate register for the benefit of the relevant Group Company where the relevant jurisdiction requires the registration of ownership regarding real estate.
- (iii) No Property (excluding Property leased by any Group Company or, in relation to a Business, any Business Seller) is subject to any Property Encumbrance other than Permitted Property Encumbrances.

107

Confidential

Execution version

- (iv) No Group Company or Business Seller has received any written claim from any third party disputing such Group Company's or Business Seller's ownership, occupation or use of any Property which, if such third party prevailed, would reasonably be expected to have a material adverse effect on the Operations carried out on the Property.
- (v) No Group Company or Business Seller has received any written notice as to any contravention of the relevant planning legislation or regulations in relation to any Property which, if implemented or enforced, would reasonably be expected to have a material adverse effect on the Operations carried out on the Property;
- (vi) So far as Seller is aware, each Property that is material to the Operations is in compliance with all Laws that affect the use of such Property, including zoning, building and other similar restrictions, except as would not have a material adverse effect on the Operations as carried out on such Property.

(b) Leases

Where any Property is leased by a Group Company or, in relation to a Business, by a Business Seller:

- (iii) Details, as at Signing, of the lessor, lease period, address, monthly rental, and purpose, together with a correct and complete copy of the relevant lease agreement or agreements, have been included in the documents contained in section 12 of the Data Room;
- (iv) there is no subsisting breach and no non-observance of any covenant, condition or agreement contained in the lease under which the Group Company or Business Seller holds its interest in the Property, on the part of any Group Company or Business Seller or, so far as Seller is aware, as at Signing the relevant landlord, which would reasonably be expected to have a material adverse effect on the Operations carried out on the Property; and
- (*v*) so far as Seller is aware, there is no right for the landlord to terminate the lease before the expiry of the contractual term other than as a result of breach of the lease by the lessee.
- (c) Properties subject to third party leases and other agreements

Where any Property is the subject of any lease or other agreement relating to its occupation or use for the benefit of any person other than a Group Company or Business Seller:

- (i) all those details, as at Signing, regarding lessor, lease period, address, (floor) area, monthly rental, and purpose, together with a correct and complete copy of the relevant lease agreement or agreements, as are available to the relevant Group Company or Business Seller, have been included in the documents contained in section 12 of the Data Room; and
- (ii) Seller is not aware of any subsisting breach or non-observance of any

108

Confidential

Execution version

covenant, condition or agreement under such lease or agreement which, in each case, would have a material adverse effect on the Operations carried out on the Property.

(d) Ownership and transfer of assets

Each of those tangible Business Assets and each of those tangible assets of the Group Companies or which are reflected in the Accounts, other than the Properties, Group Intellectual Property and any assets disposed of or realised in the ordinary course of business:

- (i) is legally and beneficially owned by the relevant Business Seller or Group Company, except for assets which are subject to hire purchase or lease arrangements or other Contracts for use where the hire purchase or lease or use payments do not exceed USD 1,000,000 (one million US dollars) per year;
- (ii) is, where capable of possession, in the possession or under the control of the relevant Group Company or Business Seller; and
- (iii) will be transferred to the Business Purchasers or owned by the Group Companies at Closing, in each case, free from Encumbrances except for Permitted Encumbrances.
- (e) Sufficiency of assets

The Business Assets, the assets owned or leased by the Group Companies and the rights of the Group Companies, together with the rights and assets conferred upon or acquired by Purchaser or any member of the Purchaser's Group by this or pursuant to this Agreement, the IP Licence and the Transitional Services Agreement, together comprise in all material respects all of the rights and assets used by Seller's Group for the carrying on of the Operations in substantially the same manner that it was in the 12 months immediately preceding Signing.

(f) Plant and machinery and other fixed assets

The plant and machinery, vehicles and other equipment owned or used by the Business Sellers or Group Companies which are material to the Operations have been maintained in accordance with maintenance programmes adequate for the demands of production, and are in a reasonable condition having regard to their age and use.

(g) No options

No Business Seller or Group Company is obligated under, or a party to, any option, right of first refusal, right of first offer, put option, equity participation right or any other contractual right to offer, purchase, acquire, sell, assign or dispose of any portion of any of the Properties. Except for the relevant Business Seller or Group Company, no third party has any right to use, lease, possess and/or occupy any material portion of the Properties and there are no oral or written agreements between any Business Seller or Group Company and any other third party providing such third party any right to use, occupy or posses all or any material

portion of any of the Properties.

(h) Right to quiet enjoyment of leased Property

So far as Seller is aware, except as set out in Schedule 3 (Part 1), the relevant Business Seller or Group Company has the right to quiet enjoyment of each of the Properties subject to a lease (under which the relevant Business Seller or Group Company is the lessee) for the full term of the lease or similar agreement (and any renewal option related thereto) relating thereto, and, so far as Seller is aware, as at Signing the leasehold or other interest of the relevant Business Seller or Group Company is not subject or subordinate to any Encumbrance except for Permitted Encumbrances.

(i) No suit for condemnation

So far as Seller is aware, neither the whole nor any part of the Property is subject to any pending suit for condemnation or other taking by any Governmental Authority, and, so far as Seller is aware, no such condemnation or other taking is threatened or contemplated.

8 Intellectual Property and information technology

- (a) Sections 10 and 11 of the Data Room contain, as at Signing, a correct and complete list of all Acquired Registered Intellectual Property.
- (b) Sections 11, 13 and 16 of the Data Room contain copies of all licenses and authorities for use of Intellectual Property material to the Operations as at Signing (excluding any shrink-wrap or click through licenses for computer software) to which a Group Company or Business Seller is a party and that are used by any Group Company or Business Seller in the Operations, and as at Signing, there are no amendments or modifications to any such license agreement other than those included in section 11, 13 and 16 of the Data Room.
- (c) The Group Companies or the Business Sellers, as applicable own the entire right, title and interest in and to the Acquired Intellectual Property including the right to treat and/or dispose of such Acquired Intellectual Property in accordance with this Agreement.
- (d) So far as Seller is aware, save as disclosed otherwise in section 10 and 11 of the Data Room, the processes employed and the products manufactured or sold in the Operations do not infringe or misappropriate any rights in Intellectual Property of any third party. No member of the Seller's Group has received any written claim of any infringement or misappropriation of any Intellectual Property of any third party in respect of the Operations during the 24 (twenty-four) months immediately prior to the Signing Date.
- (e) The Acquired Intellectual Property, the Intellectual Property licensed under the licenses and authorities referred to in Paragraph 6(b) and rights under the IP Licence, the Transitional Services Agreement and this Agreement, taken together, include all material Intellectual Property primarily used in or necessary for the

110

Confidential

Execution version

carrying on of the Operations (including the developments of Products) in substantially the same manner as carried on in the 12 (twelve) months immediately prior to the Signing Date.

- (f) So far as Seller is aware, except as disclosed otherwise in section 10 of the Data Room, (i) there are no circumstances that may affect the enforceability and validity of the registrations for Acquired Registered Intellectual Property owned by a Group Company or Business Seller; and (ii) a Group Company or Business Seller has the sole and exclusive right to bring actions for infringement or unauthorised use of the Acquired Intellectual Property. Except as disclosed otherwise in section 10 of the Data Room, there are no pending reexaminations, oppositions or cancellation proceedings involving the Acquired Registered Intellectual Property. Except as disclosed otherwise in section 10 of the Data Room, as at Signing, all necessary registration, maintenance and/or annuity or renewal fees with respect to the Acquired Registered Intellectual Property have been paid and are up to date.
- (g) Other than (i) licence or other agreements contained in section 11, 13 and 16 of the Data Room; (ii) non-exclusive end-user licenses granted by a Group Company or Business Seller in the ordinary course, (iii) the IP Licence and (iv) confidentiality agreements entered into the ordinary course, none of the Acquired Intellectual Property material to the Operations is licensed to any third party or is a subject of any agreement not to bring an action against any third party for infringement, misuse or misappropriation.
- (h) So far as Seller is aware, with respect to all Intellectual Property of third parties that is licensed to a Group Company or a Business Seller and which is used in the Operations: (i) the applicable license agreements for such Intellectual Property are in full force and effect; (ii) a Group Company or a Business Seller, as applicable, is in compliance with the terms and conditions of the applicable licence agreement;
- (i) So far as Seller is aware, all employees who have contributed to or participated in the creation or development of any patentable, copyrightable or trade secret material on behalf of the Group Companies or Business Sellers: (i) is a party to an agreement under which a Group Company or Business Seller is deemed to be the original owner/author of all property rights therein; or (ii) has executed an assignment or an agreement to assign in favour of a Group Company or Business Seller all right, title and interest in such material, to the extent assignable.
- (j) Each Group Company and Business Seller has entered into agreements with its employees, directors, agents and contractors to maintain the confidentiality of the confidential information of the Group Companies and Business Sellers, the value of which is dependent upon

- the maintenance of the confidentiality thereof. So far as Seller is aware, there has been no unauthorised disclosure or use of confidential information of the Group Companies and Business Sellers.
- (k) Section 11.3 of the Data Room contains Seller's best estimate of the cost to acquire and deploy the network and infrastructure resources necessary to make

111

Confidential

Execution version

the information technology used to support the Operations self-sufficient and independent from the Seller's Group following the Closing.

9 Contracts and other agreements

- (a) Contracts
 - (*i*) No Group Company or, with respect to the Operations, Business Seller is a party to or subject to any Group Contract (other than in relation to any property lease or Contract of employment) that:
 - (A) is not in the ordinary course of business and involves quantities beyond those normally contracted for; or
 - (B) is not on an arm's length basis.
 - (ii) All of the following Group Contracts as at Signing are contained in sections 2, 5, 8, 9.7, 11, 12, 13,16 and 18 of the Data Room (each, a "*Material Contract*"):
 - (*A*) any (1) Group Contract with any Senior Employee relating to employment, compensation or severance payments or (2) Group Contract with any labour union relating to the Operations;
 - (*B*) other than non-exclusive end-user licenses, any license, licensing arrangement or other Group Contract providing, in whole or in part, for the use of, or limiting the use of, any Intellectual Property material to the Operations;
 - (C) any Group Contract relating to a joint venture, consortium, partnership or other association (other than a recognised trade association in relation to which the Group Company or Business Seller has no liability or obligation except for the payment of annual subscription or membership fees) or involving a sharing of profits or expenses with respect to the Operations or any part thereof (including joint research and development and joint marketing Contracts) which (i) contains an obligation of any Group Company or Business Seller to make payments in excess of USD 2,000,000 (two million US dollars) in the aggregate, or (ii) relates to a product or product line, business or other operation that contributes or is expected to contribute more than USD 2,000,000 (two million US dollars) to the annual revenues of the Group, or (iii) does not provide for control by a Group Company of such joint venture in which it has an equity participation;
 - (*D*) any Group Contract for the purchase or lease of any materials, products, services or supplies relating to the Operations that involves annual payments in excess of USD 2,000,000 (two million US dollars) on an aggregate basis or pursuant to which payments of at least USD 1,000,000 (one million US dollars) were made during the 12 (twelve) months ended on the Accounts Date;

112

Confidential

Execution version

- (*E*) any Group Contract for the sale of products of the Group exceeding USD 2,000,000 (two million US dollars) in aggregate sales or pursuant to which sales of at least USD 1,000,000 (one million US dollars) were made during the 12 (twelve) months ended on the Accounts Date;
- (F) any Group Contract with any agent, dealer or other independent distributor of products of the Group which involves sales of products of the Group exceeding USD 2,000,000 (two million US dollars) in the aggregate or sales of at least USD 1,000,000 (one million US dollars) during the 12 (twelve) months ended on the Accounts Date, or grants any such agent, dealer or distributor any exclusive right to sell, market or distribute any product in any specified territory;
- (*G*) any Group Contract relating to the Operations, not otherwise covered by this Paragraph 7(a)(ii), (i) containing a covenant not to compete, (ii) involving sales of USD 1,000,000 (one million US dollars) in the aggregate and containing a most favoured nation provision or (iii) containing an exclusivity provision;
- (*H*) any Group Contract for the purchase of goods or services relating primarily to the Operations and not otherwise covered by this Paragraph 7(a)(ii), containing any provision requiring any Group Company to make payments in excess of USD

- 2,000,000 (two million US dollars) in the aggregate unless such Group Company purchases a required amount of such goods or services;
- (*I*) any Group Contract for the purchase, sale or lease of any real property that involves the payment by the Group or any part thereof of USD 2,000,000 (two million US dollars) or more;
- (J) any Group Contract for the purchase, licensing or development of software (other than shrinkwrap or other commercially available off-the-shelf software) that is material to the Operations as currently conducted;
- (*K*) any Group Contract relating to the Group or the Operations that involves the payment or receipt by the Group or any part thereof, of USD 2,000,000 (two million US dollars) or more on an aggregate basis in any 12-month period and is on the Signing Date not terminable without penalty and without reason upon notice of 12 (twelve) months or less; and
- (L) any other Group Contract which in the view of management of the Group is material to the Operations.
- (iii) The Data Room contains correct and complete copies as at Signing of the Material Contracts together with all material amendments thereto.
- (iv) No member of Seller's Group (other than the Business Sellers and Group Companies) is a party to any contract relating primarily to the Operations that

113

Confidential

Execution version

would be a Material Contract if a Business Seller or Group Company were a party thereto.

- (v) Except as would not reasonably be expected to have a material adverse effect on the Operations, there are no situations with respect to the Operations which involved or involves (A) the use of any corporate funds for unlawful contributions, gifts, entertainment or other unlawful expenses in each case related to political activity; (B) the making of any direct or indirect unlawful payments to government officials or others from corporate funds or the establishment or maintenance of any unlawful or unrecorded funds; (C) the violation of any of the provisions of the United States Foreign Corrupt Practices Act of 1977, or any rules or regulations promulgated thereunder; or (D) the receipt of any illegal discounts or rebates or any other violation of Competition Laws.
- (b) Agreements with connected parties
 - (i) There are no existing Contracts between, on the one hand, any Business Seller or Group Company and, on the other hand, any member of Seller's Group (excluding the Group Companies) or any shareholder of any member of Seller's Group (excluding the Group Companies) other than on normal commercial terms.
 - (ii) No Business Seller or Group Company is party to any Contract or has entered into any transaction with any current or former employee, director or officer of any such Business Seller or Group Company or any person connected (as defined by applicable Law in the relevant jurisdiction) with any of such persons, or in which any such person is interested (whether directly or indirectly), other than on normal commercial terms in the ordinary course of business.
- (c) Compliance with agreements
 - (i) All Material Contracts are valid and binding obligations of, and the terms thereof have been complied with in all material respects by:
 - (A) the relevant Group Company or Business Seller; and
 - (*B*) so far as Seller is aware as at Signing any other party to such Material Contracts.
 - (ii) No written notice of termination or of intention to terminate has been received by any member of Seller's Group in respect of any Material Contract.
- (d) Effect of Transaction

Except as disclosed in any Material Contract, so far as Seller is aware, the Transaction will not constitute a default under, result in a breach of, or give any third party a right to terminate, or result in any Encumbrance, other than a Permitted Encumbrance, under, any Material Contract.

Confidential

(e) NPG Contracts

Neither Seller nor any member of Seller's Group has assigned any of its rights under section 7.5(b) of the Master NPG Agreement (or any similar provision of any other Hive-out Agreement) to Henkel KGaA and any of its Affiliates or any other party, other than a member of Seller's Group.

10 Employees and employee benefits

- (a) Employees and terms of employment
 - (i) The information disclosed in section 7 and 14 of the Data Room contains complete and correct details, as at Signing, in relation to each Group Company and Business, of:
 - (A) the total number of Employees employed therein as at close of business on the Business Day immediately preceding the Signing Date;
 - (*B*) all collective agreements, works council agreements, social plans or trade union Contracts affecting the terms of employment of any Employee;
 - (*C*) on an anonymised basis, each Employee's employing entity, salary and other benefits, start date, period of continuous employment, location, length of notice necessary to terminate the employment if more than 3 (three) months or, if a fixed term, the expiry date of the fixed term as at close of business on the Business Day immediately preceding the Signing Date;
 - (*D*) the benefits and standard terms and conditions of employment applicable to Employees at each location at which Employees are employed or, where there are no written terms, a summary of the material terms of employment;
 - (*E*) copies of anonymised and impersonalised contracts of employment of each Senior Employee or, where there are no written terms, a summary of the material terms of employment plus copies of all other agreements with Senior Employees providing for compensation or benefits.
 - (ii) Except as set out in section 7 of the Data Room, all contracts of employment between the Seller's Group and the Senior Employees are terminable at any time by the employer on three (3) months notice or less without damages or compensation (other than compensation payable pursuant to applicable Law or under any collective agreements, works council agreements social plans or trade union Contracts).
 - (iii) No executory binding commitment has been made to vary or amend the terms of employment of any Senior Employee.
 - (iv) There are no outstanding offers of employment or engagement to any person whose annual base salary is in excess of an amount that would qualify such

115

Confidential

Execution version

person as a Senior Employee.

- (v) There are no contractual terms of employment for any Employee which expressly provide that a change of control of his or her employer (howsoever change of control may be defined, if at all) shall entitle the Employee to treat the change of control as amounting to a breach of contract or entitling him or her to any payment or benefit whatsoever or entitling him or her to treat himself or herself as redundant or dismissed or released from any obligation.
- (vi) Section 7 of the Data Room contains complete and correct, as at Signing, copies of, or a description of all of the material terms, as at Signing, of, any share incentive, share option, profit sharing, bonus or other incentive arrangements for or affecting any Senior Employees and, other than those granted in the ordinary course of business, for any other Employees.
- (vii) In relation to any Employee, or with respect to a Group Company, former employee, director, officer or consultant, there are no current or pending claims or disputes, or claims or disputes threatened in writing, whether individual or collective, or, so far as the Seller is aware, matters which may give rise to such a claim or dispute.
- (viii) There is no outstanding obligation or amount due to or in respect of any Employee in connection with or arising from his or her employment or termination of employment other than his or her normal salary or remuneration for part of the month current at the date of this Agreement.
- (ix) Other than in relation to the documents referred to in Paragraph 8(a)(i)(B), no trade union, works council, staff association or other body representing Employees is recognised by any member of Seller's Group in any way for bargaining, information or consultation purposes and no requests for recognition are pending.
- (*x*) The Seller's Group has maintained current, adequate and suitable records regarding the employment of the Employees.

- (xi) There are no Non-Automatic Transfer Employees or Automatic Transfer Employees who are entitled to a benefit under the Home Country Guarantee Plan or any other pension plan or pension program providing benefits to expatriate employees, and neither Purchaser nor its Affiliates shall assume any liabilities of Seller or any of its Affiliates under the Home Country Guarantee Plan or any other similar plan or program.
- (*xii*) There are no Non-Automatic Transfer Employees or Automatic Transfer Employees who are entitled to (A) any change in control benefits payable upon the Closing or (B) any payments under any retention plan, program or agreement.
- (b) Independent contractors

All independent contractors and consultants in each case whose basis of

116

Confidential

Execution version

remuneration is more than USD 100,000 (one hundred thousand US dollars) per year before tax engaged primarily with respect to the Operations are and have been properly classified as independent contractors under applicable Law and are not employees of any member of Seller's Group.

- (c) Termination of employment
 - (i) As at Signing, no member of Seller's Group has received any written notice of resignation from any Senior Employee.
 - (ii) Save as specifically provided for in the Accounts, no liability which remains to be discharged has been incurred by any Group Company or Business Seller for (A) breach of any contract of employment with any Employee or (B) breach of any statutory employment right.
 - (iii) As at Signing, there is no extant notice to terminate the employment of any Senior Employee.
 - (iv) Except as specifically provided for in the Accounts, in the last 12 (twelve) months no Group Company or Business Seller has made or agreed to make any payment or agreed to provide any benefit to any Employee or former employee employed by the relevant Group Company or, in relation to the Businesses, any Employee employed by any Business Seller or to any dependant of such Employee or former employee, in connection with the actual or proposed termination or suspension of employment of any such Employee or former employee.
 - (v) There is no person previously employed or engaged by any Group Company who now has or may in the future have a contractual or statutory right to return to work or a right to be re-instated or re-engaged or to any other compensation other than as required for FMLA or military leave.
- (d) Industrial disputes; compliance with Laws
 - (i) No Group Company or, in relation to the Businesses, Business Seller is involved in any strike or industrial or trade dispute or any dispute or negotiation regarding a claim of material importance with any trade union or other body representing employees or former employees of any Group Company or any Business Seller, and so far as Seller is aware, no such dispute or negotiation is pending or threatened.
 - (ii) Neither Seller nor any Affiliate of any Seller has received, with respect to any Employee: (i) written notice of any unfair labour practice charge or complaint pending or threatened before the National Labor Relations Board or any other Governmental Authority against or relating to any Employee, (ii) written notice of any complaints, grievances or arbitrations arising out of any collective bargaining agreement, similar trade union agreement or otherwise, (iii) written notice of any charge or complaint with respect to or relating to any Group Company or the Operations pending before the Equal Employment Opportunity

117

Confidential

Execution version

Commission or any other Governmental Authority responsible for the prevention of unlawful employment practices, (iv) written notice of the intent of any Governmental Authority responsible for the enforcement of labour, employment, wages and hours of work, child labour, immigration, classification of employees, or occupational safety and health laws to conduct an investigation with respect to or relating to any Employee or any site or facility at which any Employee is located or notice that such investigation is in progress, or (v) written notice of any judicial or administrative proceeding pending by or on behalf of any such present or former Employee, alleging breach of any express or implied Contract of employment, any applicable Law governing employment or the termination thereof or other discriminatory, wrongful or tortious conduct in connection with the employment relationship.

- (iii) The Group Companies and the Business Sellers are in compliance in all material respects with applicable Law and with any collective bargaining or other agreement respecting employment, employment customs and practices, codes of conduct, employee classification, labour relations, safety and health, wages, hours and terms and conditions of employment with respect to the Employees.
- (iv) No member of Seller's Group is currently engaged in any layoffs or employment terminations sufficient in number to trigger application of the Worker Adjustment and Retraining Notification Act of 1988, as amended (the "WARN Act") or any similar Law. No employee has suffered an "employment loss" (as defined in the WARN Act or any similar Law) at any site of employment or facility of any member of Seller's Group during the 90-day period ending on the Signing Date. Immediately prior to the Closing Date, Seller shall provide a correct and complete list of all employees who have suffered such an "employment loss" in the 90-day period ending on the Closing Date, and the sites of employment or facilities of each such employee.

(e) US Benefit Plans

"US Benefit Plan" means each "employee pension benefit plan" (as such term is defined in Section 3(2) of ERISA), each "employee welfare benefit plan" (as such term is defined in Section 3(1) of ERISA) and each other material employee benefit plan, program, agreement or arrangement that is maintained in or established under the laws of the United States of America by Seller or a member of Seller's Group and which provides benefits to any employee of any Group Company or any person employed primarily in the Operations by any member of Seller's Group (other than the Group Companies);

- (i) Schedule 7 (Part 5) contains a correct and complete list of each US Benefit Plan as at the date of this Agreement.
- (ii) With respect to each of the US Benefit Plans, Seller has set out in the Data Room a copy of the US Benefit Plan or a written summary of the US Benefit

118

Confidential

Execution version

Plan.

- (iii) Each US Benefit Plan has been maintained and administered in substantial compliance with its terms and applicable Law, including ERISA and the Code. Each US Benefit Plan that is intended to be a qualified plan within the meaning of section 401(a) of the Code is the subject of a favorable determination letter or may obtain such a letter by application prior to expiration of its remedial amendment period, subject to making requisite amendments.
- (iv) There are no audits, actions, suits or claims (other than routine claims for benefits) pending or, so far as Seller is aware, threatened involving the US Plans or the assets of such Benefit Plans.
- (v) Except for the plans listed in Schedule 7 (Part 4), no US Benefit Plan provides health or death benefits to or in respect of former Employees, except as specifically required by the continuation requirements of Part 6 of Title I of ERISA.
- (vi) Except for the plans listed in Schedule 7 (Part 4), if terminated on the Closing Date, no US Benefit Plan would impose any liability on any Group Company.
- (vii) With respect to any US Benefit Plan for which a separate fund of assets is or is required to be maintained, full and timely payment and contribution has been made of all amounts due (including premiums, fees and administrative expenses) and required under the terms of each such US Benefit Plan or requirements of Laws and all obligations accrued on or prior to the Closing Date which relate to employees of any Group Company and which are not yet due have either been made or have been accrued on the Working Capital Statement.
- (viii) Except for the plans listed in Schedule 7 (Part 4), none of the Business Sellers, any Group Company or any ERISA Affiliate sponsors, has sponsored, contributes to, has contributed to, or has or had an obligation to contribute to a plan subject to Title IV of ERISA, including any defined benefit plan (as defined in section 3(35) of ERISA), a multiemployer plan (as defined in section 3(37) of ERISA), or a multiple employer plan subject to section 4063 or 4064 of ERISA, a multiple employer welfare benefit arrangement (as defined in section 3(40)(A) of ERISA), or a plan subject to section 302 of ERISA or section 412 of the Code. With respect to each US Plan that is a multiemployer plan (within the meaning of section 3(37) of ERISA), no Group Company has incurred any liability that is outstanding nor does any Group Company expect to incur, prior to the Closing, any withdrawal liability with respect to any such multiemployer plan under Subtitle E of Title IV of ERISA. None of the Business Sellers, any Group Company or any ERISA Affiliate has terminated an employee benefit plan for which any Group Company could have any existing or continuing liability or obligation relating thereto.
- (ix) None of the Business Sellers nor any Group Company maintains, contributes

to or has any liability with respect to any multiemployer welfare plans other than those listed in Schedule 7 (Part 4).

- (x) Except for the Plans listed in Schedule 7 (Part 4) neither the execution of this Agreement nor the consummation of the Transaction (either alone or in combination with another event) will or can be reasonably expected to entitle any current or former director, officer, employee or consultant to any payment (including severance pay or similar compensation), any cancellation of Indebtedness, or any increase in compensation; result in the acceleration of payment, funding or vesting under any US Plan; or result in any increase in benefits payable under any US Plan.
- (xi) No amount paid or payable (whether in cash, in property, or in the form of benefits) by any Business Seller or by a Group Company in connection with the Transaction (either solely as a result thereof or as a result of such transactions in conjunction with any other event) will be an "excess parachute payment" within the meaning of section 280G of the Code, or would constitute an "excess parachute payment" if such amounts were subject to the provisions of section 280G of the Code.
- (xii) Each US Benefit Plan (including any employment agreement) that is a nonqualified deferred compensation plan within the meaning of section 409A of the Code has been administered, operated and maintained in all material respects according to the requirements of section 409A of the Code, and the terms of the written plan document of each such Plan comply with section 409A of the Code.
- (xiii) Save for the National Starch LLC Excess Benefit Plan, the individual unfunded pension agreement between Mr Davies and ICI Americas Inc., and the National Starch LLC Deferred Compensation and Survivor's Benefits Plan, no plan providing retirement benefits, or any liabilities associated with such other unfunded plans or agreements, shall be transferred to Purchaser or any of its Affiliates as a result of the transactions contemplated by this Agreement.

(f) Non US Plans

- (i) Schedule 7 (Part 3) contains a correct and complete list of all Seller Transferring Plans as at the date of this Agreement. The Data Room contains correct material details of the benefits of the Seller Transferring Plans except for arrangements to which any of the Business Seller or Group Companies contribute in compliance with any applicable Law.
- (ii) So far as the Seller is aware any favourable tax treatment of any Non-US Plan is not the subject of examination or pending cancellation by the relevant tax authority. So far as the Seller is aware there has been no act or omission on the part of the Seller or any of its Affiliates in relation to the Non-US Plans that could subject the Purchaser or any Group Company to the imposition of a penalty tax, fine, contribution notice or financial support direction deriving from

120

Confidential

Execution version

- applicable law. Each Non-US Plan in all material respects has been maintained in compliance with any minimum funding standards of applicable Law. The Seller's Group has complied in all material respects with all mandatory obligations in relation to the provisions of pension or retirement arrangements in respect of the Employees.
- (iii) Each Non-US Plan has been operating in all material respects in accordance with its provisions and in compliance with applicable Laws.
- (iv) The execution of and performance of the transactions contemplated by this Agreement, will not constitute an event under any Non-US Plan that will result in any acceleration of vesting or increase in benefits with respect to any beneficiary of such a plan.
- (v) No claims or complaints have been made are current or so far as the Seller is aware are pending or threatened in relation to any Non-US Plan including for unlawful restrictions on access to membership of or in respect of the benefits payable under any Non-US Plan to any Transferring Employee.
- (vi) So far as the Seller is aware no Transferring Employee has made any complaint or claim against the Group or the Trustees of the UK Defined Benefit Plans or any other occupational pension plan in which the Seller participates to the Pensions Advisory Service, the Pension Ombudsman or the Pensions Regulator or any equivalent governmental authority.

11 Legal compliance

(a) Licences and consents

All licences, permits, consents, authorisations, certificates and registrations ("**Permits**") from, with or to any third party, including any Governmental Authority that are required to conduct the Operations, excluding Environmental Permits (in respect of which Paragraph 10 applies), have been obtained, are in force and effect and, during the 24 (twenty-four) months immediately prior to the Signing Date, have been complied with in all material respects. Seller is not aware of any reason why any of them will be suspended or revoked.

(b) Compliance with Law

(i) Each Group Company and Business Seller is conducting the Operations in material compliance with applicable Law, in each country in which the Operations are carried on, other than Environmental Laws and anti-trust or similar legislation, in respect of which Paragraphs 10 and 11 apply respectively.

Confidential

Execution version

material adverse effect upon the Operations as at Signing.

- (iii) No Group Company or Business Seller has received any written notice during the past 24 (twenty-four) months from any Governmental Authority with respect to a violation and/or failure to comply with any applicable Law or requiring it to take or omit any action which in any case would have a material adverse effect on the Operations as at Signing.
- (c) Except as required by the Competition Laws, and the notification to and consultation of employees and unions under applicable Laws, no consent, approval, authorisation or waiver of any person or Governmental Authority is required to be obtained by any member of Seller's Group from, and no notice or filing is required to be given by any member of Seller's Group to, (a) any Governmental Authority or (b) any other person in connection with the execution, delivery or performance by Seller, the Share Seller, US Seller or any Business Seller of this Agreement.

12 Environment

- (a) For the purposes of this Paragraph 12:
 - "Environment" means any or all of the following media (alone or in combination): air; water (including water under or within land); soil and land, natural resources and any ecological systems and living organisms supported by these media;
 - "Environmental Authority" means any Governmental Authority having jurisdiction to determine any matter arising under Environmental Law and/or relating to the Environment;
 - "Environmental Law" means all Law of any relevant jurisdiction whose purpose is to protect or prevent pollution of the Environment, to address spills or releases of Hazardous Substances or to regulate emissions, discharges, or releases of Hazardous Substances into the Environment, or to regulate the use, treatment, storage, burial, disposal, transport or handling of Hazardous Substances;
 - "Environmental Permit" means any licence, permit, consent, authorisation, certificate, registration and exemption which is issued, granted or required under Environmental Law;
 - "Hazardous Substances" means any wastes, pollutants, contaminants and any other natural or artificial substance (whether in the form of a solid, liquid, gas or vapour) which is capable of causing harm or damage to the Environment or a nuisance or damage to any person and which is regulated, or for which liability or standards of care are imposed by Environmental Law; and
 - "Relevant Period" means the period since January 2, 2008.
- (b) Each Group Company and Business Seller is conducting the Operations and during the Relevant Period has conducted, the Operations in material compliance with Environmental Law.

122

Confidential

Execution version

- (c) All Environmental Permits have been obtained, are in force and have been complied with in all material respects during the Relevant Period.
- (d) No Group Company or Business Seller has received written notice during the Relevant Period that an Environmental Authority is intending to revoke or suspend any Environmental Permits as a result of a change in applicable regulation or as a result of a contravention of Environmental Law or Environmental Permits or that any amendment or modification to any Environmental Permit is required to enable the continued operation of the business activities of the Group.
- (e) No Group Company or Business Seller in respect of the Operations has received any written notice during the Relevant Period of any civil, criminal, regulatory or administrative action, claim, investigation or other proceeding or suit relating to a contravention of Environmental Law or Environmental Permit or with respect to a spill or release of any Hazardous Substances, and, so far as Seller is aware, there are no facts or conditions that would reasonably be expected to give rise to any such action, claim, investigation, or other proceeding or suit.

(f) In the period of 5 (five) years immediately preceding the Signing Date, there has been no spill, release, disposal or discharge of Hazardous Substances (i) on, to or from any Business Property or any property formerly owned, leased or used in the Operations or the operation of the business activities of the Group during such ownership, lease or use or (ii) at any third party property to which wastes generated by the operation of the business activities of the Group were sent or came to be present, in quantity or condition that would reasonably be expected to result in a material Liability to any Group Company or Business Seller in respect of the Operations under Environmental Laws.

13 Anti-competitive agreements and practices

- (a) No Group Company and, in relation to the Businesses, no Business Seller is a party to any agreement, activity, practice or arrangement which is the subject of written notice received by any member of Seller's Group from any relevant regulatory authority that it (i) is unenforceable or void or (ii) renders the Group Company or Business Seller that is party thereto liable to civil, criminal or administrative proceedings, in either case by virtue of any anti-trust or similar legislation in any jurisdiction in which any Group Company carries on business or any Business Seller carries on the Business.
- (b) No Group Company and, in relation to the Businesses, no Business Seller is a party to any agreement, activity, practice or arrangement which in whole or in part contravenes (i) Articles 101(1) or 102 of the Treaty on the Functioning of the European Union, (ii) any United States federal or state Competition Laws, including the Sherman Act, or (iii) any other anti-trust, anti-monopoly or anti-cartel legislation or regulations or any other similar applicable statute, order or regulation in any other jurisdiction, including the Netherlands Competition Act ('Mededingingswet').

123

Confidential

Execution version

14 Litigation

- (a) No Group Company and, in relation to the Businesses, no Business Seller is involved whether as claimant or defendant or other party in any claim, proceeding, litigation, prosecution, investigation, hearing, mediation, enquiry or arbitration, whether civil, criminal or administrative (other than as claimant in the collection of debts arising in the ordinary course of its business), in each case involving a claim for damages in excess of USD 1,000,000 (one million US dollars) or requiring equitable relief that could have a Group Material Adverse Effect.
- (b) So far as Seller is aware, no such claim, proceeding, litigation, prosecution, investigation, hearing, mediation, enquiry or arbitration related to or in connection with the Operations is threatened against any Group Company or Business Seller.
- (c) There is no outstanding, unfulfilled order, judgment or injunction of any Governmental Authority of competent jurisdiction adversely affecting the Group or Operations with a financial impact which could reasonably be expected to exceed the amount referred to in Paragraph 12(a).

15 Insurance

- (a) The insurances of the Group Companies or Business Sellers, as at Signing, are summarised in section 4 of the Data Room.
- (b) In respect of the insurances referred to in Paragraph 13(a), all premiums have been duly paid to date and no member of Seller's Group has received any notification that such insurances are not valid or enforceable.
- (c) All the insurances referred to in Paragraph 13(a), or reasonable renewals or replacements thereof, are in full force and effect.
- (d) Details, as at Signing, of all outstanding insurance claims in excess of USD 1,000,000 (one million US dollars) and of all insurance claims in excess of USD 1,000,000 (one million US dollars) made during the past 5 (five) years preceding the Signing Date have been disclosed in section [•] of the Data Room.

16 Product Claims; Warranties

- (a) During the three-year period prior to the Signing Date, no individual product-related claims of more than USD 1,000,000 (one million US dollars) and in aggregate not more than USD 5,000,000 (five million US dollars) have been made against any Group Company or, in relation to the Businesses, any Business Seller in respect of any products which have been manufactured or sold by any Group Company or Business Seller.
- (b) Except for warranties customary in the industry (such as but not limited to fitness for intended use, compliance with applicable Laws, shelf life and absence of defects), no member of Seller's Group has given or made any express or implied warranties to third parties with respect to any products sold or services performed by any Group Company or, in relation to the Businesses, any Business Seller and

Confidential

Seller is not aware of any fact or of the occurrence of any event forming the basis of any present or future claim against any member of Seller's Group not covered by insurance, for liability in connection with the sale of goods or performance of services by any member of Seller's Group or on account of any express or implied warranty in connection therewith.

17 Tax

- (a) All Tax Returns that are required to be filed by or with respect to any Group Company, or any Business, Business Asset or Business Liability, have been timely filed, and all such Tax Returns are correct and complete in all material respects.
- (b) All Taxes shown to be due on the Tax Returns referred to in Paragraph 15(a) above have been timely paid in full and all material withholding Taxes required to be withheld by any Group Company, or with respect to any Business, have been withheld (including from employees, customers and any other applicable payees) and timely paid to the appropriate Governmental Authority for all periods in compliance with all Tax withholding provisions of applicable Law.
- (c) No issues that have been raised by the relevant Tax Authority in connection with the examination of any of the Tax Returns referred to in Paragraph 15(a) above are currently pending.
- (d) No waivers of statutes of limitation have been given by or requested with respect to any Taxes of any Group Company, any Business, Business Asset or Business Liability.
- (e) To the best of Seller's knowledge, no Group Company will be required, as a result of (i) a change in accounting method made effective prior to the Closing or (ii) any agreement between a Group Company and a Tax Authority made prior to the Closing, to include any item of income in, or to exclude any item of deduction from, taxable income for any Tax period beginning on or after the Closing Date.
- (f) There are no Encumbrances on any of the assets of any Group Company or any Business Asset that arose in connection with any failure (or alleged failure) to pay any Tax.
- (g) To the best of Seller's knowledge, no closing agreements, private letter rulings, technical advice memoranda or similar agreement or rulings have been entered into or issued by any Tax authority with respect to any Group Company.
- (h) No Group Company has participated in any "reportable transaction" within the meaning of U.S. Treasury Regulations section 1.6011-4 or in any other transaction for which specific disclosure to a Tax Authority is required under a similar provision of Law, and no Group Company has provided material aid, assistance or advice with respect to any such transaction (including acting as a "material advisor" to or as a promoter in respect of any such transaction within the meaning of section 6111 of the Code) for which a Group Company was or will be required to report such aid, assistance or advice to a Tax Authority.

125

Confidential

Execution version

- (i) No Group Company has been a distributing corporation or a controlled corporation within the meaning of section 355 of the Code in a transaction intended to be governed by section 355 of the Code.
- (j) No transaction contemplated by this Agreement is subject to withholding under section 1445 of the Code (relating to "FIRPTA").
- (k) All Tax Sharing Arrangements and Tax indemnity arrangements relating to any Group Company (other than this Agreement) will terminate prior to the Closing date, and no Group Company will have any Liability thereunder on or after the Closing Date.
- (l) To the best of Seller's knowledge, any document in the possession of any Group Company or to the production of which any Group Company is entitled and which was executed on or before the Closing Date and which attracts stamp duty or any other documentary or registration Tax and which document may be necessary for such Group Company in proving title of such Group Company to any asset or to produce in evidence for the purposes of any litigation or on which to rely in relation to any matter relating to Tax has been properly stamped or the relevant documentary or registration Tax has been duly paid (as the case may be).
- (m) Each Group Company has maintained accurate and up to date records, invoices, accounts and other documents required by or necessary for the purposes of applicable VAT Law.
- (n) For Tax purposes, each of the Group Companies is and has been resident only in the jurisdiction in which it is incorporated and does not have nor has it had a permanent establishment or (permanent) representative or other taxable presence in any jurisdiction other than that in which it is resident for Tax purposes. None of the Group Companies constitutes or has constituted a permanent establishment or is or has been a (permanent) representative of another person.

18 Customers

Section 18.1.20 of the Data Room contains an anonymised list, as at Signing, of the top 20 (twenty) industrial customers of the Group, and an indication of net sales by the Group attributable to each such customer, for the 2009 fiscal year, and Section 18.1.21 of the Data Room contains an anonymised list of the top 32 (thirty-two) food customers of the Group, and an indication of net sales by the Group attributable to each such customer, for the 2009 fiscal year.

So far as Seller is aware, no customer referenced in Section 18.1.20 or Section 18.1.21 of the Data Room will cease to do business with the Group as a consequence of the consummation of the Transaction on substantially the same terms and at substantially the same levels (considered on an annualised basis) as previously conducted with the Group.

19 Events since the Accounts Date

(a) Between the Accounts Date and Signing, there has been no material adverse

126

Confidential

Execution version

change in the financial position or operations of the Group, other than a change affecting or likely to affect all companies carrying on similar business in the countries in which the Group carries on business.

- (b) Since the Accounts Date the business of the Group has been carried on as a going concern in the ordinary course without any material interruption or material alteration in its nature, scope or manner.
- (c) Between March 31, 2010 and Signing, the Group Companies have not declared, authorized, made or paid any dividend, distribution or other similar payment in excess of USD 1,000,000 (one million US dollars) to any of its members, shareholders or partners, other than in the ordinary course of trading activities consistent with past practice.

20 Bank Accounts

Before Closing, a correct and compete list of all bank accounts and safe deposit boxes of the Group Companies and persons authorized to sign or otherwise act with respect thereto and a correct and complete of all persons holding a general or special power of attorney granted by the Group Companies will be provided to the Purchaser.

21 Hedging

In relation to arrangements for the sale of products to customers the Group Companies and, in relation to the Operations, the Business Sellers have engaged in corn hedging transactions which comply in all material respects with the relevant hedging policy set out in section 2.9.1 of the Data Room. None of the Group Companies and, in relation to the Operations, the Business Sellers have entered into any hedging contracts (other than foreign exchange hedging) in effect as at Signing other than in accordance with the relevant hedging policy set out in section 2.9.1 of the Data Room.

22 Related and Other Transactions

Sections 1.2, 11, 13.1 and 13.11 of the Data Room and the term sheets attached to the Transitional Services Agreement sets forth (i) a description of (A) all services provided by any Group Company to Seller or any Affiliate thereof (other than a Group Company) or (B) any use by Seller or any Affiliate thereof (other than a Group Company) of either (x) any assets or properties of any Group Company or (y) any employees of any Group Company for any purpose, and the manner in which the Group Company has been compensated for the costs of providing such services or use, and (ii) a description of (A) all services provided by Seller or any Affiliate of Seller (other than a Group Company) to any Group Company or (B) any use by any Group Company of either (x) any assets or properties of Seller or any Affiliate thereof (other than a Group Company) or (y) any employees of Seller or any Affiliate thereof (other than a Group Company), and the manner in which the Group Company has compensated the Seller or such Affiliate for the costs of providing such services or use.

127

Confidential

Execution version

23 Finders' Fees

Except for any fees that will be paid by Seller, there is no investment banker, broker, finder or other intermediary that has been retained by or is authorised to act on behalf of Seller or any other member of Seller's Group that might be entitled to any fee or commission from Seller, the Company, or after the Effective Time, by Purchaser, any Share Purchaser or any Business Purchaser in connection with the transactions contemplated by this Agreement.

24 Disclosure of information

The following information has been collated by Seller in good faith and Seller has not knowingly included any matter which is untrue or knowingly omitted any matter the omission of which would make the contents untrue or inaccurate in any material respects:

(a) all disclosures made pursuant to Paragraph 2 of the Disclosure Letter;

- (b) all information made available in section 14, 15 and 18 of the Data Room and in the answers given to questions raised by Purchaser during its due diligence review; and
- all written information provided during the interviews and question-and-answer sessions with management, site visits and expert meetings.

25 Data Room materials

On the Signing Date, Seller delivered to Purchaser an electronic copy of all of the materials contained in the Data Room as of the Signing Date.

128

Confidential

Execution version

Schedule 15 (Part 4) Warranties of Purchaser

(Clause 9.4.1)

- 1 Authority and capacity
 - (a) Incorporation

Each of Purchaser and the purchasers nominated in terms of the provisions of Clause 2.3 is or will be validly existing and is a company duly incorporated under the Law of its jurisdiction of incorporation.

- (b) Authority to enter into Agreement
 - (i) Each of Purchaser and the purchasers nominated in terms of the provisions of Clause 2.3 has the legal right and full power and authority to enter into and perform this Agreement and any documents to be executed by it pursuant to or in connection with this Agreement.
 - (ii) The documents referred to in Paragraph 1(b)(i) will, when executed, constitute valid and binding obligations on each of Purchaser and the purchasers nominated in terms of the provisions of Clause 2.3 in accordance with their respective terms.
- (c) Authorisation

Each of the Purchaser and the purchasers nominated in terms of the provisions of Clause 2.3 has taken or will have taken by Closing all corporate action required by it to authorise it to enter into and to perform this Agreement and any documents to be executed by it pursuant to or in connection with this Agreement.

2 Financing

At the time set for Closing pursuant to Clause 6.1.1 the Purchaser will be able to pay the Estimated Purchase Price from its existing banking facilities and available cash.

129

Confidential

Execution version

Schedule 18 Conduct of Business

Schedule 18 (Part 1)

Without limiting the generality of Clause 5.1, Seller undertakes to procure that between Signing and Closing the relevant members of the Seller's Group, and between the Closing Date and any Deferred Closing, the applicable Non-Material Business and/or Non-Material Group Company, shall, except to the extent required to effect the transactions contemplated by this Agreement or approved in writing by Purchaser, which approval shall not be unreasonably withheld, conditioned or delayed:

- 2.1 maintain in effect full coverage under the Insurance Policies (including all insurance coverage by Seller on behalf of any member of the Seller's Group, whether funded by a third party or Seller) for events, occurrences or accidents occurring prior to the Closing to the extent that same relate to any Group Company, any Business Seller, the Properties or the Operations;
- 2.2 use reasonable best efforts to preserve intact the business of the Group and preserve the goodwill of the Group's officers, employees, suppliers, distributors, customers and other third parties having business relations with it;

- **2.3** maintain and keep the Business Assets and the assets of the Group Companies in at least as good repair, working order and condition as is sufficient for operation in the ordinary course of business;
- **2.4** perform in all material respects all of their obligations under each Group Contract;
- 2.5 maintain the books of account and records of the Group in the usual and regular manner consistent with past practice;
- **2.6** comply in all material respects with all Laws applicable to the Operations and the Group;
- 2.7 not make any commitments (contractual or otherwise), expenditures or investments (other than (a) purchases in the ordinary course of business and (b) capital expenditures, including any expenditures in connection with Project Europe, in excess of USD 500,000 (five hundred thousand US dollars) individually and USD 2,500,000 (two million five hundred thousand US dollars) in aggregate;
- 2.8 not enter into or amend any Group Contract with any member of Seller's Group (other than any Group Company) other than in the ordinary course of trading activities consistent with past practice;
- 2.9 promptly advise Purchaser in writing of any governmental or any material other third party complaints, investigations or hearings (or communications indicating that the same may be contemplated) with respect to any Business or any Group Company;
- **2.10** not accelerate or delay collection of any notes or accounts receivable generated by the Operations in advance of or beyond their regular due dates or the dates when the same would have been collected in the ordinary course of the Operations consistent with past practice;

130

Confidential

Execution version

- **2.11** not delay or accelerate payment of any account payable or other Liability beyond or in advance of its due date or the date when such Liability would have been paid in the ordinary course of the Operations consistent with past practice;
- **2.12** file all Tax Returns of the Group Companies and each Business Seller and pay all Taxes of each Group Company and Business Seller as and when due as required under applicable Law;
- 2.13 not change any accounting methods or practices of the Group, except to the extent required by applicable accounting standards;
- **2.14** not change any Tax accounting method of, or make any Tax election (except as permitted by this Agreement) for, or settle or compromise any Tax liability of, any Group Company or Business Seller;
- 2.15 not create, issue or allot, or allow to be created, issued or allotted, any share capital of any Group Company;
- **2.16** not declare, authorize, make or pay any dividends, distributions or other similar payments in excess of USD 1,000,000 (one million US dollars) to any of its members, shareholders or partners, other than in the ordinary course of trading activities consistent with past practice;
- 2.17 not repay, redeem or repurchase, or allow to be repaid, redeemed or repurchased, any share capital of any Group Company;
- **2.18** continue to perform or cause to be performed any environmental remedial or compliance projects or employee health or safety programs affecting the Operations that were in progress at Signing, subject to reasonable modifications to such projects and programs;
- **2.19** not sell, lease (as lessor or lessee), sublease, transfer or otherwise dispose of, or mortgage or pledge, or impose any Encumbrance on, any of the Properties, except for Permitted Encumbrances;
- **2.20** not purchase any real property or enter into any lease with respect to any real property with a lease period longer than 12 (twelve) months;
- **2.21** not enter into any Contract that would constitute a Material Contract or amend, terminate or otherwise modify any Material Contract in any material respect;
- 2.22 not increase the salary or wages payable or to become payable to its directors, executive officers or employees, except in the ordinary course of business consistent with past practice; enter into any employment, change in control or severance arrangement with any director, executive officer or employee, except in the case of employees (other than executive officers) in the ordinary course of business consistent with past practice; establish, adopt, enter into or amend in any material respect any employee benefit plan; or enter into or amend any collective bargaining or similar agreement;
- **2.23** not enter into any share incentive, share option, profit sharing, bonus or other incentive arrangements for or affecting any Senior Employees or, other than those granted in the ordinary course of business, for any other Employees;
- 2.24 protect and defend the validity and enforceability of the Acquired Intellectual Property in the

Execution version

- ordinary course of business;
- **2.25** continue to pay the maintenance fees when due and make all necessary filings and respond to all official actions in respect of the Acquired Registered Intellectual Property;
- **2.26** not adopt or enter into a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization involving any Group Company or Business Seller;
- 2.27 not sell, assign, pledge, hypothecate or otherwise transfer any asset or property used or held for use in connection with the Group over a value of USD 2,000,000 (two million US dollars), other than sales of inventory in the ordinary course of business;
- 2.28 not enter into any Contract that requires any third-party consent in connection with the transactions contemplated by this Agreement;
- **2.29** not amend the charter, bylaws or similar organizational documents of any Group Company;
- **2.30** not allow the levels of raw materials, supplies, work-in-process, finished goods or other materials included in the inventory of the Group Companies and the Business Sellers with respect to the Operations to vary in any material respect from the levels customarily maintained by such entities:
- **2.31** not create, incur or assume, or agree to create, incur or assume, any indebtedness for borrowed money or enter into, as lessee, any capitalized lease obligation;
- 2.32 not enter into any fixed price arrangement for the sale of products to customers with a term of more than 12 (twelve) months without engaging in any hedging transactions which comply in all material respects with the relevant hedging policy set out in section 2.9.1 of the Data Room;
- **2.33** not enter into any corn hedging contract other than in accordance with the relevant hedging policy set out in section 2.9.1 of the Data Room;
- 2.34 not settle or agree to settle any legal action, suit, arbitration or proceeding which settlements, in the aggregate, would be in excess of USD 1,000,000 (one million US dollars) or which would impose any material injunctive, equitable or other non-monetary relief or remedy which in any case impacts the Operations in any material respect; and
- 2.35 not enter into any agreement to do any of the foregoing.



June 19, 2010

Corn Products International, Inc.
Senior Bridge Credit Facility
Commitment Letter

Corn Products International, Inc. 5 Westbrook Corporate Center Westchester, Illinois 60154 Attention: Cheryl K. Beebe, Chief Financial Officer

Ladies and Gentlemen:

You (the "Borrower") have requested that J.P. Morgan Securities Inc. ("JPMorgan") agree to structure, arrange and syndicate a \$1,350,000,000 senior term loan bridge facility (the "Facility"), and that JPMorgan Chase Bank, National Association ("JPMCB") commit to provide the entire principal amount of the Facility and to serve as administrative agent for the Facility. You have requested the Facility to finance in part the proposed acquisition (the "Acquisition") by you and/or one or more of your subsidiaries of the specialty starch business (the "Business") of Azko Nobel N.V. (the "Seller"), which Business is comprised of certain assets of the Seller and its subsidiaries (including equity interests in certain of the Seller's subsidiaries), for an aggregate purchase price of approximately \$1,300,000,000. You intend to consummate the Acquisition pursuant to the terms of an international share and business sale agreement (the "Acquisition Agreement") to be entered into between and among the Borrower and the Seller.

JPMorgan is pleased to advise you that it is willing to act as exclusive arranger for the Facility.

Furthermore, each of JPMCB and JPMorgan, respectively, are pleased to advise you of JPMCB's commitment to provide the entire principal amount of the Facility upon the terms and subject to the conditions set forth or referred to in this commitment letter (the "Commitment Letter") and in the Summary of Terms and Conditions attached hereto as Exhibit A (the "Term Sheet"). The commitments of JPMCB in respect of the Facility shall be reduced as provided in Exhibit A.

It is agreed that JPMCB will act as the sole and exclusive Administrative Agent, and that JPMorgan will act as the sole and exclusive Lead Arranger and Bookrunner (in such capacities, the "Lead Arranger") for the Facility. You agree that no other agents, co-agents or arrangers will be appointed, no other titles will be awarded and no compensation (other than that expressly contemplated by the Term Sheet and the Fee Letter referred to below) will be paid in connection with the Facility unless you and we shall so agree.

We intend to syndicate the Facility to a group of financial institutions (together with JPMCB, the "Lenders") identified by us in consultation with you. The financial institutions identified and selected to act as the Lenders shall be subject to your prior consent (not to be unreasonably withheld). JPMorgan

intends to commence syndication efforts promptly following the public announcement of the Acquisition, and you agree actively to assist JPMorgan (and to use your commercially reasonable efforts consistent with the Acquisition Agreement to cause Seller to cooperate with JPMorgan) in completing a syndication satisfactory to it. Such assistance shall include (a) your using commercially reasonable efforts to ensure that the syndication efforts benefit materially from your existing lending relationships, (b) direct contact between senior management and advisors of the Borrower, the Business and the proposed Lenders, (c) the hosting, with JPMorgan, of one or more meetings of prospective Lenders and (d) as set forth in the next paragraph, assistance in the preparation of materials to be used in connection with the syndication (collectively with the Term Sheet, the "Information Materials").

You will assist us in preparing Information Materials, including but not limited to a Confidential Information Memorandum or lender slides, for distribution to prospective Lenders. If requested, you also will assist us (and will use your commercially reasonable efforts consistent with the Acquisition Agreement to cause Seller to cooperate with us) in preparing an additional version of the Information Materials (the "Public-Side Version") to be used by prospective Lenders' public-side employees and representatives ("Public-Siders") who do not wish to receive material non-public information (within the meaning of United States federal securities laws) with respect to the Borrower, the Seller, their respective affiliates and any of their respective securities ("MNPI") and who may be engaged in investment and other market related activities with respect to any such entity's securities or loans. Before distribution of any Information Materials, you agree to execute and deliver to us (i) a letter in which you authorize distribution of the Information Materials to a prospective Lender's employees willing to receive MNPI ("Private-Siders") and (ii) a separate letter in which you authorize distribution of the Public-Side Version to Public-Siders and represent that no MNPI is contained therein. You also acknowledge that JPMorgan Public-Siders consisting of publishing debt analysts may participate in any meeting or telephone conference calls held pursuant to clause (c) of the immediately previous paragraph; provided that such analysts shall not publish any information obtained from such meetings or calls (i) prior to the completion of a Successful Syndication or (ii) in violation of any confidentiality agreement between you and JPMorgan or JPMBC.

Without limiting your obligations to assist with syndication efforts as set forth in this letter, notwithstanding anything to the contrary contained in this Commitment Letter or the Fee Letter or any other letter agreement or undertaking concerning the financing of the Acquisition to which JPMCB is a party, JPMCB agrees that the completion of a Successful Syndication is not a condition to its commitment hereunder or the funding of the Facility on the Closing Date and JPMCB further agrees that (except for purposes of determining whether a Successful Syndication has been achieved under the applicable provisions of the Fee Letter) JPMCB will not be released from its commitment hereunder in connection with such a syndication to any Lender unless (a) any such Lender (reasonably acceptable to you) has entered into an amendment or joinder with respect to this Commitment Letter committing to provide a portion of the Facility (in which case our commitments hereunder shall be reduced at such time by an amount equal to the commitment assumed by such Lender) or (b) such Lender shall have entered into the Credit Documentation and funded the portion of the Facility required to be funded by it on the Closing Date.

The Borrower agrees that the following documents may be distributed to both Private-Siders and Public-Siders, unless the Borrower advises JPMorgan in writing (including by email) within a reasonable time prior to their intended distribution that such materials should only be distributed to

Private-Siders: (a) administrative materials prepared by JPMorgan for prospective Lenders (such as a lender meeting invitation, lender allocation, if any, and funding and closing memoranda), (b) notification of changes in the Facility's terms and (c) other materials intended for prospective Lenders after the initial distribution of Information Materials. The Lead Arranger will provide the foregoing (other than any of the foregoing

2

that constitutes administrative or other materials that would not customarily be reviewed by a borrower prior to being distributed to lenders in similar circumstances) to you and will give you a reasonable amount of time to review and revise such materials prior to such distribution to determine whether such documents contain MNPI and to give JPMorgan the written notice referred to above. If you advise us that any of the foregoing should be distributed only to Private-Siders, then Public-Siders will not receive such materials without further discussions with you.

The Borrower hereby authorizes JPMorgan to distribute drafts of definitive documentation with respect to the Facility to Private-Siders and Public-Siders.

As the Lead Arranger, JPMorgan will manage all aspects of the syndication, including decisions as to the selection of institutions to be approached and when they will be approached, when their commitments will be accepted, which institutions will participate (subject to the Borrower's consent right described above), the allocations of the commitments among the Lenders and the amount and distribution of fees among the Lenders and the Lead Arranger will notify you regarding such final determinations. In acting as the Lead Arranger, JPMorgan will have no responsibility other than to arrange the syndication as set forth herein and shall in no event be subject to any fiduciary or other implied duties. Additionally, the Borrower acknowledges and agrees that, as Lead Arranger, JPMorgan is not advising the Borrower as to any legal, tax, investment, accounting or regulatory matters in any jurisdiction. The Borrower shall consult with its own advisors concerning such matters and shall be responsible for making its own independent investigation and appraisal of the transactions contemplated hereby, and JPMorgan shall have no responsibility or liability to the Borrower with respect thereto. Any review by JPMorgan of the Borrower, the transactions contemplated hereby or other matters relating to such transactions will be performed solely for the benefit of JPMorgan and JPMCB and shall not be on behalf of the Borrower.

To assist JPMorgan in its syndication efforts, you agree promptly to prepare and provide to JPMorgan and JPMCB all information with respect to the Borrower, the Acquisition, the Business (to the extent available to the Borrower under the Acquisition Agreement) and the transactions contemplated hereby, including all financial information and projections (the "Projections"), as we may reasonably request in connection with the arrangement and syndication of the Facility. You hereby represent and covenant that (a) all written information, other than the Projections, estimates and forward looking statements and information of a general economic or industry public nature with respect to the Borrower, its subsidiaries and the Business, that has been or will be furnished to JPMCB or JPMorgan by you or any of your representatives (the "Information") is or will be, when furnished, taken as a whole, complete and correct in all material respects and does not or will not, when furnished, contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements contained therein not materially misleading in light of the circumstances under which such statements are made and (b) the Projections, estimates, forward looking statements and information of a general economic or industry public nature that have been or will be made available to JPMCB or JPMorgan by you or any of your representatives have been or will be prepared in good faith based upon, as applicable, reasonable assumptions that are believed by the preparer thereof to be reasonable at the time such Projections or other materials are furnished to the Commitment Parties (it being recognized by the Commitment Parties that such Projections and other forward looking information are not a guarantee of financial performance, are not to be viewed as facts and that actual results during the period or periods covered by any such Projections and other forward looking information may differ from the projected results, and such differences may be material). You agree that if at any time prior to the date of consummation of the Acquisition (the "Closing Date") and, if requested by us, for such period thereafter as is necessary to complete a Successful Syndication any of the representations in the preceding sentence would be incorrect in any material respect if the Information and Projections were being furnished, and such

3

representations were being made, at such time, then you will promptly supplement, or cause to be supplemented, the Information and Projections so that such representations will be correct at such time.

As consideration for JPMCB's commitment hereunder and JPMorgan's agreement to perform the services described herein, you agree to pay to JPMCB the nonrefundable fees set forth in Annex I to the Term Sheet and in the Fee Letter dated the date hereof and delivered herewith (the "Fee Letter").

JPMCB's commitment hereunder and JPMorgan's agreement to perform the services described herein are subject only to (a) there not having occurred any "Group Material Adverse Effect" (as defined in the Acquisition Agreement), (b) our satisfaction that prior to and during the syndication of the Facility, if any, there shall be no competing offering, placement or arrangement of any debt securities or bank financing by or on behalf of the Borrower or any affiliate thereof other than (i) any of the foregoing the proceeds of which result in a reduction of the commitments made hereunder or will be required to be used to prepay the Facility, (ii) that certain \$750,000,000 senior revolving credit facility to be arranged by JPMorgan pursuant to the terms of that certain commitment letter dated the date hereof among JPMorgan, JPMCB and the Borrower or (iii) the incurrence of debt in the ordinary course of business for normal working capital needs of the Borrower and its affiliates, (c) the negotiation, execution and delivery on or before December 18, 2010 of loan documentation for the Facility satisfactory to JPMCB and its counsel, (d) the other conditions set forth or referred to in the Term Sheet and (e) your compliance with your obligations hereunder (including the Term Sheet) and under the Fee Letter.

Notwithstanding anything in this Commitment Letter (including each of the annexes attached hereto), the Fee Letter, the Credit Documentation or any other letter agreement or other undertaking concerning the financing of the Acquisition to which either Commitment Party is party to the contrary, the only representations relating to the Borrower, its subsidiaries, and their respective businesses, or the Business, the accuracy of which shall be a condition to availability of the Facility on the Closing Date, shall be (i) those warranties made by the Seller in the Acquisition Agreement as are material to the interests of the Lenders, but only to the extent that the Borrower has the right to terminate its obligations under the Acquisition Agreement as a result of a breach of such warranties in the Acquisition Agreement (the "Acquisition Agreement Warranties") and (ii) the below-defined Specified Representations. For purposes hereof, "Specified Representations" means the representations and warranties referred to in the Term Sheet relating to corporate existence, power and authority, due authorization, execution and delivery and the enforceability of the Credit Documentation, the non-contravention of the Facility with applicable law or any order or judgment, financial statements, Federal Reserve margin regulations and Investment Company Act. In addition, the terms of the Credit Documentation shall be in a form such that they do not impair availability of the Facility on the Closing Date if the conditions set forth in (y) the immediately preceding paragraph of this Commitment Letter, and (z) the Term Sheet are satisfied. This paragraph, and the provisions of this paragraph, shall be referred to as the "Conditionality Provision."

You agree (a) to indemnify and hold harmless JPMCB, JPMorgan and their affiliates and their respective officers, directors, employees, advisors, and agents (each, an "<u>indemnified person</u>") from and against any and all losses, claims, damages and liabilities to which any such indemnified person may become subject arising out of or in connection with this Commitment Letter, the Facility, the Acquisition, the use of the proceeds thereof or any related transaction or any claim, litigation, investigation or proceeding relating to any of the foregoing, regardless of whether (i) any indemnified person is a party thereto and (ii) the transactions contemplated hereunder, including without limitation, the Acquisition, have been consummated, and to reimburse each indemnified person upon demand for any legal or other expenses incurred in connection with investigating or defending any of the foregoing, <u>provided</u> that the foregoing indemnity will not, as to any indemnified person, apply to losses, claims, damages, liabilities or related expenses to the extent they are found by a final, non-appealable judgment

4

of a court to arise from the willful misconduct, bad faith or gross negligence of such indemnified person, and (b) to reimburse JPMCB, JPMorgan and their affiliates within five business days after demand for all documented reasonable out-of-pocket expenses (including due diligence expenses, syndication expenses, consultant's fees and expenses, travel expenses, and reasonable fees, charges and disbursements of counsel) incurred in connection with the Facility and any related documentation (including this Commitment Letter, the Term Sheet, the Fee Letter and the definitive financing documentation) or the administration, amendment, modification or waiver thereof. No indemnified person shall be liable for any damages arising from the use by others of Information or other materials obtained through electronic, telecommunications or other information transmission systems or for any special, indirect, consequential or punitive damages in connection with the Facility, except to the extent such damage arises from the gross negligence or willful misconduct of such person.

This Commitment Letter shall not be assignable by you without the prior written consent of JPMCB and JPMorgan (and any purported assignment without such consent shall be null and void), is intended to be solely for the benefit of the parties hereto and is not intended to confer any benefits upon, or create any rights in favor of, any person other than the parties hereto and the indemnified persons. This Commitment Letter may not be amended or waived except by an instrument in writing signed by you, JPMCB and JPMorgan. This Commitment Letter may be executed in any number of counterparts, each of which shall be an original, and all of which, when taken together, shall constitute one agreement. Delivery of an executed signature page of this Commitment Letter by facsimile transmission or pdf electronic transmission shall be effective as delivery of a manually executed counterpart hereof. This Commitment Letter and the Fee Letter are the only agreements that have been entered into among us with respect to the Facility and set forth the entire understanding of the parties with respect thereto.

This Commitment Letter shall be governed by, and construed in accordance with, the law of the State of New York. Each party hereto consents to the nonexclusive jurisdiction and venue of the state or federal courts located in the City of New York. Each party hereto irrevocably waives, to the fullest extent permitted by applicable law, (a) any right it may have to a trial by jury in any legal proceeding arising out of or relating to this Commitment Letter, the Fee Letter, the Acquisition or the transactions contemplated hereby or thereby (whether based on contract, tort or any other theory) and (b) any objection that it may now or hereafter have to the laying of venue of any such legal proceeding in the state or federal courts located in the City of New York.

This Commitment Letter is delivered to you on the understanding that neither this Commitment Letter, the Term Sheet or the Fee Letter nor any of their terms or substance shall be disclosed, directly or indirectly, to any other person except (a) to your officers, agents, directors, employees, affiliates, accountants and advisors who are directly involved in the consideration of this matter and have been instructed to maintain the confidentiality thereof and for whom you shall be responsible for any breach by any one of them of this confidentiality undertaking, (b) as may be compelled in a judicial or administrative proceeding or as otherwise required by law (in which case you agree to inform us promptly thereof), (c) to S&P and Moody's in connection with the Facility (provided, however, that in no event shall the Fee Letter be delivered to S&P or Moody's), (d) upon notice to the Lead Arranger, this Commitment Letter and the existence and contents hereof (but not the Fee Letter or the contents thereof other than the existence thereof and the contents thereof as part of projections, pro forma financial information and a generic disclosure of aggregate sources and uses to the extent customary in marketing materials or other disclosures) may be disclosed in any syndication or other marketing materials relating to the Facility or in connection with any public filing requirement or (e) the Term Sheet may be disclosed to potential Lenders in connection with the Acquisition. Notwithstanding the foregoing, following its acceptance by the Borrower (and the payment of the fees required by the Fee Letter to be paid upon such acceptance), the Commitment Letter and the Term Sheet (but not the Fee Letter) may be disclosed to the

5

Seller and its officers, directors, agents, employees, affiliates, attorneys, accountants, and advisors who are directly involved in the Acquisition and have been instructed to maintain the confidentiality thereof.

You acknowledge that JPMorgan, JPMCB and their affiliates may be providing debt financing, equity capital or other services (including financial advisory services) to other companies in respect of which you may have conflicting interests regarding the transactions described herein and otherwise. Neither JPMorgan nor JPMCB will use confidential information obtained from you by virtue of the transactions contemplated by this letter or their other relationships with you in connection with the performance by JPMorgan or JPMCB of services for other companies, and neither JPMorgan nor JPMCB will furnish any such information to other companies. You also acknowledge that JPMorgan and JPMCB have no obligation to use in connection with the transactions contemplated by this letter, or to furnish to you, confidential information obtained from other companies. You further acknowledge that JPMorgan is a full service securities firm and JPMorgan may from time to time effect transactions, for its own or its affiliates' account or the account of customers, and hold positions in loans, securities or options on loans or securities of the Borrower and its affiliates and of other companies that may be the subject of the transactions contemplated by this Commitment Letter.

Each of JPMorgan and JPMCB may employ the services of its affiliates in providing certain services hereunder and, in connection with the provision of such services, may exchange with such affiliates information concerning you and the other companies that may be the subject of the transactions contemplated by this Commitment Letter, and, to the extent so employed, such affiliates shall be entitled to the benefits afforded JPMorgan or JPMCB, as applicable, hereunder.

JPMCB and JPMorgan hereby notify you that, pursuant to the requirements of the USA Patriot Act, Title III of Pub. L. 107-56 (signed into law on October 26, 2001) (the "Patriot Act"), it is required to obtain, verify and record information that identifies the Borrower, which information includes names and addresses and other information that will allow JPMCB and JPMorgan to identify the Borrower in accordance with the Patriot Act.

The compensation, reimbursement, indemnification and confidentiality provisions contained herein and in the Fee Letter shall remain in full force and effect regardless of whether definitive financing documentation shall be executed and delivered and notwithstanding the termination of this Commitment Letter or JPMCB's commitment hereunder; provided that your obligations under this Commitment Letter (other than your obligations with respect to (a) assistance to be provided in connection with the syndication thereof and (b) confidentiality of the Fee Letter and the contents thereof) shall automatically terminate and be superseded by the provisions of the Credit Documentation (as defined in the Term Sheet) upon the initial funding thereunder, and you shall automatically be released from all liability in connection therewith at such time.

If the foregoing correctly sets forth our agreement, please indicate your acceptance of the terms hereof and of the Term Sheet and the Fee Letter by returning to us executed counterparts hereof and of the Fee Letter, together with the amounts agreed upon pursuant to the Fee Letter to be payable upon the acceptance hereof, not later than 9:00 a.m., New York City time, on June 21, 2010 (it being understood that fees paid by 5 p.m., New York City time, on June 21, 2010 shall be deemed paid upon acceptance hereof for purposes hereof and of the Fee Letter). JPMCB's commitment and JPMorgan's agreements herein will expire at such time in the event JPMCB has not received such executed counterparts and such amounts in accordance with the immediately preceding sentence.

6

JPMCB and JPMorgan are pleased to have been given the opportunity to assist you in connection with this important financing.

Very truly yours,

JPMORGAN CHASE BANK, NATIONAL ASSOCIATION

By: /s/ William J. Oleferchik

Name: William J. Oleferchik Title: Managing Director

J.P. MORGAN SECURITIES INC.

By: /s/ Brit J. Bartter

Name: Brit J. Bartter Title: Vice Chairman

Accepted and agreed to as of the date first written above by:

CORN PRODUCTS INTERNATIONAL, INC.

By: /s/ Cheryl K. Beebe

Name: Cheryl K. Beebe Title: Chief Financial Officer

7

EXHIBIT A

\$1,350,000,000

SENIOR BRIDGE FACILITY

Summary of Terms and Conditions

June 19, 2010

I. Parties

Borrower: Corn Products International, Inc., a Delaware corporation (the "Borrower").

Sole Lead Arranger and

Sole Bookrunner: J.P. Morgan Securities Inc. (in such capacity, the "<u>Lead Arranger</u>").

Administrative Agent: JPMorgan Chase Bank, N.A. ("<u>JPMCB</u>" and, in such capacity, the "<u>Administrative Agent</u>").

Syndication Agent: To be determined.

Documentation Agents: To be determined.

Lenders: A syndicate of banks, financial institutions and other entities, including JPMCB, arranged by the Lead

Arranger and reasonably acceptable to the Borrower (collectively, the "Lenders").

II. **Term Loan Facility**

Type and Amount of Facility:

364 day term loan facility (the "Facility.") in the amount of \$1,350,000,000 (the loans thereunder, the

"Term Loans"). The Facility shall be denominated entirely in US Dollars.

Availability:

Available in a single drawing on the Closing Date (which shall be the date of the execution and

delivery of the Credit Documentation (as defined below) by the parties thereto).

Amortization:

None.

Maturity:

All Term Loans will be payable in full on the date 364 days after the Closing Date.

Purpose:

The proceeds of the Term Loans shall be used (i) to finance in part the acquisition (the "Acquisition") by the Borrower and/or one or more of its subsidiaries of the specialty starch business

(the "Business") of Akzo Nobel N.V. (the "Seller"), which Business is comprised of certain assets of the Seller and its subsidiaries (including equity interests in certain of the Seller's subsidiaries), for an aggregate purchase price of approximately \$1,300,000,000 and (ii) to pay related fees and expenses.

III. **Certain Payment Provisions**

Fees and Interest Rates:

As set forth on Annex I.

Optional Prepayments and **Commitment Reductions:**

Loans may be prepaid and commitments may be reduced by the Borrower in minimum amounts to be mutually agreed upon without premium or penalty (other than customary breakage). Voluntary prepayments shall be applied as directed by the Borrower.

Mandatory Prepayments and Commitment Reductions:

The Term Loans shall be prepaid with (and, on or prior to the Closing Date the commitments in respect of the Facility shall be reduced by) (i) 100% of the net cash proceeds of any sale or issuance of equity by the Borrower or its subsidiaries (excluding the issuance and sale of equity to the Borrower or its subsidiaries or pursuant to employee stock plans and other similar arrangements), (ii) 100% of the net cash proceeds of any incurrence of indebtedness for borrowed money by the Borrower or its subsidiaries (other than (a) borrowings under the Existing Credit Agreement or the Revolving Credit Agreement (each as defined below), (b) intercompany indebtedness or (c) debt of foreign subsidiaries for working capital purposes or otherwise in the ordinary course of business (but excluding, in any event, debt issued in any public or Rule 144A-style capital markets transaction), (iii) 100% of the net cash proceeds in the aggregate from any sales or other dispositions of any assets by the Borrower or its subsidiaries (other than (a) sales of inventory in the ordinary course of business and (b) dispositions of assets by the Borrower or its subsidiaries with aggregate fair market value of less than \$25,000,000 from and after the date of the Commitment Letter), (iv) 100% of all casualty or condemnation insurance proceeds relating to assets of the Borrower or its subsidiaries (subject to 365 day reinvestment rights and other terms to be agreed) and (v) 50% of Free Cash Flow for each quarter ending on or after December 31, 2010 (with definitions to be agreed).

The Borrower shall give prompt notice to JPMorgan (prior to the execution and delivery of the Credit Documentation) and to the Administrative Agent after the execution and delivery of the Credit Documentation) of any event that reduces commitments or requires prepayment pursuant to the above.

2

IV. **Conditions Precedent**

Closing Date Conditions:

The availability of the Facility shall be conditioned only upon satisfaction of (i) the conditions precedent specified in the Commitment Letter and (ii) the following conditions precedent (the date upon which all such conditions precedent shall be satisfied and the Facility is funded, the "Closing Date") on or before December 18, 2010:

- The Borrower shall have executed and delivered mutually satisfactory definitive financing documentation (including a credit agreement) with respect to the Facility (the "Credit Documentation").
- The Lenders, the Administrative Agent and the Lead Arranger shall have received all fees required to be paid, and all out-of-pocket expenses for which invoices have been presented, on or before the Closing Date.
- The Lenders shall have received (i) audited consolidated financial statements of each of the Borrower and the Business for the two most recent fiscal years ended prior to the Closing Date as to

which such financial statements are available (it being understood that financial statements for the Business are carve-out financials that have been prepared according to International Reporting Financial Standards) and (ii) unaudited interim consolidated financial statements of the Borrower for each quarterly period ended subsequent to the date of the latest financial statements delivered pursuant to clause (i) of this paragraph as to which such financial statements are available as of the Closing Date (the most recent such quarterly period being the "Reference Quarter").

- (d) The Administrative Agent shall have received with respect to the Borrower and its subsidiaries (including subsidiaries acquired in the Acquisition) (i) pro forma financial statements as of the end of the Reference Quarter (predicated upon the financial statements of the Borrower as of the end of the Reference Quarter and the financial statements of the Business for the most recent quarter for which such financial statements are available) giving effect to the Acquisition and the financing contemplated hereunder and (ii) projections through 2014.
- (e) An agreement or agreements pursuant to which the Acquisition shall be effected (the "Acquisition Agreement") shall have been entered into by the Seller and the Borrower, shall be in form and substance reasonably satisfactory to the Administrative Agent (it being acknowledged that the drafts of such documentation most recently delivered to the

3

Administrative Agent by the Borrower prior to the date of this Term Sheet are reasonably satisfactory to the Administrative Agent) and shall be in full force and effect. All conditions precedent under the Acquisition Agreement shall have been satisfied (and not waived) and any amendment to the Acquisition Agreement materially adverse to the interests of the Lead Arranger, the Administrative Agent or the Lenders shall be satisfactory in form and substance to the Administrative Agent; it being understood and agreed that any material change to the transaction structure, any amendment of the definition of "Group Material Adverse Effect" contained in the Acquisition Agreement, any change to the definition of "Deferred Jurisdiction" in the Acquisition Agreement and any increase or decrease in the "Offer Value" referred to in the Acquisition Agreement, shall in each case be deemed to be materially adverse to the interests of the Lead Arranger, the Administrative Agent and the Lenders. The Acquisition shall be consummated substantially contemporaneously with the making of the Term Loans.

- (f) All regulatory, legal and other third party approvals necessary in connection with the Acquisition and the financing thereof shall have been obtained (except for those with respect to which (i) the failure to have obtained could not reasonably be expected to have a material adverse effect on the Borrower, the Business, the Acquisition, the financing thereof or any of the other transactions contemplated hereby, or (ii) the sole consequence of the failure to obtain any such approval is to defer the consummation of the acquisition of certain assets of the Business in "Deferred Jurisdictions" (as defined in the Acquisition Agreement) pursuant to the terms of the Acquisition Agreement).
- (g) No injunction or temporary restraining order exists and no litigation has commenced or is otherwise pending which would prohibit the execution of the Facility, the financings thereunder or any of the other transactions contemplated thereby.
- (h) Substantially contemporaneously with the making of the Term Loans, the Borrower shall (i) enter into a \$750,000,000 revolving credit agreement (the "Revolving Credit Agreement") in form and substance reasonably satisfactory to the Administrative Agent, repay in full all obligations under the Revolving Credit Agreement (the "Existing Credit Agreement") dated April 26, 2006 with SunTrust Bank, as administrative agent, and cause the Existing Credit Agreement and all related liens to be terminated or released or (ii) cause the Existing Credit Agreement to be amended (and/or cause the provisions

4

thereof to be waived) on terms satisfactory to the Administrative Agent to permit the transactions contemplated hereby.

- (i) Immediately after giving effect to the consummation of the Acquisition and the incurrence of any indebtedness in connection therewith, the Borrower shall be in pro-forma compliance with the financial covenants set forth under the heading "Financial Covenants" below and shall have delivered to the Administrative Agent a certificate of its Chief Financial Officer to such effect (which certificate shall include calculations reflecting such compliance and shall be predicated upon the financial statements of the Borrower as of the end of the Reference Quarter and the financial statements of the Business for the most recent quarter for which such financial statements are available) in form and substance reasonably satisfactory to the Administrative Agent.
- (j) The Lenders shall have received such legal opinions, documents and other instruments as are customary for transactions of this type or as they may reasonably request.

- (k) All representations and warranties in the Credit Documentation shall be accurate in all material respects (subject to the Conditionality Provision).
- (l) There shall be no default or event of default in existence at the time of, or after giving effect to the making of, such extension of credit.

V. <u>Certain Documentation Matters</u>

The Credit Documentation shall contain the following representations, warranties, covenants and events of default (subject to materiality thresholds, exceptions, qualifications and baskets to be mutually agreed):

Representations and Warranties:

Financial statements; no material adverse change of the Borrower since December 31, 2009 as of the Closing Date only; no Group Material Adverse Effect since the date of execution and delivery of the Acquisition Agreement; corporate existence; compliance with law; corporate power and authority; execution, delivery and enforceability of Credit Documentation; no conflict with law or contractual obligations; no material litigation; ownership of property; liens; intellectual property; taxes; margin regulations; ERISA; Investment Company Act; environmental matters; accuracy of disclosure; labor relations; OFAC; Patriot Act/Anti-Terrorism.

Affirmative Covenants:

Delivery of financial statements, reports, officers' certificates and other information requested by the Lenders; payment of

5

other material obligations; continuation of business and maintenance of existence and material rights and privileges; compliance with laws and payment of taxes; maintenance of property and insurance; maintenance of books and records; right of the Lenders to inspect property and books and records; notices of defaults, litigation and other material events; margin regulations; and compliance with environmental laws.

Financial Covenants:

Financial covenants shall be the following (with definitions to be mutually agreed):

- Ratio of Total Debt (net of cash on the consolidated balance sheet of the Borrower in excess of \$50,000,000) to trailing four quarter Consolidated EBITDA (giving pro forma effect to acquisitions and dispositions) shall at no time exceed 3.25:1.00.
- Ratio of trailing four quarter Consolidated EBITDA to trailing four quarter Interest Expense shall not be less than 3.50:1.00 as of any quarter end.
- "Consolidated EBITDA" shall mean, for any period, an amount equal to consolidated net income (or net loss) of the Borrower plus, to the extent deducted in determining consolidated net income for such period, the sum of (a) net interest expense, (b) income tax expense, (c) depreciation expense, (d) amortization expense, (e) non-recurring, non-cash charges and noncash restructuring charges and (f) minority interest earnings, minus, to the extent included in determining consolidated net income for such period, the sum of (y) minority interest losses and (z) non-recurring, non-cash gains and non-cash restructuring gains, in each case determined in accordance with GAAP by reference to the consolidated financial statements of the Company required to be delivered pursuant to the Credit Documentation. If the Borrower or a subsidiary consummates or has consummated an acquisition or a disposition during any period, then, for the purposes of calculating the financial covenants for such period, Consolidated EBITDA for such period shall be adjusted on a proforma basis to give effect to such acquisition or disposition as though such acquisition or disposition had been consummated as of the first day of such period.

Negative Covenants:

Limitations on: indebtedness; liens; guarantee obligations; mergers, consolidations, liquidations and dissolutions; sales of assets; dividends and other payments in respect of capital stock (to be permitted so long as no default exists or would result therefrom and the Borrower is in pro forma compliance with the financial covenants); transactions with affiliates; negative pledge clauses; restrictive agreements; and changes in lines of business.

(

Events of Default:

Non-payment of principal when due; non-payment of interest, fees or other amounts after a grace period to be agreed upon; material inaccuracy of representations and warranties; violation of covenants (subject, in the case of certain affirmative covenants, to a grace period of 30 days); cross-default to other material indebtedness; bankruptcy events; certain ERISA events; material judgments; invalidity of loan documents; and a change of control (the definition of which is to be agreed).

Voting:

Amendments and waivers with respect to the Credit Documentation shall require the approval of Lenders holding greater than 50% of the aggregate amount of the Term Loans (the "Required Lenders"), except that (a) the consent of each Lender directly affected thereby shall be required with respect to (i) reductions in the amount or extensions of the final maturity of any Loan, (ii) reductions

in the rate of interest or any fee or extensions of any stated due date thereof, and (iii) increases in the amount or extensions of the expiry date of any Lender's commitment and (b) the consent of 100% of the Lenders shall be required with respect to modifications to any of the voting percentages and modifications to pro rata treatment.

In addition, for so long as the Bridge Facility is outstanding, the Credit Documentation shall be automatically amended to reflect any changes or amendments to the Revolving Credit Agreement (or Existing Credit Agreement, as applicable), to the extent favorable to the Administrative Agent or the Lenders under the Facility (other than changes to the interest rates under the Revolving Credit Agreement or Existing Credit Agreement).

Assignments and Participations:

The Lenders shall be permitted to assign all or a portion of their loans and commitments with the consent, not to be unreasonably withheld, of (a) the Borrower, unless (i) the assignee is a Lender, an affiliate of a Lender or an approved fund or (ii) an Event of Default has occurred and is continuing and (b) the Administrative Agent, unless a Term Loan is being assigned to a Lender, an affiliate of a Lender or an approved fund. In the case of partial assignments (other than to another Lender, to an affiliate of a Lender or an approved fund), the minimum assignment amount shall be \$1,000,000, unless otherwise agreed to by the Borrower and the Administrative Agent. An assignment fee of \$3,500 per assignment shall be payable to the Administrative Agent in connection with each assignment.

The Lenders shall also be permitted to sell participations in their Loans. Participants shall have the same benefits as the applicable selling Lenders with respect to yield protection and increased cost provisions. Voting rights of participants shall be limited to those matters with respect to which the affirmative vote of the Lender from which it purchased the applicable

7

participation would be required as described under "Voting" above. Pledges of Loans in accordance with applicable law shall be permitted without restriction. Promissory notes shall be issued under the Facility promptly following request.

Yield Protection:

The Credit Documentation shall contain customary provisions (a) protecting the Lenders against increased costs or loss of yield resulting from changes in reserve, tax, capital adequacy and other requirements of law and from the imposition of or changes in withholding or other taxes after the Closing Date, (b) indemnifying the Lenders for "breakage costs" incurred in connection with, among other things, any prepayment of a Eurodollar Loan (as defined in Annex I) on a day other than the last day of an interest period with respect thereto and (c) allowing replacement of Lenders requesting payment of amounts referred to in clause (a) above.

Expenses and Indemnification:

The Borrower shall pay (a) all reasonable and documented out-of-pocket costs and expenses of the Administrative Agent and the Lead Arranger associated with the syndication of the Facility and the preparation, execution, delivery and administration of the Credit Documentation and any amendment or waiver with respect thereto (including the reasonable fees, disbursements and other charges of counsel) and (b) all reasonable and documented out-of-pocket costs and expenses of the Administrative Agent and the Lenders (including the fees, disbursements and other charges of counsel) in connection with the enforcement of the Credit Documentation.

The Administrative Agent, the Lead Arranger and the Lenders (and their affiliates and their respective officers, directors, employees, advisors and agents) will have no liability for, and will be indemnified and held harmless against, any loss, liability, cost or expense incurred in respect of the financing contemplated hereby or the use or the proposed use of proceeds thereof (except to the extent resulting from the gross negligence, bad faith or willful misconduct of the indemnified party).

Governing Law and Forum:

State of New York.

Counsel to the Administrative Agent:

Winston & Strawn LLP.

8

Annex I

Interest and Certain Fees

Interest Rate Options:

The Borrower may elect that the Term Loans comprising each borrowing bear interest at a rate per annum equal to:

- (i) the ABR plus the Applicable Margin; or
- (ii) the Adjusted LIBO Rate plus the Applicable Margin.

As used herein:

"ABR" means the highest of (i) the rate of interest publicly announced by JPMCB as its prime rate in effect at its principal office in New York City (the "Prime Rate"), (ii) the federal funds effective rate from time to time plus 0.5% and (iii) the Adjusted LIBO Rate for a one month interest period on such day (or if such day is not a business day, the immediately preceding business day) plus 1%, provided that, for the avoidance of doubt, the Adjusted LIBO Rate for any day shall be based upon the rate appearing on the Reuters Screen LIBOR01 (or on any successor or substitute page of such page) at approximately 11:00 a.m. London time on such day. Any change in the ABR due to a change in the Prime Rate, the federal funds effective rate or the Adjusted LIBO Rate shall be effective rate or the Adjusted LIBO Rate, respectively.

"Adjusted LIBO Rate" means the LIBO Rate, as adjusted for statutory reserve requirements for eurocurrency liabilities.

"<u>Applicable Margin</u>" means (a) 1.25%, in the case of ABR Loans (as defined below) and (b) 2.25%, in the case of Eurodollar Loans (as defined below); provided, however, that in each case, the Applicable Margin shall be increased by an incremental .50% on each of the 90th day after the Closing Date, the 180th day after the Closing Date and the 270th day after the Closing Date.

"LIBO Rate" means the rate at which eurodollar deposits in the London interbank market for one, two, three or six months, or if available to all Lenders, nine or twelve months (as selected by the Borrower) are quoted on the Reuters Screen LIBOR01.

Interest Payment Dates:

Default Rate:

Duration Fee:

In the case of Term Loans bearing interest based upon the ABR ("ABR Loans"), quarterly in arrears.

In the case of Term Loans bearing interest based upon the Adjusted LIBO Rate ("<u>Eurodollar Loans</u>"), on the last day of each relevant interest period and, in the case of any interest period longer than three months, on each successive date three months after the first day of such interest period.

At any time when the Borrower is in default in the payment of any amount of principal due under the Facility, and at the election of the Required Lenders when the Borrower is in default under another provision of the Facility, the Term Loans shall bear interest at 2% above the rate otherwise applicable thereto. Overdue interest, fees and other amounts shall bear interest at 2% above the rate applicable to ABR Loans.

The Borrower shall pay a fee for the ratable benefit of the Lenders on the dates set forth below equal to the percentage of the aggregate principal amount of Term Loans outstanding on such date:

90 days after the Closing Date:	0.75%
180 days after the Closing Date:	1.25%
270 days after the Closing Date:	1.75%

All per annum rates shall be calculated on the basis of a year of 360 days (or 365/366 days, in the case of ABR Loans the interest rate payable on which is then based on the Prime Rate) for actual days elapsed.

Rate and Fee Basis:



Corn Products International, Inc. 5 Westbrook Corporate Center Westchester, IL 60154

NEWS RELEASE

FOR IMMEDIATE RELEASE:

June 21, 2009, 2:00 A.M. ET

CONTACT:

Investors: John Barry, (708) 551-2592 Media: Mark Lindley, (708) 551-2602

CORN PRODUCTS INTERNATIONAL TO ACQUIRE NATIONAL STARCH FOR \$1.3 BILLION IN CASH, TO CREATE A NEARLY \$5 BILLION GLOBAL INGREDIENT SOLUTIONS LEADER

ACQUISITION BROADENS PRODUCT PORTFOLIO AND ENHANCES INNOVATION
EXPANDS CORN PRODUCTS' GEOGRAPHIC FOOTPRINT IN EUROPE AND ASIA-PACIFIC
COMPANY ANTICIPATES SYNERGIES OF AT LEAST \$50 MILLION

WESTCHESTER, ILLINOIS, JUNE 21, 2010 — Corn Products International, Inc. (NYSE: CPO), a leading global provider of ingredient solutions for diversified industries, announced today that it has entered into a definitive agreement to acquire National Starch, a New Jersey-based global provider of specialty starches, from AkzoNobel, the largest global coatings and specialty chemicals company headquartered in The Netherlands.

"The acquisition of National Starch represents an exceptional opportunity for our Company and a significant step forward toward achieving our strategic goals," Gordon said. "The acquisition aligns with our strategic priorities to grow our ingredient portfolio, increase our presence in priority food processing segments, enter new markets, and develop innovative solutions that better serve our customers."

"Combining National Starch with Corn Products will create an ingredient solutions leader with nearly \$5 billion in revenues, "Gordon added. "The combined company will benefit from leading, innovative technology; enhanced geographic scale; a broader and deeper product portfolio; and access to new market segments."

National Starch is a recognized innovator in food ingredients and specialty starches. The company's technologies are supported by a world-class research and development infrastructure and protected by more than 800 patents and patents pending, which drive the development of advanced specialty starches for the next generation of food products. National Starch had 2009 revenues of \$1.2 billion from sales of specialty starches to both local and multinational customers in the food, papermaking, consumer and industrial segments. It has 2,250 employees around the world and operates 11 plants in eight countries, including new geographies for Corn Products such as the United Kingdom, Germany, Australia, and New Zealand.

National Starch is a leader in the development of specialty and modified starches — two of the ingredient solutions that Corn Products has identified as important to its ongoing growth. National Starch's ingredient solutions, such as texturants, blends and environmentally "green" solutions, will broaden Corn Products' offerings to its global customers.

"This acquisition gives us access to new markets such as Europe, and improves our scale and capabilities in many of our existing global locations," said Gordon. "We will be able to better serve our global customers with our enhanced geographic footprint and the addition of National Starch's technology. Further, we expect significant production efficiencies and cost synergies that will make us more competitive around the world."

"We are acquiring National Starch, not just because of its products and manufacturing assets, but also because of its experienced and dedicated people," Gordon added. "They bring to Corn Products exceptional know-how in ingredient development, solutions and applications, along with proven marketing and customer service skills."

"We are confident that this acquisition will create lasting, long-term value for our shareholders," Gordon said. The transaction is expected to generate cost synergies of at least \$50 million, primarily from efficiencies in the areas of manufacturing, procurement, logistics and general & administrative functions.

2

"We expect to move quickly to integrate and attain synergy savings" Gordon concluded.

On cash basis, the transaction is expected to be accretive by the end of 2011.

The acquisition, which has been approved by the Boards of Directors of both Corn Products International and AkzoNobel, is expected to close in the third quarter of 2010, subject to customary regulatory approvals.

The Company expects to finance the transaction through cash, debt and new equity. Corn Products is committed to maintaining its investment grade rating.

J.P. Morgan Securities Inc. acted as exclusive financial advisor to Corn Products International.

CONFERENCE CALL INFORMATION

Corn Products International will discuss this acquisition on a conference call today at 9:30 a.m. Eastern Time (8:30 a.m. Central Time) to be hosted by Ilene Gordon, Chairman, President and Chief Executive Officer, and Cheryl Beebe, Chief Financial Officer. The call will be broadcast in a real-time webcast that will consist of the call and a visual presentation accessible through the Corn Products International web site at www.cornproducts.com.

A replay of the webcast will be available at www.cornproducts.com. Individuals without Internet access may listen to the live conference call by dialing 719-3254753. A replay of the audio call will be available through Friday, July 2, by calling 719-457-0820 and using passcode 6350941.

About the Company

Corn Products International is one of the world's largest corn refiners and a major supplier of high-quality food ingredients and industrial products derived from the wet milling and processing of corn and other starch-based materials. The Company, headquartered in Westchester, Ill., is a leading worldwide producer of dextrose and a major regional producer of starches, sweeteners and other ingredients. It has 8,000 employees and operations in 13 countries at 28 plants, including wholly

3

owned businesses, affiliates and alliances. In 2009, Corn Products International reported net sales of \$3.67 billion. For more information, visit www.cornproducts.com.

Forward-Looking Statements

This news release contains or may contain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. The Company intends these forward-looking statements to be covered by the safe harbor provisions for such statements. These statements include, among other things, any predictions regarding the Company's prospects or future financial condition, earnings, revenues, expenses or other financial items, any statements concerning the Company's prospects or future operations, including management's plans or strategies and objectives therefor, expectations regarding the proposed acquisition, including synergies, time of closing, accretion and credit ratings, and any assumptions, expectations or beliefs underlying the foregoing. These statements can sometimes be identified by the use of forward looking words such as "may," "should," "will," "anticipate," "believe," "plan," "project," "estimate," "expect," "intend," "continue," "pro forma," "forecast" or other similar expressions or the negative thereof. All statements other than statements of historical facts in this news release or referred to in this news release are "forward-looking statements." These statements are based on current expectations, but are subject to certain inherent risks and uncertainties, many of which are difficult to predict and are beyond our control. Although we believe our expectations reflected in these forward-looking statements are based on reasonable assumptions, stockholders are cautioned that no assurance can be given that our expectations will prove correct. Actual results and developments may differ materially from the expectations expressed in or implied by these statements, based on various factors, including the effects of the global economic recession and its impact on our sales volumes and pricing of our products, our ability to collect our receivables from customers and our ability to raise funds at reasonable rates; fluctuations in worldwide markets for corn and other commodities, and the associated risks of hedging against such fluctuations; fluctuations in the markets and prices for our co-products, particularly corn oil; fluctuations in aggregate industry supply and market demand; the behavior of financial markets, including foreign currency fluctuations and fluctuations in interest and exchange rates; continued volatility and turmoil in the capital markets; the commercial and consumer credit environment; general political, economic, business, market and weather conditions in the various geographic regions and countries in which we manufacture and/or sell our products; future financial performance of major industries which we serve, including, without limitation, the food and beverage, pharmaceuticals, paper, corrugated, textile and brewing industries; energy costs and availability, freight and shipping costs, changes in regulatory controls regarding quotas, tariffs, duties, taxes and income tax rates; operating difficulties; boiler reliability; our ability to effectively integrate acquired businesses; labor disputes; genetic and biotechnology issues; changing consumption preferences and trends; increased competitive and/or customer pressure in the corn-refining industry; and the outbreak or continuation of serious communicable disease or hostilities including acts of terrorism. Factors relating to the proposed acquisition that could cause actual results and developments to differ from expectations include: required regulatory approvals may not be obtained in a timely manner, if at all; the proposed acquisition may not be consummated in a timely manner or at all; the anticipated benefits of the proposed acquisition, including synergies, may not be realized; the integration of National Starch's operations with those of Corn Products may be materially delayed or may be more costly or difficult than expected, and we may be unable to maintain our current credit ratings. Our forward-looking statements speak only as of the date on which they are made and we do not undertake any obligation to update any forward-looking statement to reflect events or circumstances after the date of the statement as a result of new information or future events or developments. If we do update or correct one or more of these statements, investors and others should not conclude that we will make additional updates or corrections. For a further description of these and other risks, see "Risk Factors" included in our Annual Report on Form 10-K for the year ended December 31, 2009 and subsequent reports on Forms 10-Q or 8-K. This news release also may contain references to the Company's long term objectives and goals or targets with respect to certain metrics. These objectives, goals and targets are used as a motivational and management tool and are indicative of the Company's long term aspirations only, and they are not intended to constitute, nor should they be interpreted as, an estimate, projection, forecast or prediction of the Company's future performance.